Fabrication and characterization of germanium-on-silicon photodiodes

by

Nicole Ann DiLello

B.S.E., Princeton University (2005) S.M., Massachusetts Institute of Technology (2008)

Submitted to the Department of Electrical Engineering and Computer Science

in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

February 2012

© Massachusetts Institute of Technology 2012. All rights reserved.

Certified by U Judy L. Hoyt Professor Thesis Supervisor

Accepted by Professor Leslie Kolodziejski Chair, Department Committee on Graduate Students



ARCHIVES

2

.

.

Fabrication and characterization of germanium-on-silicon

photodiodes

by

Nicole Ann DiLello

Submitted to the Department of Electrical Engineering and Computer Science on January 26, 2012, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

Germanium is becoming an increasingly popular material to use in photonic systems. Due to its strong absorption in the near infrared and its relative ease of integration on silicon, it is a promising candidate for the fabrication of CMOS-compatible photodetectors.

The goal of this thesis is to understand the physics of Ge-on-Si photodiodes, especially the dark current. Low-pressure chemical vapor deposition was used to deposit thick $(1 - 2 \mu m)$ films on silicon substrates either selectively in oxide windows or in blanket films. Photodetectors were fabricated in both types of films and their optical and electronic properties are discussed.

It was found that the main source of leakage current in these detectors is the generation of carriers at the Ge/passivation interface. This especially affects small devices, as the perimeter/area ratio is much larger than for large devices. A post-metallization anneal in nitrogen at 400°C was found to reduce the dark current of small devices ($10 \times 10 \mu$ m) by ~1000X at -1 V. The same anneal reduces the dark current of larger devices ($100 \times 100 \mu$ m) by ~140X. Through metal-oxide-semiconductor capacitor and doping studies, it was found that the anneal draws holes to the surface of the germanium, leading to better isolation of the devices and reduced leakage current.

It was also found that threading defects play a role in leakage current. Threading defects arise because of the 4% lattice mismatch between germanium and the underlying silicon. For 1 μ m-thick germanium films, as-grown samples are expected to have ~5 x 10⁸ cm⁻² threading defects. At this level, these defects are the dominant leakage current mechanism. Annealing the films at high temperatures can reduce the defect density.

Large-area $(300 \times 300 \ \mu\text{m})$ devices fabricated with a post-metallization anneal and with a threading defect density of $\sim 2 \times 10^7 \text{ cm}^{-2}$ were found to have a dark current density of $\sim 1 \text{ mA/cm}^2$ and a responsivity of 0.32 A/W at -1 V and 1550 nm.

Thesis Supervisor: Judy L. Hoyt Title: Professor

Acknowledgments

This thesis would not have been possible without the support and help from a small army of people. I must first thank my adviser, Judy Hoyt, for her technical support throughout my years of graduate school. Her mind is like a steel trap and the solution to whatever problem I was facing was, more often than not, something she had read about or experienced ten or twenty years ago. I must also thank the other members of my committee, Tayo Akinwande and Clif Fonstad. Their unique perspectives have made this thesis more complete and thorough and their encouragement was invaluable.

Nearly all of the work for this thesis was done in MIT's Microsystems Technology Lab, truly the best lab on campus. Someone told me when I showed up that MTL was like a family and they were right. From the fabrication staff to the administrative staff to the professors to the janitors, the folks around building 39 are some of the loveliest people ever to work at MIT. I must thank Bernard, Bob, Eric, Paudely, Paul, Donal, Kurt, Dave, and Dennis not only for training me and helping me in the fab but for being friendly faces (behind the bunny suits!) and a joy to work with. I also must thank Vicky, Debb, and Mara for their organization and passion for MTL. The staff member who has done the most to help this thesis is Gary Riggott. He grew many germanium films for this research and never once complained when I asked him to track down epi numbers and recipes.

The sixth floor, in all its iterations, has been a source of laughter and technical help throughout my years in graduate school. It was great to have women like Niamh, Anita, Joyce, Liang-Yu, Annie, Melissa, and Lan around. The men of the sixth floor weren't too shabby either: Ling, Jung Woo, John, Osama, Jae-Kyu, Omair, Allen, Bin, Steve, and Michael were always good for a laugh or a drink.

I must thank past and present members of the Hoyt group. Drs. Aberg, Xia, Olubuyide, and Ni Chleirigh set great examples of PhD-level work and were always willing to help when I was first getting started. More recently, Drs. Gomez, Hashemi, and Kim were with me for most of my time at MIT and were always there to celebrate a success or commiserate over a failure. It's been a pleasure to work with Jamie, Winston, and Tau over the last few years and I wish them the best of luck in their studies.

I'd also like to thank the Hoyt/Antoniadis administrators over the years. Rose, Michelle, and Whitney kept the group running and never laughed at me when I couldn't properly package a box of wafers to ship somewhere. Whitney was especially patient with me as I went through numerous processing freak-outs.

The best thing a graduate student can do is have friends outside of his department (or even not in graduate school altogether). To that end, I'd like to thank Dan, Sarah, Brad, Erik, Melissa, Jon, Pete, Mackenzie, and Ben for having very few ties to MIT and for talking more about sports and current events than semiconductors. I'd like to apologize to Sarah for not having time to be in her movie - now that the thesis is done, I'll gladly be in the next one. I also must thank Maureen, Stefanie, Julie, Yeppie, and Courtney for being some of the best girlfriends in the world. I'm honored that they're all going to be in the wedding.

There were a handful of people at MIT who played a very integral role in making these last six years great. Jonathan Morse, Jennifer Roberts, Vanessa Wood, Orit Shamir, Tania Khanna, and Yu Gu were unfailingly supportive and reliable throughout everything. I also need to thank Junjay, Kyle, Abe, Zoe, and Jesse for adopting me into their ESD group. After I took Oye's class, I fit right in! I also must thank Cory for being a wonderful roommate and fashion consultant. I have no idea how I'll pick out clothes without her around.

Finally, I must thank my family, especially my parents and brother, for being a constant stream of support and love. They may not have had a clue about what I was doing for the last six years, but they were constantly telling me I could do it. I

also have to thank Tim, for being a source of support, adventure, and many hugs. I look forward to our life together - it's going to be hard to find places on the wall for all these degrees of ours!

Contents

1	Intr	oduct	ion	15		
	1.1	Impor	tance of the diode dark current	17		
	1.2	Basic	structure of Ge-on-Si photodiodes	18		
	1.3	Roadr	nap of this thesis	19		
2	The	esis Ba	ckground	21		
	2.1	Physic	es of photodiodes	21		
	2.2	Optica	al properties of germanium	23		
	2.3	Integr	ation of germanium on silicon	25		
		2.3.1	Germanium growth	27		
	2.4	Ge-on	-Si diodes	29		
		2.4.1	Figures of merit for photodiodes	29		
		2.4.2	Previous work on blanket diodes	30		
		2.4.3	Previous work on selective diodes	33		
		2.4.4	Waveguide integrated Ge diodes	35		
		2.4.5	Other types of Ge-on-Si diodes	37		
	2.5	Scope	of this thesis and chapter summary	38		
વ		Photodiodes fabricated using selectively grown germanium				
3	Pho	otodioo	les fabricated using selectively grown germanium	39		
3	Pho 3.1	otodioo Select	les fabricated using selectively grown germanium	39 40		
3	Pho 3.1	otodioo Select 3.1.1	les fabricated using selectively grown germanium ive germanium growth	39 40 40		
3	Pho 3.1	todioc Select 3.1.1 3.1.2	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting	39 40 40 42		
3	Pho 3.1	todioo Select 3.1.1 3.1.2 3.1.3	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations	39 40 40 42 43		
3	Phc 3.1 3.2	todioc Select 3.1.1 3.1.2 3.1.3 Fabric	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations ation of photodiodes	39 40 40 42 43 43		
3	Pho 3.1 3.2 3.3	Select 3.1.1 3.1.2 3.1.3 Fabric Electr	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations ation of photodiodes ical characterization	 39 40 40 42 43 43 45 		
3	Pho 3.1 3.2 3.3	5 Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations ation of photodiodes ical characterization Effect of cyclic annealing on dark current	 39 40 40 42 43 43 45 45 		
3	Pho 3.1 3.2 3.3	todiod Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations ation of photodiodes Effect of cyclic annealing on dark current Effect of orientation on dark current	 39 40 40 42 43 43 45 45 45 47 		
3	Pho 3.1 3.2 3.3	5 Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3	les fabricated using selectively grown germanium ive germanium growth Surface preparation Hole filling and faceting Growth parameters and threading dislocations ation of photodiodes Effect of cyclic annealing on dark current Effect of orientation on dark current Layout of the $n+$ region	 39 40 40 42 43 43 45 45 47 47 		
3	Pho 3.1 3.2 3.3	todiod Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentLayout of the $n+$ regionLowest dark current	 39 40 40 42 43 43 45 45 47 47 48 		
3	Pho 3.1 3.2 3.3	todiod Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentEffect of orientation on dark currentLayout of the $n+$ regionLowest dark currentSize dependence of dark current densities	 39 40 40 42 43 43 45 45 47 47 48 48 		
3	Pho 3.1 3.2 3.3 3.4	todiod Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 Optica	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentEffect of orientation on dark currentLayout of the $n+$ regionLowest dark currentSize dependence of dark current densitiesat characterization	 39 40 40 42 43 43 45 45 47 47 48 48 52 		
3	Pho 3.1 3.2 3.3 3.4	Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 Optica 3.4.1	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentLayout of the $n+$ regionLowest dark currentSize dependence of dark current densitiesal characterizationThickness of Ge film	 39 40 40 42 43 43 45 45 47 47 48 48 52 52 		
3	Pho 3.1 3.2 3.3 3.4	Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 Optica 3.4.1 3.4.2	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentLayout of the $n+$ regionLowest dark currentSize dependence of dark current densitiesal characterizationThickness of Ge filmDifferent size Ge islands	 39 40 40 42 43 43 45 45 47 47 48 48 52 52 52 		
3	Pho 3.1 3.2 3.3 3.4 3.4	Select 3.1.1 3.1.2 3.1.3 Fabric Electr 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 Optica 3.4.1 3.4.2 Waves	les fabricated using selectively grown germaniumive germanium growthSurface preparationHole filling and facetingGrowth parameters and threading dislocationsation of photodiodesical characterizationEffect of cyclic annealing on dark currentLayout of the $n+$ regionLowest dark currentSize dependence of dark current densitiesSize dependence of dark current densitiesDifferent size Ge islandsguide integrated devices	 39 40 40 42 43 43 45 45 47 47 48 52 52 56 		

		3.5.2 Diodes grown on 50 nm of Si	59
		3.5.3 Diodes grown laterally	64
	3.6	Chapter summary	68
4	Pho	otodiodes fabricated in blanket Ge-on-Si	69
	4.1	Ge growth and fabrication	70
		4.1.1 Process flow with mesa isolation	71
		4.1.2 Process flow without mesa etch	72
	4.2	Electrical characterization of blanket Ge-on-Si diodes	74
		4.2.1 Electrical characterization of mesa-isolated devices	75
		4.2.2 Electrical characterization of non-mesa isolated diodes	80
	4.3	Photoresponse of diodes fabricated from blanket Ge-on-Si	90
	4.4	Chapter summary	93
5	Dar	k current leakage mechanisms	95
	5.1	Overview of dark current mechanisms	95
	5.2	Experimental scaling and temperature dependencies	98
		5.2.1 Scaling and voltage dependence of devices	99
		5.2.2 Summary of scaling and voltage dependencies	104
	5.3	Modeling of annealed devices	105
	5.4	Modeling of small devices	108
	5.5	Origin of dark current	117
	5.6	Chapter summary	128
6	The	esis summary and future work	129
	6.1	Thesis summary	129
	6.2	Contributions	130
	6.3	Future work	131
\mathbf{A}	Pro	cess flows used in this thesis	133
	A.1	Mesa isolation flow	133
	A.2	Process flow without mesa isolation	135
	A.3	Alignment mark protection flow	137
В	\mathbf{Pro}	cedure for taking photoresponse measurements	139
\mathbf{C}	Ind	ucing strain in blanket Ge-on-Si	147
	C.1	Wafer bending theory	147
	C.2	Fabrication	148
	C.3	Results	149
D	Fre	quency response of diodes fabricated in selectively-grown Ge-on	-
	\mathbf{Si}		151
	D.1	Frequency response of vertical devices	151
	D.2	Frequency response of waveguide-integrated devices	152

List of Figures

1-1	A schematic diagram of the electronic-photonic analog to digital con- verter in this study.	16
1-2	An optical micrograph of part of the integrated electronic-photonic ADC.	17
1-3	A cross-sectional schematic diagram of a typical Ge-on-Si diode	19
2-1	Schematic illustrations of (a) a pn junction diode and (b) a pin junction	
	diode	22
2-2	The band structure of Ge at 300K	24
2-3	Absorption coefficient versus wavelength for a number of different semi- conductors.	24
2-4	A schematic illustration of how dislocations propagate in (a) a large	
	area and (b) a small area.	27
2-5	A schematic illustration of aspect ratio trapping.	28
2-6	Nomarski micrographs of $10 \ge 10 mu$ m Ge islands grown selectively.	28
2-7	Plot of absorption coefficient versus energy for bulk Ge and Ge-on-Si.	33
2-8	A cross-sectional SEM of selectively-grown Ge on Si utilizing a high	
	aspect ratio seed region.	34
2-9	Schematic diagrams of two types of waveguide integrated detectors.	36
2 - 10	A schematic illustration of (a) a lateral <i>pin</i> structure and (b) a vertical	
	pin structure.	36
3-1	Nomarski micrographs polycrystalline Ge and single crystalline Ge.	41
3-2	Spreading resistance profiling of three different activation anneal con-	
	ditions	44
3-3	Current versus voltage characteristic illustrating the effect of a cyclic	10
	anneal on the dark current of 5 x 10 μ m photodiodes	46
3-4	IV curves illustrating the effect of the orientation of the Ge film on the	10
~ ~	dark current of devices.	48
3-5	IV curves from diodes with different <i>n</i> -type layouts	49
3-6	IV curve from the diode with the lowest dark current.	50
3-7	Responsivity curve for two different Ge thicknesses.	53
3-8	Normalized responsivity curves for three different Ge island sizes.	54
3-9	Kesidual tensile strain in three different Ge island sizes	99
3-10	A cross-sectional scanning electron micrograph of the facets in the Ge	56
0 1 1	IIIII	50
9-11	Schematic diagram of a shicon modulator fabricated by MIT LL	00

3-12	Schematic diagrams of the two different types of waveguide-integrated	-0
3-13	Schematic diagrams of intended and actual waveguide-integrated Ge	58
0 10	diode.	59
3-14	Current versus voltage characteristic for a 100 x 0.25 $\mu \rm{m}$ waveguide-	
0.15	integrated device.	60
3-15	Responsivity versus wavelength curve for a 100 x 0.25 μ m waveguide- integrated device	61
3-16	Cross-sectional schematic of simulated device in Sentaurus.	62
3-17	Electric field at (a) -1 V and (b) -5 V.	63
3-18	Electric field down a vertical cut in the germanium.	64
3-19	A schematic cross-sectional diagram of a Ge diode grown laterally from	
2 00	Si sidewalls.	65
3-20	current versus voltage characteristics for two different lengths of laterally-	66
3-21	Responsivity versus wavelength characteristics for 5 mm long diodes.	67
4 1		771
4-1	Schematic process now of the mesa-isolated diode fabrication process.	71 72
4-2 1_3	Schematic process flow of the non-mesa isolated diodes	73 74
4-4	Current versus voltage characteristics for $5 \ge 5 \mu m$ devices with a doped	• •
	seed and an undoped seed	75
4-5	Current versus voltage characteristics for 10 x 10 μ m mesa-isolated	
	devices with different passivation materials.	77
4-6	Current versus voltage characteristics for mesa-isolated devices passi-	70
47	vated with Al_2O_3 with dopants activated at different temperatures.	18
4-1	and without an n -layer at the Ge surface	79
4-8	Current versus voltage characteristics for devices fabricated without a	
	mesa etch. Both curves are of 100 x 100 μ m devices passivated with	
	LTO or Al_2O_3 .	81
4-9	Current versus voltage characteristics (a) $10 \times 10 \ \mu m$ devices and (b)	
	100 x 100 μ m devices after post-metallization anneals at various tem-	ຊາ
4-10	Perimeter/area analysis at -1 V for devices (a) without and with (b) a	02
1 10	PMA.	84
4-11	Capacitance versus voltage characteristics for 300 x 300 μm devices	
	annealed at different temperatures in N_2	85
4-12	Capacitance versus voltage characteristics for 300 x 300 μ m devices	0.0
1 19	annealed in different ambients at 400°C.	86
4-13	with an anneal	87
4-14	(a) Current versus voltage charcteristics for $5 \ge 5 \mu m$ devices with	01
	different PMA and p -type doping characteristics. (b) A cross-sectional	
	schematic diagram of a device doped <i>in-situ p</i> -type at the surface	89

4-15	Current versus voltage characteristics for (a) 10 x 10 μ m devices and (b) 100 x 100 μ m devices fabricated in different runs	90
4-16	Photoresponse versus wavelength characteristic for a 200 x 200 μm device passivated with Al_2O_3 and fabricated with a mesa etch	91
4-17	Responsivity versus wavelength curve for a device fabricated without mesa isolation.	92
4-18	Responsivity versus wavelength curves for 200 x 200 μ m devices measured at -1 V	93
5-1	A schematic diagram of where the different sources of leakage current originate.	98
5-2	Current versus voltage characteristics scaled by (a) perimeter and (b) area for unannealed diodes.	100
5-3	Current versus voltage characteristics scaled by (a) perimeter and (b) area for annealed diodes.	100
5-4	Current versus voltage characteristics at different temperatures for (a) a 10 x 10 μ m device and (b) a 300 x 300 μ m device on an unannealed	
5-5	wafer	101
5-6	wafer	101
	nealed devices.	102
5-7	The natural log of the current v. the natural log of the voltage for a (a) $10 \ge 10 \ \mu \text{m}$ device and a (b) $300 \ge 300 \ \mu \text{m}$ device from an unannealed	109
5-8	The natural log of the current versus the natural log of the voltage for a (a) 10 x 10 μ m device and a (b) 300 x 300 μ m device from an annealed wafer	103
5-9	Spreading resistance profiling of boron in the germanium film.	106
5-10 5-11	Cross-sectional diagram of device modeled in Sentaurus Device. (a) Current versus voltage curves for a 300 x 300 μ m device and simu-	106
F 10	lated device with a minority carrier lifetime of 0.2 μ s. (b) In(J) versus ln(V) of experimental and simulated devices.	107
5-12	carrier lifetime.	108
5-13	A cross-sectional schematic diagram of the structure used to calculate the surface recombination velocity.	109
5-14	(a)Current versus voltage characteristics and (b) α extraction for devices of different sizes after an anneal	110
5-15	(a) A two dimensional plot of the surface generation current in the x direction	
5-16	just below the surface of the Ge	111
	sizes and (b) their extracted α dependence	113

5-17	Current versus voltage characteristics for simulated devices with $Q_f = -1 \ge 10^{12} \text{ cm}^{-2}$ and $Q_f = 0.5 \ge 10^{12} \text{ cm}^{-2}$.	114
5-18	Current versus voltage characteristics for simulated devices with $Q_f = 0.5 \times 10^{12} \text{ cm}^{-2}$ with and without a <i>n</i> -type layer	114
5-19	Current versus voltage characteristics for simulated devices with <i>p</i> -type	114
5-20	layers with $Q_f = -1 \ge 10^{12} \text{ cm}^{-2}$ and $Q_f = 0.5 \ge 10^{12} \text{ cm}^{-2}$ Current versus voltage characteristics for $5 \ge 5 \mu \text{m}$ experimental de-	115
	vices. One device received a PMA while the other had a <i>p</i> -type layer grown in the top of the germanium.	116
5-21	Schematic diagrams of the location of the high electric field for (a) a device doped p -type in the top of the germanium and (b) a device	
	which received a PMA	116
5 - 22	Defect levels in germanium due to material contaminants	117
5 - 23	SIMS profiles for (a) Ge, Si, O, C, and Au and (b) Ge, Si, Cu, Ag, Fe,	
	and Cr	118
5 - 24	Arrhenius plot of different trap levels found in a 300 x 300 μm device	
	that received a PMA	120
5 - 25	DLTS signal at 100K as a function of filling pulse duration for a 300 x	
	300 μ m device that received a PMA	121
5-26	Threading dislocation density as measured by plan-view transmission	
F 0 F	electron microscopy versus thickness for different annealing cycles.	122
5-27	Current versus voltage for 300 x 300 μ m devices annealed for a different	192
r 00	Dark surrout density sensus selters for different device sizes fobricated	123
0-20	Dark current density versus voltage for different device sizes fabricated	194
5 20	On a water which did not see a post-growth afficial. \dots \dots \dots \dots	124
J-29	decode (a) with a PMA and (b) without a PMA	195
5 20	Thermal denor concentration as a function of time and temperature	120
0-00	Thermal donor concentration as a function of time and temperature.	141
B-1	A schematic diagram of the photoresponse measurement setup	140
C-1	Responsivity versus wavelength curves for 200 x 200 μ m devices with various backside metallization schemes.	150
D-1	Frequency response of a vertical <i>nin</i> device	152
D_{-2}	Current versus voltage characteristic of a vertical <i>nin</i> device	153
D-3	Frequency response of a waveguide-integrated device.	154
	requency response of a waveguide integrated device.	- U I

Chapter 1

Introduction

As electronic systems reach their physical limits, new system architectures will be needed to continue technological progress. Integrated electronic-photonic systems are emerging as a potential solution to overcome the scaling limits of electronics. By moving certain functions from the electrical realm to the photonic realm, systems can save energy, operate faster, and exploit parallelism to a larger degree.

Due to the relative expense of materials traditionally used in optics compared to those in complementary metal-oxide-semiconductor (CMOS) systems, a large research effort is underway to integrate photonic components on a silicon chip [1] [2]. Germanium has an indirect bandgap at 0.66 eV, but it has a direct bandgap that is only slightly larger. Due to this small direct bandgap of 0.80 eV, germanium is an attractive candidate for use in optically absorbing devices and optically emitting devices. Researchers have recently found a way to fabricate a germanium laser [3] and germanium has also been used as an electro-optic modulator [4] [5]. Furthermore, SiGe alloys are already being used in CMOS systems [6], making integration of pure germanium an attractive possibility.

Photodiodes, or diodes that generate current when illuminated, are often used in integrated electronic-photonic systems either as an interconnect between the electron-



Figure 1-1: A schematic diagram of the electronic-photonic analog to digital converter in this study. Everything inside the dotted box can be integrated on a single silicon substrate. From [7].

ics and the photonics or as an infrared sensor. This thesis focuses on the fabrication, characterization, and modeling of Ge-on-Si photodiodes made for two applications. In the first application, the germanium photodiode will be used as a photodetector in an integrated electronic-photonic analog-to-digital converter (ADC). In this application, the sampling source of the ADC is switched from an electronic source to an optical source [7]. This change has the potential to reduce the timing jitter in the system by 100X. If, however, this substitution is made, a new family of silicon-compatible photonic devices is needed to complete the ADC. Silicon electro-optic modulators are needed to sample the outside signal, silicon-compatible waveguides are needed to transmit the signal across the chip, ring filters with microheaters are needed to select the signal, and, finally, photodetectors are needed to convert the signal back into an electronic signal at the end. A schematic diagram of such a system is illustrated in Figure 1-1 and an optical micrograph of a partial system is in Figure 1-2.

The second application for germanium photodiodes in this thesis is as infrared sensors for imaging applications. In this case, the goal is to integrate the sensor and the read-out circuitry on a Si substrate [8]. Currently, to absorb wavelengths in the



Figure 1-2: An optical micrograph of part of the integrated electronic-photonic ADC. The image shows Mach-Zender Si modulators, ring filters with micro-heaters, and photodetectors. From [7].

visible to near infrared, these systems are made using InGaAs for the detector and InP for the substrate [9]. While the detectors display excellent noise characteristics, the materials set is expensive. Switching to a silicon substrate has the potential to reduce the cost of the system. However, if silicon is used as the substrate, a new material would be needed for the detector, as InGaAs is very difficult to integrate on silicon. Germanium is a promising candidate for these new detectors because of its similar absorption spectrum to InGaAs and its relative ease of integration on silicon.

1.1 Importance of the diode dark current

The leakage current of photodiodes in the dark is known as the "dark current" of the device. In both applications discussed above, achieving a low dark current is of paramount importance. In the analog-to-digital converter, the system can tolerate a certain level of dark current (usually quoted as $\sim 1 \ \mu$ A), because the subsequent amplifier also introduces noise into the system. A dark current higher than $\sim 1 \ \mu A$ will reduce the signal-to-noise ratio of the ADC and potentially introduce errors in the system. Furthermore, because the photodiodes are normally operated in reverse bias, any leakage current will contribute to standby power consumption. Reducing the dark current of the photodiodes can lead to a higher SNR and a lower power consumption. Furthermore, one can relax other design parameters, like the germanium thickness or size of the device.

For the infrared imaging application, the dark current of the photodiodes directly impacts the sensitivity of the system. State-of-the-art InGaAs devices have demonstrated 18 nA/cm^2 dark current at -1 V at 273 K [9]. To fully compete with these diodes and not lose resolution of the signal, the dark current of germanium devices must be comparable.

This thesis will mainly focus on the mechanisms of dark current in Ge-on-Si photodiodes and methods to reduce it. It is found that the material quality and surface conditions of the germanium contribute greatly to the dark current. Optimizing these characteristics is key to fabricating germanium photodiodes on silicon substrates with low leakage current.

1.2 Basic structure of Ge-on-Si photodiodes

There are many variations of Ge-on-Si photodiodes, but they all share some basic characteristics. Germanium is grown on a silicon substrate, usually without the benefit of any interfacial buffer layer. The junction may be laterally oriented or vertically oriented, but it is always a *pin* junction. The dopants may be either grown *in-situ* or ion implanted after germanium growth. The dark current of the device is the leakage current of this *pin* diode in the dark.

To measure an optical response, the device can either be vertically illuminated or



Figure 1-3: A cross-sectional schematic diagram of a typical Ge-on-Si diode. This diode is fabricated from Ge grown on a thin *i*-Si buffer and is a vertical *pin* junction. From [10].

edge coupled. Optical absorbtion occurs mainly in the intrinsic region of the device, where a high electric field exists. The high electric field moves the carriers out of the intrinsic region and to the contacts where they are collected. An example of a typical Ge-on-Si diode can be found in Figure 1-3.

1.3 Roadmap of this thesis

This thesis primarily discusses the effect of material quality and surface passivation on the dark current of Ge-on-Si photodiodes. Chapter 2 gives a brief overview of photodiode operation and previous work done in this field. Chapter 3 discusses the results from diodes fabricated from selectively-grown Ge-on-Si. It primarily looks at the effect of material quality on dark current density and photoresponse. Chapter 4 discusses results from diodes fabricated from blanket Ge-on-Si films. This chapter is primarily concerned with the effects of surface passivation on the dark current. Chapter 5 discusses the physics and modeling of the blanket Ge-on-Si photodiodes. A model for these devices is developed and experiments were conducted to prove the model. Chapter 6 summarizes the thesis and suggests future work.

Chapter 2

Thesis Background

This chapter discusses the operation of germanium diodes and gives an overview of previous research done on these devices. It will first discuss the operation and basic physics of photodiodes and then it will discuss the advantages and disadvantages of using germanium for photodiode applications. The chapter will then discuss previous work in the field to improve the germanium material quality. Finally, the chapter concludes with a discussion of state-of-the-art Ge-on-Si devices.

2.1 Physics of photodiodes

Semiconductor photodiodes are generally fabricated from standard pn-junctions or pin-junctions, schematics of which can be seen in Figure 2-1. Upon illumination with an energy greater than or equal to the bandgap of the material, electron-hole pairs are generated resulting in current flow.

In an ideal diode, only those minority carriers generated within a diffusion length of the depletion region contribute to the current. These carriers reach the depletion region, are swept across by the high electric field, and reach the other side. Minority carriers generated beyond a diffusion length from the depletion region recombine



Figure 2-1: Schematic illustrations of (a) a pn junction diode and (b) a pin junction diode. Carriers generated in the depletion regions will be collected as photo current. before they reach the high field region. In the pn-junction diode case, the total current in the device can be expressed as [11]:

$$I = I_0(exp(\frac{qV}{kT}) - 1) - qAG_0(L_p + L_n)$$
(2.1)

where the first term is the diode dark current and the second term is the current due to the light. In the above equation, q is the charge of an electron, I_0 is the reverse saturation current, k is Boltzmann's constant, T is the temperature, A is the cross-sectional area, G_0 is the optical carrier generation rate in steady state (assumed to be constant), L_p is the diffusion length for holes, and L_n is the diffusion length for electrons.

The responsivity of a photodiode is defined as

$$R = \frac{I_{photo}}{P_{optical}} \tag{2.2}$$

where I_{photo} is the photogenerated current and $P_{optical}$ is the input optical power. As indicated by the above expressions, the responsivity of a *pn*-junction diode is limited by the diffusion lengths of the carriers. Inserting an intrinsic layer to create a *pin*-junction diode is a common method used to increase responsivity. In a properly designed *pin* diode, the entire intrinsic region is depleted. Any carriers generated in the intrinsic (depleted) region or within a diffusion length on either side will be swept across the junction and collected on the opposite side, assuming negligible recombination of photogenerated carriers in the depletion region itself. Germanium's absorption depth at 1550 nm is on the order of 1 μ m [12], so the intrinsic layer in *pin*-photodiodes is usually ~1 μ m wide in order to provide ample space in which to generate carriers. It should be noted that in the case of relatively high minority carrier lifetimes (10⁻⁶ - 10⁻³ s) and moderately high doping concentrations (10¹⁸ cm⁻³), the diffusion length of minority carriers will be on the order of 1 μ m, indicating that significant photocurrent will originate in the quasineutral regions. As discussed in later sections, the germanium in this work is defected due to threading dislocations that originate during growth. These dislocations may have a deleterious effect on lifetime, reducing the diffusion length of minority carriers in this thesis.

2.2 Optical properties of germanium

Germanium is an excellent candidate for use in *pin* photodiodes that absorb over the visible to near infrared part of the electromagnetic spectrum. While technically germanium is an indirect bandgap material ($E_g = 0.66 \text{ eV}$), it also has a direct bandgap that is only slightly larger (0.80 eV). The bandstructure of Ge can be seen in Figure 2-2

Due to this direct bandgap, Ge absorbs strongly out to ~1.55 μ m (~0.80 eV). An absorption curve of Ge compared to other materials can be seen in Figure 2-3. In the near infrared, germanium's main competitors are InGaAs and other III-V materials that are generally more expensive and harder to integrate onto Si.

It should be noted that some III-V materials, like InGaAs, have a smaller bandgap than Ge, allowing them to absorb light at longer wavelengths. This is an important difference especially in infrared sensor applications and methods to increase the spectrum over which germanium absorbs will be discussed later in section 2.4.2.



Figure 2-2: The band structure of Ge at 300K. The indirect bandgap of 0.66 eV is at the L point while the direct bandgap of 0.8 eV is at the Γ point. From [13].



Figure 2-3: Absorption coefficient versus wavelength for a number of different semiconductors. Ge absorbs strongly out to ~ 1550 nm, making it a strong candidate to compete with materials like InGaAs and InGaAsP. From [12].

2.3 Integration of germanium on silicon

Germanium and SiGe integration on Si has long been studied for metal-oxide-semiconductor field effect transistors (MOSFETs) [14] [15] [16] and heterojunction bipolar transistors (HBT) [17] [18]. SiGe alloys are currently used in state-of-the-art pMOSFETs in the source/drain regions. Since SiGe has a larger lattice constant than Si, when SiGe is grown in these regions, it exerts strain on the channel. The strain in the channel can improve device performance [6]. Furthermore, with its improved mobility over Si, Ge is an attractive candidate for use as a channel material in MOSFETs. Ge MOSFETs, or even strained Ge MOSFETs, are an active area of research which has been producing promising results [19].

Regardless of application, germanium is too expensive to be used as a substrate. A four inch germanium wafer can cost up to a few thousand dollars while a six inch silicon wafer is only ~\$10. For the final product to be cost effective, the germanium must be integrated on a silicon substrate. Integration of Ge on Si substrates is not without its challenges. Due to the 4% lattice mismatch between Ge and Si, when Ge is grown directly on a silicon substrate, it starts to form misfit/threading defects quickly. The critical thickess, or the thickness of film that can be deposited before defects start to form, of Ge on Si is ~2 nm [20], although researchers are looking at ways to increase it [21]. For a MOSFET application, a few nanometers of germanium may be all that is needed. For a photodiode, however, much thicker Ge is needed to absorb light. Depositing 1 - 2 μ m of germanium directly on silicon results in films with 10⁸ - 10⁹ defects per square centimeter.

Researchers have been studying ways in which to reduce the threading defects in thick $(1 - 2 \ \mu m)$ Ge films grown on silicon substrates. One possibility is to grow the Ge on a relaxed graded buffer layer. In this method, prior to growing pure Ge, a SiGe alloy is grown, which gradually increases the concentration of germanium. The goal is to confine all of the threading defects to this buffer layer, leaving the top layer of

Ge with virtually no defects. Fitzgerald, *et al.* have done this and achieved a defect density of $2 \ge 10^6$ cm⁻² [22].

The difficulty with growing a SiGe buffer layer is that it takes both time and resources. The buffer layer is usually 5 - 7 μ m, while the top Ge layer is often only 1 - 2 μ m. This thick buffer layer is needed to fully relax the Si_xGe_{1-x} alloy and effectively "trap" the threading dislocations in the buffer layer. Thus, much of the growth time is actually spent growing the buffer layer and not the Ge. Furthermore, some applications are not compatible with photodiodes grown on a thick buffer layer. An alternative is to grow the Ge on a one-step buffer, *i.e.* grow a thin layer of Si_{0.8}Ge_{0.2} [23].

Another possibility for reducing the threading defect density is to anneal the wafer after growth. This is usually done in a cyclic fashion by alternating between a high temperature ($\sim 800^{\circ}$ C) and a low temperature ($\sim 400^{\circ}$ C) multiple times. By annealing the film, the defects move, annihilating one another or making their way to the edge of the wafer. Cyclic annealing has been studied in great detail and it was found that it is possible to reduce the defect density by $\sim 10X$ [24] [25].

Finally, the defect density can be reduced by growing the germanium selectively in small areas. In this case, oxide is first deposited on a silicon wafer and patterned into holes. Growing Ge selectively in the holes benefits the quality of the Ge film in two ways. First, defects generated on one area of the wafer cannot propagate to another area of the wafer because the oxide is in the way. This is called the small area effect and is illustrated schematically in Figure 2-4.

Furthermore, when defects do form at the Ge/Si interface, they generally propagate towards the Ge surface at an angle (rather than straight up). If the height of the oxide is tall enough, the defect will be annihilated when it reaches the oxide sidewall and it will never reach the top of the film. This is called aspect ratio trapping [27] and is illustrated in Figure 2.3. By combining both selective growth and annealing,



Figure 2-4: A schematic illustration of how dislocations propagate in (a) a large area and (b) a small area. In a large area, the defects can propagate a long way, potentially impacting a large portion of the film. In a small area, defects from one square cannot propagate into a neighboring square, thus reducing the defect density. From [26].

researchers have achieved very high quality germanium grown directly on silicon [25]. Figure 2-6 shows Nomarski micrographs of 10 x 10 μ m Ge islands after a 10 minute anneal and after a cyclic anneal. After a 10 cycle anneal, the film had 2.3 x 10⁶ cm² defects. While still somewhat high, this value is ~10X lower than similarly annealed blanket films. Furthermore, some Ge islands were defect free after the anneal.

2.3.1 Germanium growth

Germanium can be grown on silicon substrates in a number of different ways. Researchers have used molecular beam epitaxy (MBE) [29] [10], ultrahigh vacuum chemical vapor deposition (UHVCVD) [30] [31], reduced pressure chemical vapor deposition (RPCVD) [32] [33], and low energy plasma enhance chemical vapor deposition (LEP-ECVD) [34].

Regardless of the growth system, most germanium is grown using a two-step



Figure 2-5: A schematic illustration of aspect ratio trapping. The defects terminate at the oxide sidewalls leaving the top of the film virtually defect free. For materials with crystal orientations like the one in the figure, the angle that the defects form with the substrate is 45°. This is because the threading dislocations will propagate in the $\langle 110 \rangle$ direction on the (111) plane. From [28].



Figure 2-6: Nomarski micrographs of $10 \ge 10 mu$ m Ge islands grown selectively. The dimples in the Ge film were caused by an iodine etch as a way to highlight the defects. (a) With a 900°C anneal for 10 minutes, the defect density was $4.3 \ge 10^7 \text{ cm}^2$. (b) After a 10X anneal between 900°C and 100°C, the defect density was $2.3 \ge 10^6 \text{ cm}^2$. From [25].

growth process. The first step is generally done at low temperature (300°C - 400°C). The low temperature is necessary to create a smooth germanium film and to suppress islanding. After this seed layer is deposited, the temperature is increased to the range of 600°C to 750°C. This high temperature promotes faster germanium deposition while maintaining a smooth film. Finally, there is sometimes a post-growth anneal in a hydrogen ambient to reduce the threading defects.

2.4 Ge-on-Si diodes

2.4.1 Figures of merit for photodiodes

There are three typical figures of merit for photodiodes. Their relative importance differs in different applications. The first is dark current. Dark current is simply the leakage current when the diodes are in the dark. A low dark current is important to keep the signal-to-noise (SNR) ratio high. A low dark current is of the utmost importance for infrared imaging applications because it is the dominant noise in the system. For high-speed applications, low dark current is not the most important characteristic, but by reducing the dark current, standby power consumption and other design paramters can be relaxed. The dark current, and factors that affect it, will be discussed in detail in Chapter 5. Comparing the dark current of different devices is not always straightforward. Some groups only quote the dark current does not always scale with the area of the device. This thesis attempts to quote both the absolute dark current and the size of devices fabricated by other groups. However, sometimes this data is not available and the dark current density is quoted instead.

The second figure of merit is the photoresponse, or responsivity, of the devices. This is a measure of how well the diode absorbs light, converts it to electron-hole pairs, and extracts the generated carriers from the device. Having a high responsivity is also important for a high SNR.

The final figure of merit is the frequency response of the devices. It is important for these devices to switch on and off rapidly, especially for high-speed applications like the analog-to-digital converter mentioned in Chapter 1. For imaging applications, the frequency response of the device is not very important and is often ignored. The frequency response is often limited by one of two things. It can be limited by the RC time constant of the overall device or it can be limited by the transit time of the carriers across the device. If the frequency response is dominated by the transit time, it is important to keep the intrinsic region narrow so that the carriers have a short distance to travel. However, making this region narrow can mean that there is not as much material for absorbing light, resulting in lower SNR ratios. For this reason, it is important for high speed devices to still have a low dark current.

2.4.2 Previous work on blanket diodes

As mentioned earlier, the simplest way to fabricate a Ge-on-Si photodiode is to grow Ge directly on Si without a graded buffer layer. When the germanium is grown across the entire wafer, it is referred to as a "blanket film." Depending on the application, germanium is grown using different growth tools. Using MBE allows tight control over doping profiles, resulting in high internal electric fields. Oehme, *et al.* have used MBE to fabricate a Ge-on-Si diode with a 39 GHz bandwidth at -2 V. At -1 V, the dark current of a 5 μ m radius circular device was 75 nA (95 mA/cm²) [35].

Germanium is also often grown using ultra-high vacuum chemical vapor deposition (UHV-CVD). Colace, *et al.* achieved dark current densities of $\sim 250 \text{ mA/cm}^2$ (16 *mu*A for an 80 x 80 μ m device) and a bandwidth of 10 Gbit/s. Furthermore, the responsivity of their devices at 1550 nm was 0.2 A/W [31]. Liu *et al.* achieved dark current densities of 10 mA/cm² while increasing the responsivity to 0.52 A/W [36].

More recently, reduced-pressure chemical vapor deposition (RPCVD) has been

used as a cost-effective substitute for MBE or UHVCVD to grow germaium films. Osmond *et al.* have reported a 10 μ m device in diameter with a dark current of 225 nA (~300 mA/cm²) at -1 V. Furthermore, these devices had a responsivity of 0.21 A/W at 1500 nm and a frequency response of 49 GHz when biased at -5 V [37].

Kim, et al. have used RPCVD to fabricate devices that operate at 36 GHz at -1 V [33]. These devices used germanium that was 1.2 μ m thick. Devices with 1.7 μ m of Ge that had a diameter of 17 μ m displayed dark current of 42 nA, or 19 mA/cm². Thicker devices had a frequency response of 13 GHz at -1 V and a responsivity of 0.74 A/W at 1550 nm.

Finally, a new growth method called low-energy plasma-enhanced chemical vapor deposition (LEPECVD) had been used to deposit germanium films. Comparable levels of threading defects to UHVCVD and MBE growth were reported with anneals ($\sim 2 \times 10^7 \text{ cm}^{-2}$). When fabricating large area devices (radius $\sim 1 \text{ mm}$), they report dark current densities as low as 1 μ A/cm² [34].

When comparing leakage current densities of diodes fabricated Ge-on-Si films, it is often useful to compare them to leakage current densities on bulk Ge wafers. Ge-on-Si films will have some number of threading defects (10^6 cm² or more), but a bulk Ge wafer should be virtually defect free. Diodes fabricated on bulk Ge wafers are most often *pn* structures rather than *pin* because they are used to study source/drain regions of Ge MOSFETs. This different structure makes a direct comparison difficult; however, work has been done to minimize damage due to ion implantation and reduce surface effects.

Takagi, *et al.* used a process known as gas phase doping to eliminate ion implantation entirely. After annealing at 500°C - 600°C, they achieved an *n*-type activation of $\sim 10^{19}$ cm⁻³. Diodes fabricated from these junctions displayed leakage current densities of ~ 1 mA/cm² at -1 V [38]. Furthermore, Woo, *et al.* did a study on implantation energy, dose, and annealing conditions to minimize the leakage current. They found that the most important aspect of junction design was keeping the junction as far from the surface as possible. In their study, a high implant energy (100 KeV) produced devices with the lowest leakage current density, despite the larger amount of damage caused by the implant. At -1 V, these devices had a leakage current density of ~0.1 mA/cm². They also found that it was possible to modify the activation anneal conditions to repair the crystal damage done by the implantation [39].

In addition to focusing on the dark current, responsivity, and frequency response of devices, some groups have been investigating ways to extend the wavelength range over which germanium absorbs. To do this, they add strain to the device which shrinks the bandgap and allows germanium to absorb at longer wavelengths.

After growth at high temperature, there is residual strain in the Ge films. This strain is not due to the lattice mismatch as in thinner films, but rather due to the difference in thermal expansion coefficients. As the germanium and silicon layers cool to room temperature, they contract at different rates. This leaves a residual tensile strain in the film. This was first documented by Kimerling, *et al.* [40]. A plot of germanium's absorption coefficient versus energy can be seen in Figure 2-7. For germanium grown on silicon, the direct bandgap shrinks from 0.80 eV to 0.77 eV.

Other groups have tried to add additional strain to the germanium. Liu *et al.* added a backside silicide to the wafer after processing. The silicide induced a slight bowing in the wafer, increasing the tensile strain in the Ge from 0.20% to 0.25% [36].

Saraswat, et al. characterized germanium photodetectors when applying external strain using a four point bending setup. They found that the photoresponse at 1550 nm increased from 0.67 A/W to 0.75 A/W with 113 MPa of tensile stress and decreased from 0.67 A/W to 0.48 A/W with 177 MPa of compressive stress [41]. This is consistent with the theoretical calculations on the deformanation potentials reported in [42].



Figure 2-7: Plot of absorption coefficient versus energy for bulk Ge and Ge-on-Si. The difference in thermal expansion coefficients between Ge and Si leaves a residual tensile strain in the Ge after high temperature growth. The result is that Ge-on-Si will absorb light of lower energy than bulk Ge. From [40].

2.4.3 Previous work on selective diodes

As mentioned earlier, one way to reduce the threading defects in germanium films grown on silicon substrates is to grow them selectively. As shown by Kimerling *et al.*, the threading defect density has a direct impact on the dark current of Si_{0.75}Ge_{0.25} diodes grown on buffers of different grades [43]. By growing germanium selectively, it is possible to reduce the defect density and improve the dark current density. Yu, *et al.* reported on selectively-grown films with a defect density of $1 \ge 10^7$ cm². Large diodes (150 μ m radius) fabricated from this germanium displayed a dark current of ~6 μ A (dark current density of ~8 mA/cm²) at -1 V. The responsivity of these devices was 0.64 A/W at -1 V and 1550 nm [44].

Aberg, et al. have devised a new way to grow germanium selectively. Rather than patterning somewhat large (~10 μ m) oxide holes, they pattern a very small (~100 nm) window with a high aspect ratio. Germanium is grown in this seed region to a height of ~1 μ m. Over that space, nearly all of the dislocations terminate at the



Figure 2-8: A cross-sectional SEM of selectively-grown Ge on Si utilizing a high aspect ratio seed region. Nearly all of the defects terminate before they reach the higher region where the device is located. From [45].

oxide sidewalls. Then the window widens to $\sim 5 \ \mu m$, which is a large enough area to fabricate an infrared imager [45]. A cross-sectional scanning electron micrograph (SEM) can be seen in Figure 2-8.

At room temperature, devices fabricated from these germanium films demonstrated ~30 pA of dark current at -1 V. For a device size of 10 x 10 μ m, this corresponds to a dark current density of ~30 μ A/cm².

Loh, et al. have been fabricating germanium photodiodes from selectively grown UHVCVD. In an experiment where they included a thin $Si_{0.8}Ge_{0.2}$ buffer layer, they achieved dark current densities on the order of ~20 mA/cm² (~170 nA for a 25 x 25 μ m square). Furthermore, leakage current analysis indicated that the bulk dark current density is ~2 mA/cm². These devices also displayed a 3 dB frequency of 15 GHz at -1 V [46]. The same group has also investigated adding strain to these selective devices by growing the germanium on different buffer layers. They report an increase in photoresponse from 0.1 to 0.2 A/W at 1520 nm and -1 V [23].

A summary table of the dark current and dark current densities for devices fabricated from blanket and selective Ge-on-Si can be found in Table 2.1. The values of the dark current density varies widely from $1 \ \mu A/cm^2$ to 300 mA/cm² and could be due to a number of parameters. Some groups were more interested in fabricating a device that operated at high speed and, thus, did not investigate the sources of leakage current very much. The dark current is somewhat dependent on the threading defects in the germanium film. The concentration of threading defects is due to a number of processing conditions such as the anneal conditions, the thickness of the germanium, and whether or not the germanium was grown on a buffer layer. Unfortunately, very few of the references quantify the dislocation density in their germanium films. Finally, as will be discussed in Chapter 5, the dark current is also dependent on the doping profiles in the device. There will be variations in the doping profile due to the implant conditions, the activation anneal thermal budget, and the growth method used. This background is meant to be an overview of the field. Mechanisms responsible for the dark current will be discussed in Chapter 5.

Author and ref-	Size of device	Dark current at -1 V	Dark current
erence		(nA)	density at -1 V
			(mA/cm^2)
Oehme [35]	$5 \ \mu m$ radius	75	95
Colace [31]	$80 \ge 80 \mu m$	16000	250
Liu [36]			10
Osmond [37]	$5 \ \mu m$ radius	225	300
Kim [33]	$8.5 \ \mu m$ radius	42	19
Osmond [34]			0.001
Yu [44]	150 mum radius	6000	8
Aberg [45]	$10 \ge 10 \ \mu m$	0.03	0.03
Loh [46]	$25 \ge 25 \ \mu m$	170	20

Table 2.1: Summary of previous work on dark current

2.4.4 Waveguide integrated Ge diodes

To move towards full integration of a germanium detector in an integrated photonic system, many groups have fabricated Ge diodes and Si waveguides on a single chip. In these systems, germanium detectors can either be butt-coupled or evanescently coupled to the silicon waveguide. The layouts of each are illustrated in Figure 2-9. In addition to variation of coupling schemes, the devices may either be oriented with



Figure 2-9: Schematic diagrams of two types of waveguide integrated detectors. (a) An evanescently coupled Ge detector grown on top of a Si waveguide. (b) A butt coupled detector grown where Si has been etched away.



Figure 2-10: A schematic illustration of (a) a lateral *pin* structure and (b) a vertical *pin* structure. Each offers its own benefits and integration challenges.

a vertical *pin* structure or a lateral one. This difference is illustrated in Figure 2-10 and the implications will be discussed later in the section.

Feng, et al. and Vivien, et al. have both fabricated butt-coupled Ge detectors with silicon waveguides [47] [48]. They achieved 3 dB frequencies of 32 GHz and 42 GHz, respectively, and report responsivities of ≥ 1 A/W. The devices in Feng had a much higher dark current density of ~16 A/cm² while Vivien achieved 60 mA/cm². The dark current density in [48] was given explicitly in the paper and the dark current density of [47] was calculated from the absolute dark current and active area of the device, both of which were given explicitly in the paper.

Yin, et al. [49] and Feng, et al. [50] have both fabricated vertical pin junctions
that are evanescently coupled to a Si waveguide. The dark current densities were comparable (60 mA/cm^2 and 30 mA/cm^2 respectively), but the devices in [50] displayed a 3 dB frequency of 12 GHz while [49] operated at 29.4 GHz. Neither group is forthcoming with their growth system or details, so it is difficult to say whether these differences are due to the quality of the Ge or the device architecture itself.

Wang, et al. have fabricated a similar device, except the junction was a lateral pin structure evanescently coupled to a Si waveguide. They report a 3 dB frequency of 18 GHz and an absolute dark current as low as 15 nA at -1 V [51]. Wang also compared both laterial pin and vertical pin structures which were evanescently coupled on a silicon waveguide. They found that the vertical diode had a higher bandwidth (5.5 GHz compared to 3.4 GHz) but a lower responsivity (0.29 A/W compared to 1.16 A/W). They attributed the lower responsivity of the vertical device to a smaller optical mode overlap due to the highly doped Ge region at the diode/waveguide interface [52].

From a statistical perspective, Kopp, *et al.* achieved a yield of 99% for germanium detectors integrated in a CMOS-compatible process flow on a 200 mm wafer, thus demonstrating the overall feasibility of germanium diodes integrated in a CMOS process. They achieved a dark current density of 7 mA/cm^2 and a median bandwidth of 9 GHz over 1100 photodiodes [53].

2.4.5 Other types of Ge-on-Si diodes

There are also other forms of Ge-on-Si diodes that have been fabricated. Metalsemiconductor-metal structures have high dark current but can operate at high speeds (25 GHz) [54]. Diodes have also been fabricated from superlattice structures [55] and from avalanche structures [56]. Futhermore, the germanium film is not always grown using some sort of chemical vapor deposition or epitaxy. Waveguide integrated Ge detectors have been fabricated by rapid melt growth of the Ge [57], by bonding of a Ge wafer to a Si wafer [58], and by depositing poly-crystalline Ge by thermal evaporation [59].

2.5 Scope of this thesis and chapter summary

Previous work has focused primarily on germanium integration on silicon substrates and on fabricating devices with a high 3 dB frequency. This thesis focuses on reducing the leakage current in small devices. When device dimensions are on the order of ~100 μ m, perimeter or surface effects on the dark current are dwarfed by the leakage current of the active area. However, when devices are scaled down, the dark current does not usually scale proportional to the area. By understanding the physics of the dark current, it may be possible to make devices that are small (~10 μ m) with low leakage current.

This chapter outlined the operation of germanium photodiodes and discussed methods which have been used to integrate Ge photodiodes on silicon substrates. It discussed three ways of reducing the threading defects that arise from the lattice mismatch: growing germanium on a SiGe buffer, using a high temperature anneal after growth, or growing germanium in small areas. It then discussed previous work on germanium photodiodes fabricated from blanket films and from selective films and it also discussed how tensile strain affects the photoresponse of devices. The dark current and dark current density of previous devices varies widely and its origin is not well understood. This thesis attempts to study and explain the dark current mechanisms in small Ge-on-Si photodiodes. Finally, this chapter discussed waveguide-integrated device architectures and results. Full integration of these diodes in an electronicphotonic system is a major goal of the field and this thesis will discuss attempts at MIT to integrate a germanium photodiode with a silicon waveguide.

Chapter 3

Photodiodes fabricated using selectively grown germanium

This chapter discusses the fabrication and characterization of photodiodes made from selectively-grown germanium. There are numerous benefits from using selectivelygrown germanium. It enables integration with larger and more complex electronicphotonic systems and increases the germanium growth rate. Furthermore, there is also the possibility to grow germanium films with fewer defects as compared to a blanket film. In this thesis, we pursue the integration of a germanium photodiode into an electronic-photonic analog-to-digital converter.

This chapter first describes the growth of germanium on silicon wafers in oxide holes. Next, it outlines the fabrication of diodes made from these films. It then describes the electrical and optical characteristics of stand-alone test devices. Finally, it describes the fabrication and characterization of diodes integrated on top of a silicon waveguide.

3.1 Selective germanium growth

Growing germanium selectively in oxide windows presents a series of challenges that are different from blanket Ge growth. Optimum growth parameters must be found so that the germanium fills the oxide windows completely and uniformly. Furthermore, different planes of atoms grow at different rates, which leads to faceting. Care must be taken to reduce faceting on the germanium surface. Finally, and most importantly, the surface of the Si wafer must be extremely clean. A contaminated surface will cause the germanium film to have many threading dislocations or even be polycrystalline instead of single crystalline, resulting in fabrication difficulties and poor device performance.

3.1.1 Surface preparation

For blanket germanium growth a virgin wafer is typically cleaned and then placed into an epitaxial reactor for the growth itself. The wafer sees very minimal processing prior to growth (typically only a wet chemical clean) and a high temperature ($\sim 1080^{\circ}$ C) pre-bake is used to drive off any contaminants from the surface. This results in a very clean Si surface and a high quality Ge film. For selective growth, however, the wafer must go through more processing prior to growth, potentially leaving the surface contaminated.

For this work, the wafers were first cleaned using a standard RCA clean [60]. Then 1 - 2 μ m of a PE-CVD oxide was deposited. The oxide was patterned with photolithography and then etched using a C₄F₈:CH₂F₂ 1:1 plasma. This dry etch was intended to produce vertical sidewalls. The photoresist was then stripped using an oxygen plasma and the wafers received a double piranha clean. Finally, the wafers received an RCA clean with two short HF dips to remove any oxide in the windows while minimizing oxide loss in the field. Immediately before growth, the wafers received a 5 minute pre-bake in the growth chamber at 900°C to drive off any remaining



Figure 3-1: Nomarski micrographs polycrystalline Ge and single crystalline Ge. (a) A contaminated Si surface caused the germanium to grow in a poly-crystalline film rather than single crystalline. The poly-crystalline Ge can be seen in the dark, sandy parts of the image. In the clear parts of the image, the Ge growth was inconsistent. (b) A sacrificial oxidation step cleaned the Si surface and resulted in crystalline Ge growth. The square in the image shows a smooth Ge film.

surface contaminants.

This process resulted in a silicon surface that was still not clean enough for highquality Ge growth. The germanium grown on this surface was virtually always polycrystalline, as seen in Figure 3-1(a). It was hypothesized that the etch chemistry of the oxide etch left fluorocarbons on the surface of the silicon, which could not be removed with basic wet cleans [61].

To ensure a clean Si surface, a sacrificial oxidation step was used [62]. After the piranha cleans above, a thermal oxidation was done on the wafers at 800°C. 5 - 10 nm of oxide was grown and then subsequently stripped off in a 10:1 BOE bath. The wafers then received an RCA clean and a 900°C pre-bake. This reliably left the surface of the wafer clean and resulted in single crystalline germanium growth, as seen in Figure 3-1(b). Figure 3-1 shows images of germanium grown using identical growth parameters but the two different cleaning processes. Table 3.1 outlines the differences in surface preparation steps. Process B was used on all of the subsequent wafers.

After the surface preparation, germanium was grown using a general two-step method. The first step was done at low temperature (365°C) and 60 T chamber pressure to ensure a smooth Ge film. Next, the temperature was increased to 750°C to increase the growth rate while maintaining high quality Ge. Details of the temperature and pressure for this final step can be found in the next section.

Table 5.1. Surface preparation procedures			
Process A	Process B		
Pattern field oxide	Pattern field oxide		
Dry etch in C_4F_8/CH_2F_2 plasma	Dry etch in C_4F_8/CH_2F_2 plasma		
Wet etch in BOE	Wet etch in BOE		
Double piranha with HF dip	Double piranha with HF dip		
RCA clean with short HF dip	RCA clean with short HF dip		
-	Thermal oxidation		
	Oxide removal in BOE		
	RCA clean with short HF dip		
Pre-bake at 900°C	Pre-bake at 900°C		
Ge growth	Ge growth		

Table 3.1: Surface preparation procedures

3.1.2 Hole filling and faceting

Growth techniques for the high temperature step were studied by Kim to optimize hole filling and germanium faceting [26] [63]. Details can be found elsewhere [63] and a summary is provided here. Kim demonstrated that increasing the Ge growth temperature greatly reduced the $\{311\}$ faceting, leading to better hole filling behavior. Increasing the germane partial pressure also reduces the faceting. Furthermore, low pressure is needed to ensure good selectivity of germanium on silicon (rather than deposition on the SiO₂). It was found that growing Ge at 750°C with a germane partial pressure of 133 Pa and a chamber pressure of 10 T yielded germanium which filled the oxide holes the best while maintaining good selectivity to silicon over oxide.

3.1.3 Growth parameters and threading dislocations

The final growth parameters of 750°C, 10 T, and 133 Pa of germane partial pressure resulted in germanium that filled the oxide holes well and minimized faceting [63]. The threading dislocation density of these films was found to vary with window size [63]. It was found that a cyclic anneal can reduce the defect density of a large area, $0.8 \ \mu\text{m}$ -thick film from 10 x 10^8 cm^{-2} for an as-grown sample to $1.2 \text{ x} 10^8 \text{ cm}^{-2}$ for a film annealed four times. Furthermore, the defects in an an unannealed, $1 \ \mu\text{m}$ -thick film decrease with the size of the window opening. A 5 μ m-wide opening had a defect density of 5.5 x 10^8 cm^{-2} . This decreased to 1.7 x 10^8 cm^{-2} for a 0.65 μ m-wide opening. A 0.65 μ m-wide opening with 1 μ m of Ge that was cyclically annealed four times showed very few defects. This number was not quantified because the sparseness of the defects led to low confidence in the measurement.

3.2 Fabrication of photodiodes

Photodiodes were fabricated from the selectively-grown Ge films. The fabrication was a joint effort between MIT Lincoln Laboratory (MITLL) and MIT's Microsystems Technology Laboratory (MTL). The surface preparation of p+ Si wafers, including sacrifical oxidation, was all done at MITLL. The germanium growth was done at MTL as part of this thesis using the two step method described previously. During the growth of the low-temperature seed, the germanium was optionally doped p-type with B₂H₆. This doping helps to reduce the impact of the dislocations that are present in the first 200 nm of the Ge film by incorporating them into the p-type contact. It also helps to reduce the series resistance of the device by reducing the impact of the valence band offset between the Ge and the Si on the current conduction [64]. After growth, the samples went back to MITLL for the final processing into photodiodes.

After growth, a chemical-mechanical polishing (CMP) step was used to planaraize



Figure 3-2: Spreading resistance profiling of three different activation anneal conditions. The 600°C anneal for 5 seconds was chosen both to maximize the active phosphorus concentration while minimizing the diffusion into the Ge. The implant was done at 25 keV with a dose of $1 \ge 10^{15}$ cm⁻². Courtesy of Solecon Laboratories.

the Ge surface. Due to the sensitivity of germanium to typical CMP slurries [65, 66], this step was carried out using only water. After CMP, the wafers were implanted with phosphorus at an energy of 25 keV and a dose of $1 \ge 10^{15}$ cm⁻² through a ~10 nm screen oxide. The dopants were then activated using a rapid thermal anneal at 600°C for 5 seconds to form a vertical *pin* junction. Spreading resistance profiling was performed by Solecon Laboratories on three different annealing conditions, as seen in Figure 5-9. The 600°C anneal for 5 seconds was found to have the highest overall activation of phosphorus of 3 $\ge 10^{19}$ cm⁻³. It also resulted in the steepest profile, which yielded the highest internal electric field in the device.

After activation, 90 nm of PE-CVD oxide was deposited. Contact vias were etched using a combination of dry and wet etches. Finally, metal was deposited and patterned into contacts.

3.3 Electrical characterization

As mentioned earlier, low dark current is an important figure of merit for these devices. The crystal orientation of the device, the inclusion of a post-growth cyclic anneal, and whether or not the junction intersected the sidewall all impacted the dark current.

3.3.1 Effect of cyclic annealing on dark current

After the high temperature growth step, some of the devices received a 4X cyclic anneal between 800°C and 450°C for times of 150 seconds and 90 seconds respectively. The anneal served to reduce the dislocation density as mentioned previously. Figure 3-3 compares a device that received the anneal and one that did not. In both cases, the germanium islands were 5 x 10 μ m with n+ regions that measured 1 x 4 μ m. The cyclic anneal reduced the dark current at -1 V by ~5X. This is consistent with the reduction in dislocation density in the Ge film. Furthermore, the anneal increases the forward current by reducing the series resistance. This will reduce the RC time constant of the device which is important for high-speed applications. The frequency response of these diodes is discussed in appendix D.

The reduction in series resistance could be due to two factors. First, the reduction in dislocation density could improve the series resistance by reducing the recombination of carriers in the n+ contact region, allowing those carriers to be extracted from the device more efficiently. Secondly, the anneal may also cause outdiffusion of the boron from the Si wafer into the Ge film. This autodoping of the Ge film serves to reduce the impact of the valence band offset between the Si and the Ge and reduce the series resistance [64].



Figure 3-3: Current versus voltage characteristic illustrating the effect of a cyclic anneal on the dark current of 5 x 10 μ m photodiodes. The cyclic anneal reduces the threading dislocation density, resulting in lower dark current and high forward current. The annealed wafer was #5223 and the unannealed wafer was #5222.

3.3.2 Effect of orientation on dark current

The germanium films were grown in windows that were etched both parallel to the wafer flat, the $\langle 110 \rangle$ direction, and at a 45° angle to the flat, the $\langle 100 \rangle$ direction. Previous studies have shown that the orientation of the window has an effect on the dark current, with windows with $\langle 100 \rangle$ -oriented sidewalls having lower leakage currents [67].

In this study, it was also found that the orientation of the window has an effect on the yield of the devices. This is illustrated in Figure 3-4. Eight devices were measured for each orientation. For the $\langle 110 \rangle$ direction, half of the devices were very leaky, while one had a steep slope in reverse bias, and three rectified well. For the $\langle 100 \rangle$ direction, one device was very leaky, while the remaining seven rectified well. It is possible that this effect is a result of processing conditions which impacted device yield. No attempt was made to reproduce the results to check this conjecture.

3.3.3 Layout of the n + region

The intersection of the n+ region with the oxide sidewall also had a significant impact on diode performance. Figure 3-5 shows IV characteristics for two 100 x 100 μ m square devices. In one device, the dopants were implanted to completely cover the top of the film. In the other device, the dopants were implanted 5 μ m away from the sidewall. As the figure shows, when the dopants intersect the sidewall of the device, the diode no longer rectifies. This suggests that the oxide sidewall significantly degrades the junction. Note that in the devices discussed in Section 3.3.2, the n+region was kept away from the window edge.



Figure 3-4: IV curves illustrating the effect of the orientation of the Ge film on the dark current of devices. Devices made in the windows in the $\langle 110 \rangle$ direction (a) had a much lower yield than devices made in windows in the $\langle 100 \rangle$ direction (b). The devices in this figure are all 5 x 10 μ m Ge islands with 1 x 4 μ m n+ regions. They are all unannealed. Curves represent eight different devices of the same size all from wafer #5222.

3.3.4 Lowest dark current

The lowest dark current measured on these samples was from one of the smallest devices fabricated. Unfortunately, due to its size, the yield was low for this type of device. The device is 0.35 x 10 μ m large with an n+ region of 0.25 x 5 μ m. The device was fabricated in the $\langle 100 \rangle$ direction on an unannealed wafer (fewer of the annealed devices survived processing) and the IV curve is shown in Figure 3-6. The dark current of this device was 0.5 nA at -1 V and the dark current density was 40 mA/cm².

3.3.5 Size dependence of dark current densities

As mentioned earlier, as the size of the device shrinks, the quality of the germnaium film should improve. Dark current density can be used as a measure of film quality.



Figure 3-5: IV curves from diode with different *n*-type layouts. When the junction intersects the sidewall of the device, the leakage current is very high. These devices were all from wafer #5222.



Figure 3-6: IV curve from the diode with the lowest dark current. The Ge island is $0.35 \ge 10 \ \mu\text{m}$ with an n+ region of $0.25 \ge 5 \ \mu\text{m}$. The dark current at -1 V is 0.5 nA. This device was fabricated in germanium that was grown in windows aligned along the $\langle 100 \rangle$ direction. The wafer (#5222) did not receive an anneal.

The lowest dark current densities reported to date have been $\sim 0.1 \text{ mA/cm}^2$ [22] [45]. In both cases, the defect density is expected to be low. In [22], the graded buffer layer increases the germanium concentration gradually so that threading dislocation density is minimized. In [45], the germanium is grown selectively, first in a very high aspect ratio hole and then in a larger region. Most of the defects terminate in the high aspect ratio region, resulting in a very high quality film at the top of the germanium.

Table 3.2: Size dependence of dark current density

Oxide window	n + region size	Dark current at -	Dark current
size		1 V	density at -1 V $$
$0.35 \ge 10 \ \mu \mathrm{m}$	$0.25 \ge 10 \ \mu \mathrm{m}$	0.5 nA	40 mA/cm^{-2}
$5 \ge 10 \ \mu m$	$1 \ge 4 \ \mu \mathrm{m}$	2 nA	50 mA/cm^{-2}
100 x 100 $\mu \mathrm{m}$	$4(9 \ge 89 \ \mu m)$	$2~\mu { m A}$	62 mA/cm^{-2}

The dark current densities of some different size devices in this study can be found in Table 3.2. The dark current density was calculated by dividing the absolute current by the area of the n+ region, not the area of the oxide window size, because the n+region defines the active area of the device. As the device size shrinks, the dark current density improves. This is most likely due to the reduction in threading dislocations in the germanium film for smaller devices. However, as described earlier, the decrease in window size of over 2 orders of magnitude should result in a large improvement in threading dislocation density. The dark current density, while improved, does not reflect this expected improvement in material quality, showing an improvement of ~1.5X. This indicates that something else is dominating the dark current. The sources of dark current are analyzed further in Chapter 5 and it is likely that the dark current is limited by generation in the depletion region due to another trap source.

3.4 Optical characterization

Photoresponse measurements were done on larger devices. A broadband laser (1260 nm - 1630 nm) was used as the light source and was coupled in through a fiber. The details of the measurement can be found in Appendix B. The current was measured for three different input power levels and responsivity was calculated as an average of the three points. This section will discuss the effect of the germanium film thickness and the effect of the size of the device on the photoresponse

3.4.1 Thickness of Ge film

Figure 3-7 shows the responsivity versus wavelength curves for devices with 1 μ m thick Ge and 2 μ m thick Ge. It is expected that a thicker Ge film will absorb more light; however, since the germanium absorption coefficient is an exponential curve with respect to wavelength, it is not expected that a doubling in film thickness should double the photoresponse. A doubling of the film thickness should result in a photoresponse that increases by less than 2X. Figure 3-7 shows the photoresponse increase by ~2X, suggesting that the film quality is improving with thickness. This improvement was found in [63] and is likely leading to the improvement in photoresponse.

3.4.2 Different size Ge islands

It was found that the photoresponse also varies based on the size of the Ge island. Figure 3-8 shows normalized photoresponse curves for three different size Ge islands. These devices had different absolute responsivities. To normalize them, the first 5 values (at 1300, 1305, 1310, 1315, and 1320 nm) of the responsivity were averaged. Then each point was divided by that average and the result was plotted.

The figure shows that the absorption edge of the germanium shifts to longer wavelengths as the device size grows. The total bandgap shift is ~ 25 meV. This



Figure 3-7: Responsivity curve for two different Ge thicknesses. The responsivity is nearly double for a film that is nearly double the thickness. Since most of the absorption of light happens in the top of the film, this suggests that the film quality is improving with thickness. This data of the 2 μ m film was taken from wafer #5365 and the data from the 1 μ m film was taken from wafer #5363.



Figure 3-8: Normalized responsivity curves for three different Ge island sizes. The absorption edge is shifted out to longer wavelengths for the larger devices, suggesting a difference in the strain. This data was measured on wafer #5223.

suggests that there is more strain in the larger devices.

Raman spectroscopy was performed at Freescale Semiconductor to confirm the differences in strain. The Raman analysis was done with a laser of 514.5 nm and the Ge peaks were compared to Ge peaks for a bulk Ge wafer. The strain is summarized in Figure 3-9.

The figure shows that, as the device size grows, the tensile strain increases. This tensile strain shrinks the bandgap and allows the Ge to absorb at longer wavelengths. It is important to note that the strain in these devices is not due to the lattice mismatch between the Si and the Ge (as is the case for thin Ge films). This strain is caused by the difference in thermal expansion coefficients between the Si and Ge. Immediately after Ge growth, the wafers are annealed at a high temperature. When



Figure 3-9: Residual tensile strain in three different Ge island sizes. The tensile strain is largest for the largest device, pushing the responsivity curve out to longer wavelengths. The analysis was done on wafer #5223.

they cool to room temperature, the Ge and Si contract by different amounts, leaving some residual strain. It should be noted that, according to the Raman data, the smallest device is under slight compressive strain. This compressive strain is expected to widen the bandgap [40] and is consistent with the responsivity data.

The strain in the 100 μ m device is comparable to the strain found in blanket Ge epitaxial films [40]. As the device size shrinks, the strain decreases. This size dependence of strain has been observed previously [68] and can be explained by the faceting in the Ge film. When the Ge grows in oxide windows, the films grow with facets around the sides. These facets are normally 1 - 2 μ m wide and serve to relax the strain in the film. For a large (100 μ m) device, the facets are a small portion of the overall surface area, so much of the strain is retained. For smaller devices, however, the facets are actually a large fraction of the surface area and much of the strain is relaxed. A cross-sectional SEM of a Ge film grown in this work with facets



Figure 3-10: A cross-sectional scanning electron micrograph of the facets in the Ge film. The facets serve to relax some of the strain during growth, leading to less strain in smaller devices and more strain in larger devices.

can be seen in Figure 3-10.

3.5 Waveguide integrated devices

After stand-alone Ge photodiodes were characterized, a process was developed to integrate a Ge diode in a Si electro-optic modulator process flow. This flow integrated Si modulators, Si waveguides, and Ge diodes using only CMOS-compatible processing steps. The process development was a joint effort between MIT Lincoln Laboratory and the Microsystem Technology Lab at MIT. Most of the fabrication was carried out at MITLL, with the exception of the Ge growth which was done at MTL. This section will discuss the process flow, electrical characteristics, and optical characteristics of Ge detectors integrated in this flow.

3.5.1 Ge diode fabrication

A process flow to fabricate Si waveguides and Si modulators was already well-developed by MIT LL. The goal of this study was to integrate the fabrication of a Ge diode. Due to thermal budget constraints, it was decided to integrate the germanium towards the end of the fabrication.

A cross-sectional schematic diagram of the wafer with Si modulators can be seen in Figure 3-11. For the purposes of this chapter, this is the starting substrate. The Si modulators are fabricated on silicon-on-insulator wafers so that the waveguides can be optically isolated from the bulk of the wafer. After modulator fabrication, the Si was etched. There were two flavors of Ge diodes grown. In the first, 50 nm of Si was left on which to grow Ge. In the second, all of the Si was etched away and the Ge was grown laterally from the sides of the Si modulator. This was done to ensure that the operation of the Ge diode would not be influenced by the underlying Si. Previous studies on Si diodes have shown that, under certain circumstances, they can absorb light in the near infrared [69]. By etching away all of the Si, that possibility was removed. The two different versions of the diode are shown schematically in Figure 3-12.

To minimize dopant diffusion in the Si, the Ge did not receive a post-growth cyclic anneal, regardless of the flavor of the diode. The structure was covered with either a PE-CVD oxide or a PE-CVD oxynitride. Contact vias were etched and filled with tungsten plugs. Finally, metal contacts were deposited and patterned. A cross-sectional schematic diagram of the final structure, lateral *pin* structures, can be seen in Figure 3-12.

Figure 3-13(a) shows the intended diode structure with 50 nm of silicon present. The actual diode structure had significant Ge overgrowth and looked more like the schematic in 3-13(b). The implications of this overgrowth will be discussed in Section



Figure 3-11: Schematic diagram of a silicon modulator fabricated by MIT LL. This was the starting substrate for the fabrication of waveguide-integrated Ge diodes.



Figure 3-12: Schematic diagrams of the two different types of waveguide-integrated germanium diodes. (a) A germanium detector fabricated on ~ 50 nm of Si. (b) A germanium detector fabricated by growing the germanium laterally from the Si sidewalls. In (b) all of the underlying Si is removed prior to germanium growth. Both diagrams are oriented such that the light propagates into the page and absorption in the germanium also takes place into the page.



Figure 3-13: Schematic diagrams of (a) intended and (b) actual waveguide-integrated Ge diode. The germanium overgrowth was unplanned but not detrimental to operation of the device.

3.5.2.

Due to processing difficulties, only a few sizes of devices survived each fabrication run. In the first run, where 50 nm of Si was left on which to grow Ge, only the 100 x $0.25 \ \mu \text{m}$ devices survived. In the second run, where all of the Si was etched, only the 500 x 0.35 $\ \mu \text{m}$ and 5 mm x 0.35 $\ \mu \text{m}$ devices survived. For brevity, the devices will only be referred to by their length in the subsequent sections.

3.5.2 Diodes grown on 50 nm of Si

Figure 3-14 shows a typical current versus voltage characteristic for a device grown on 50 nm of Si. The dark current is ~0.27 nA at -1 V (~1.1 mA/cm⁻²) and the $I_{\rm On}/I_{\rm off}$ ratio is 10⁷ which is one of the highest reported in the literature.

Optical measurements were taken by using the same broadband laser mentioned earlier and a lensed fiber with a 3 μ m diameter spot size. Light was coupled from the edge of the device into the Si waveguide. A typical responsivity versus wavelength curve can be seen in Figure 3-15.



Figure 3-14: Current versus voltage characteristic for a 100 x 0.25 μ m waveguide-integrated device. At -1 V, the dark current is ~0.27 nA. This measurement was taken on a device fabricated on wafer #5800.



Figure 3-15: Responsivity versus wavelength curve for a 100 x 0.25 μ m waveguideintegrated device. The responsivity is low at low voltages, but increases with increasing voltage. This device was fabricated on wafer #5800.

As seen in the figure, the responsivity at low voltages is low. At -1 V and 1550 nm, the responsivity is 0.02 A/W. However, this rises to 0.58 A/W at -12 V and 1550 nm. A remarkable feature of this responsivity curve is that the roll-off at 1550 nm is not as sharp as it is in the vertically illuminated devices. From 1550 nm to 1630 nm, the responsivity only falls by approximately 2X. For a vertically illuminated device, it falls by roughly 5X over the same range. This is likely due to the geometry of the device. Since the light is coupled from the edge and the device is so long, there is 100 μ m over which the light can be absorbed (compared to 2 μ m of Ge for the vertically illuminated devices). Germanium does not absorb long wavelengths very strongly, but, given enough material, it can. Figure 3-15 demonstrates the effect of the long Ge absorption tail.



Figure 3-16: Cross-sectional schematic of simulated device in Sentaurus. The silicon wings are doped, but the germanium is intrinsic. The entire structure sits on the burried oxide.

The strong voltage dependence of the device was investigated as well. Since the device is so narrow, it was anticipated that it would be fully depleted at or close to 0 V. Simulations were done in Sentaurus Device to further explore the electrostatics of the device.

Figure 3-16 shows the cross-section of the device in Sentaurus. The p- and n- type Si wings were each doped to 10^{19} cm⁻³ and the germanium was left intrinsic. Figure 3-17 shows the cross-section of the device with two different voltages applied. Figure 3-17(a) shows the device with an applied bias of -1 V. In this case, the maximum electric field is actually in the underlying Si and not in the Ge. The field is low in the Ge itself. At -5 V (Figure 3-17(b)), the maximum electric field is in the Ge and the underlying Si.

Figure 3-18 shows the electric field along a vertical cut down the center of the germanium. The different curves represent different applied voltages. The black dashed lines indicate the germanium region. The figure indicates that it is difficult to



Figure 3-17: Electric field at (a) -1 V and (b) -5 V. At low voltages, the electric field is in the underlying Si and not in the Ge.



Figure 3-18: Electric field down a vertical cut in the germanium. It is difficult to get a high electric field in the germanium.

get a high electric field in the germanium and that the electric field is preferentially in the underlying silicon. This is most likely due to the fact that the Ge is not doped separately. The p- and n-type contacts are in the Si.

3.5.3 Diodes grown laterally

While the responsivity that was measured in Figure 3-15 can be explained by the long tail of the absorption curve, there was some concern that the underlying Si could also be responsible for the absorption. To clarify which material was absorbing light, the Si was etched entirely down to the buried oxide and Ge was grown laterally from the



Figure 3-19: A schematic cross-sectional diagram of a Ge diode grown laterally from Si sidewalls. By removing all of the silicon, any measured photoresponse must be due to light absorbed in the the germanium.

sides. A schematic diagram of the device can be seen in Figure 3-19.

Typical current versus voltage curves for both a 5 mm long device and a 500 μ m device can be seen in Figure 3-20. The 500 μ m long device has a very flat curve in reverse bias, while the 5 mm long device has a much steeper slope. It is unclear what is the cause of these different shapes, though they are very consistent across devices and die. At -1 V, the 500 μ m devices have a dark current of 0.1 nA and the 5 mm devices have a dark current of 1 nA.

Figure 3-21 shows the responsivity versus wavelength curves for a typical 5 mm device at different reverse biases. Unfortunately, there were no working test structures on the chip to test for fiber-to-chip loss, so these measurements do not take that into account. Assuming a moderate value for the loss (2 dB), the actual responsivity of these devices could be significantly higher.

The responsivity for this device at -8 V and 1550 nm is 0.22 A/W. This is slightly lower than the photoresponse for the devices grown on 50 nm of Si, but still on the same order. This difference could easily be attributed to fiber-to-chip loss. This



Figure 3-20: Current versus voltage characteristics for two different lengths of laterally-grown diodes. The slopes of the dark current are very different for the different devices. The data was taken on devices from wafer #6665.



Figure 3-21: Responsivity versus wavelength characteristics for 5 mm long diodes. Responsivity is low at low biases, but improves to 0.22 A/W at -8 V and 1550 nm. The data was taken on devices from wafer #6665.

indicates that the germanium was indeed responsible for the photoresponse in the previous devices.

3.6 Chapter summary

This chapter discussed the fabrication and characterization of Ge diodes made from selectively-grown Ge films. It detailed the effect of junction position, annealing, and window size on the dark current of these devices. It also described the optical characterization of these devices and found that there is residual strain in the large devices. This strain, however, decreases with island size, making the bandgap slightly larger for small devices. This results in a shift of the photoresponse to shorter wavelengths.

This chapter also discussed the electrical and optical characterization of waveguideintegrated Ge diodes. These were some of the lowest dark current density devices that have been reported and they absorbed light out to longer wavelengths, due to their geometry.

Chapter 4

Photodiodes fabricated in blanket Ge-on-Si

Reproducibility was a large problem in the fabrication of diodes from selectivelygrown germanium films. Even with the sacrificial oxidation step, it was hard to judge how many threading dislocations were present in the germanium film from lot to lot. Without a firm grasp on the defects, it was hard to fully characterize the diodes. To allow us to study the physics of the devices more closely, diodes were also fabricated from blanket Ge films. This allowed more iterations of the process and let us hone in on the best device architecture.

This chapter discusses the electrical and optical characterization of diodes fabricated from a blanket Ge film. It first briefly discusses the germanium growth and then outlines the device fabrication procedure. Next, it discusses the effects of seed doping, passivation material, and a post-metallization anneal (PMA) on the dark current. To futher investigate the effects of the PMA, metal-oxide-semiconductor capacitors (MOSCAPs) were also fabricated and their electrical characteristics are discussed. Finally, the optical characteristics of these diodes are discussed.

4.1 Ge growth and fabrication

The growth of blanket Ge-on-Si films has been studied extensively in [26] and [70]. The effects of many factors on the growth and quality of the films are well documented there, but this section will summarize the important findings from that work. Unless otherwise specified, the optimal growth parameters described here were used to grow the germanium films used in this study.

Blanket germanium films were grown in a low-pressure chemical vapor deposition (LPCVD) epitaxial growth system. The starting substrates were six inch *p*-type CZ Si wafers with resistivities in the range of $0.005 - 0.02 \Omega$ -cm. The effects of chamber temperature, pressure, and hydrogen flow on the quality of the films were studied. It was found that a two-step growth procedure yielded the best results. The first step involved growing a low temperature p+ Ge seed layer. The optimum growth parameters for this layer were found to be 335° C at a pressure of 30 Torr with a hydrogen flow of 30 slpm. The seed layer must be at least 45 nm thick to yield a smooth surface morphology.

After seed layer growth, a thick ($\sim 2 \ \mu m$) Ge cap layer was grown at a higher temperature. The optimal temperature window for this step was determined to be 600°C - 700°C. At 650°C, chamber pressures between 30 Torr and 90 Torr yield smooth films. The hydrogen flow was kept at 30 slpm.

The films were cyclically annealed between 450° C and 830° C for 90 seconds and 150 seconds respectively. The samples went through 4 cycles of this anneal, all at a pressure of 30 T. It was found that a 4X anneal was a good trade-off between the amount of time needed for the anneal and the resulting film quality. Films grown with a 4X anneal previously were found to have a threading defect density of 2 x 10^7 cm⁻² [24].

Two different fabrication flows were used in this study. The first used a mesa etch to isolate the diodes while the second relied on n+ implantation for isolation. Both



Figure 4-1: Schematic process flow of the mesa-isolated diode fabrication process. There is optional doping in the Ge seed layer and at the top ~ 200 nm of Ge. The passivation is either LTO or alumina.

flows will be described here.

4.1.1 Process flow with mesa isolation

In this process flow, first 10 - 20 nm of LP-CVD oxide was deposited on the germanium film. This was to promote photoresist adhesion (it was found that standard OCG photoresist would not adhere to pure Ge). This stack was then patterned and mesa etched in a solution of H₂O:H₂O₂:HCl 4:3:1 for 30 - 60 seconds to remove 0.25 - 0.5 μ m of Ge. Following the etch, the oxide was stripped in buffered oxide etch (BOE). The wafers were then cleaned and the passivation material was deposited. The passivation material was either an LP-CVD oxide or Al₂O₃ deposited by atomic layer deposition. Next, the wafers were patterned and implanted with phosphorus at an energy of 35 keV and a dose of 5 x 10¹⁴ cm⁻². The projected range of the implant was ~25 nm below the Ge/oxide interface.

After implant, the junction was activated either at 550° C or 600° C for 10 seconds in a nitrogen ambient. Next, a thick (~100 nm) layer of PE-CVD oxide was deposited and contact vias were etched using standard dry and wet processes. A Ti/Al stack was deposited and etched to make metal contacts. A schematic diagram of this process flow can be found in Figure 4-1. In addition to varying the passivation material, doping variations were studied. The seed layer was optionally doped *p*-type using B_2H_6 at 2 x 10¹⁸ cm⁻³. This doping served two purposes. It included the highly defected region at the Si/Ge interface into the *p*-type contact, thereby removing it from the depletion region of the device. It also introduced some autodoping into the Ge cap layer, changing the doping profile of the entire device. On some wafers, the top 0.2 μ m of the Ge cap layer was doped *n*-type using PH₃ to 5 x 10¹⁷ cm⁻³. The purpose of this doping was to reduce the electric field at the surface of the germanium and reduce the recombination current. The effects of these doping variations will be discussed later in the chapter.

4.1.2 Process flow without mesa etch

The wet mesa etch of the germanium layer was difficult to control. The etch rate varied widely and it etched more quickly in the lateral direction than in the vertical direction, making alignment difficult in later steps. To fix this, the mesa etch was skipped and the top of the germanium film was never in-situ doped *n*-type. These devices relied on only the n+ implant and the field oxide to isolate them from neighboring devices.

In this process flow, PE-CVD oxide was first deposited and etched in BOE to make alignment marks. The oxide in the field was then stripped, leaving the alignment marks behind. (For the exact details on this part of the fabrication, please see appendix A.) Next, a screen oxide was deposited in an LP-CVD tube and the wafers were implanted. The energy of the implant ranged from 30 to 55 keV and the dose ranged from 5 x 10^{14} to 7 x 10^{14} cm⁻². The conditions varied to compensate for the variations in oxide thickness. The goal was to have the projected range be 50 - 60 nm below the Ge/oxide interface. A typical simulation done using the Stopping Range of Ions in Matter (SRIM) of the ion implantation profile can be seen in Figure 4-2. The intended projected range was deeper on these wafers (as compared to the


Figure 4-2: Simulated ion implantation profile of phosphorus in germanium. For this simulation, the oxide thickness was 8 nm and the implant energy was 30 keV. This put the projected range ~ 58 nm beneath the germanim/oxide interface. The dose was 5 x 10¹⁴ cm⁻².

mesa-isolated wafers) to ensure that the junction was not etched when the screen oxide was removed.

After implant, the dopants were activated at 550°C for 30 seconds. The longer anneal time was chosen to diffuse the phosphorus deeper into the germanium and to make sure that the junction was not etched when the screen oxide was removed. After removal of the screen oxide in BOE, the passivation layer, either ~ 10 nm of LP-CVD oxide or ~ 10 nm of Al₂O₃ deposited by atomic layer deposition (ALD), was used. A thicker (~ 100 nm) layer of PE-CVD oxide was then deposited and contact vias were etched using standard dry and wet processes. Finally a Ti/Al stack was deposited and patterned into contact pads. A schematic diagram of this process flow



Figure 4-3: Schematic process flow of the non-mesa isolated diodes. The steps to fabricate first-level alignment marks are omitted for brevity. The p+ Si substrate is used as the p-type contact.

can be seen in Figure 4-3.

In addition to the different passivation materials, this process allowed for the study of the effect of a post-metallization anneal. After metallization, some pieces of the wafers were annealed in nitrogen at temperatures ranging from 300°C to 450°C for 45 minutes.

4.2 Electrical characterization of blanket Ge-on-Si diodes

In this section, the electrical characterization of diodes fabricated with a mesa etch and without a mesa etch will be described. The impact of seed doping, passivation material, dopant activation anneal, and doping structure of the Ge cap layer are all studied for the mesa isolated devices. The impact of passivation material and post-metallization anneal are studied for the devices fabricated without a mesa etch.



Figure 4-4: Current versus voltage characteristics for 5 x 5 μ m devices with a doped and an undoped seed. The inclusion of doping in the seed reduces the dark current by ~5X at -1 V. These devices were passivated with Al₂O₃ and the dopants were activated at 600°C for 10 seconds. The device with the undoped seed is wafer #6444 and the device with the doped seed is wafer #6446.

4.2.1 Electrical characterization of mesa-isolated devices

In the first study, the low temperature germanium seed was grown both with and without *p*-type doping. The effect of this doping can be seen in Figure 4-4. The seed doping serves to reduce the leakage current for a 5 x 5 μ m device passived with LTO by ~5X at -1 V. Due to processing difficulties, the yield was very small on larger devices. This is likely because the highly defected region at the Si/Ge interface is incorporated into the *p*+ contact region and is not in the body of the device. When left undoped, the defects are in the depletion region of the device and can act as generation centers to contribute to the dark current.

It should be noted that the series resistance of these devices is high (~1 k Ω). This is likely due to the small size of the contact opening. Due to photolithography limitations, the contact via opening is only 1 x 1 μ m square, likely leading to a high contact resistance.

In the next study, devices were fabricated with the doped seed and the passivation material was varied. Figure 4-5 shows the current versus voltage characteristics for 10 x 10 μ m devices passived with LTO and Al₂O₃. At -1 V, the Al₂O₃ reduces the dark current by ~6X. Larger devices (20 x 20 μ m and larger) show a smaller effect of the passivation: at -1 V, the Al₂O₃ reduces the dark current by 2 - 3X. This indicates that the effect of the passivation is mainly around the perimeter of the device. The series resistance for these devices is much better because the contact via is larger than for the 5 x 5 μ m device. The contact opening is 6 x 6 μ m as compared to 1 x 1 μ m.

Figure 4-6 shows the current versus voltage characteristics for 10 x 10 μ m devices passivated with Al₂O₃ with different dopant activation anneals. The 550°C activation anneal reduces the dark current by ~3X over the 600°C anneal. This effect ranges from 2 - 5X over all device sizes. It is hypothesized that a higher activation annealling temperature degrades the leakage current because the Al₂O₃ film does not withstand high temperatures very well [71] [72].

Finally, the effect of *n*-type surface doping $(N_d = 3.5 \times 10^{17} \text{ cm}^{-3})$ at the top 0.2 μ m of the Ge film was studied. Figure 4-7 shows the current versus voltage characteristics for a 10 x 10 μ m device passivated with alumina, with dopant activation at 550°C, with and without an *n*- layer at the top of the Ge. The *n*-type layer reduces the dark current by 25% across all device sizes. The *n*- layer reduces the electric field at the surface, resulting in a lower surface generation current.

In summary, the lowest leakage diodes from the mesa isolated process flow can be made by including *p*-type doping of $\sim 2 \ge 10^{18}$ cm⁻³ in the Ge seed, passivating the



Figure 4-5: Current versus voltage characteristics for $10 \ge 10 \ \mu m$ mesa-isolated devices with different passivation materials. Using alumina to passivate the diodes reduces the dark current and improves the ideality factor. The dopants were activated at 550°C for 10 seconds. The device passivated with alumina was from wafer #6508 and the device passivated with LTO was from wafer #6505.



Figure 4-6: Current versus voltage characteristics for mesa-isolated devices passivated with Al_2O_3 with dopants activated at different temperatures. The 550°C anneal results in a lower dark current than the 600°C anneal. The device activated at 550°C was from wafer #6508 and the device activated at 600°C was from wafer #6506.



Figure 4-7: Current versus voltage characteristics for mesa-isolated devices with and without an *n*- layer at the Ge surface. The *n*- layer reduces the dark current by 25% across all device sizes. The devices were passivated with alumina and activated at 550°. The device with the *n*- doping was from wafer #6510 and the device without the *n*- doping was from wafer #6508.

structure with alumina, activating the dopants at 550°C, and including an *n*-layer at the top of the film to reduce the electric field. A 10 x 10 μ m device made with these conditions demonstrated a dark current of 2.5 μ A at -1 V.

4.2.2 Electrical characterization of non-mesa isolated diodes

As mentioned earlier, the mesa etch of the germanium film was hard to control. Removing 0.5 μ m of germanium in the vertical direction often removed 1 μ m of germanium in the lateral direction, making alignment difficult in subsequent steps. Eliminating the mesa etch and relying on n+ doping to isolate the devices fixed this problem.

These wafers were also designed to study the effect of passivation on the leakage current. Figure 4-8 shows the current versus voltage characteristics for 100 x 100 μ m devices passivated with both LTO and Al₂O₃. The Al₂O₃ reduces the dark current by ~6X at -1 V. This reduction is consistent for the larger (100 x 100 μ m and larger) device sizes. Small devices (20 x 20 μ m), however, had problems with large contact resistance, so a comparison is invalid.

To improve the series resistance of the 100 μ m square Al₂O₃ device, the wafers were annealed in nitrogen at temperatures ranging from 300°C to 450°C. The LTO devices were included as a control group. The Al₂O₃ devices did not survive these temperatures, as the dark currents often degraded. This is perhaps due to the alumina film's inability to withstand high temperatures. The LTO devices, however, showed significant improvement in dark current.

Effect of a post-metallization anneal on devices passivated with LTO

Figure 4-9(a) shows the current versus voltage characteristics for 10 x 10 μ m square devices after a post-metallization anneal (PMA) at various temperatures. Without an anneal, the dark current is roughly 10 μ A at -1 V. After an anneal at 400°C, the



Figure 4-8: Current versus voltage characteristics for devices fabricated without a mesa etch. Both curves are of 100 x 100 μ m devices passivated with LTO or Al₂O₃. Passivating the device with Al₂O₃ reduces the dark current by ~6X at -1 V. The device passivated with alumina was from wafer #6595 and the device passivated with LTO was from wafer #6596.



Figure 4-9: Current versus voltage characteristics (a) 10 x 10 μ m devices and (b) 100 x 100 μ m devices after post-metallization anneals at various temperatures. The anneal reduces the dark current by ~1000X for the 10 x 10 μ m device and by ~140X for a 100 x 100 μ m device. These devices were passivated with LTO. These devices were all from wafer #6596 with different pieces annealed at different temperatures.

dark current has dropped to 8 nA, roughly a factor of 1000X lower than for diodes without a PMA. Furthermore, the anneal dramatically changes the voltage dependence in reverse bias. The dark current for an unannealed device rises rapidly with increasing applied bias. As the anneal temperature increases, the current increases more gradually with reverse bias. As indicated in Figure 4-9(a), annealing at temperatures higher than 400°C does not yield a significant improvement in the dark current. Annealing at temperatures higher than 425°C caused the contacts to spike and resulted in shorted diodes. A similar investigation of larger devices shows a dark current reduction of ~140X for 100 μ m square devices (as shown in Figure 4-9(b)) and ~35X for 300 μ m square devices. The difference in dark current reduction factors indicates that the anneal is affecting the perimeter of the devices. Smaller devices have a larger perimeter-to-area ratio and are thus affected more by changing the passivation at the perimeter of the device.

To further study the perimeter and area dependencies of the dark current, the total current was broken down into its perimeter and area components. The total dark current can be expressed as

$$I_{Total} = J_P \times Perimeter + J_A \times Area \tag{4.1}$$

where J_P is the current due to the perimeter of the device and J_A is the perimeter due to the area of the device. Dividing the entire equation by the area of the device yields the following equation.

$$J_{Total} = J_P \times \frac{Perimeter}{Area} + J_A \tag{4.2}$$

When the total current density is plotted against the perimeter/area ratio for a range of different size devices, the slope of the line corresponds to the perimeter component of the leakage current and the y-intercept corresponds to the area component. Figure 4-10 shows the plots of J_{Total} v. $\frac{Perimeter}{Area}$ for both the unannealed and the annealed devices at -1 V. The anneal causes the J_P values to decrease by ~1000X from 2500 μ A to 2 μ A, further supporting the conclusion that the anneal primarily affects the perimeter of the devices. The fact that J_A is negative for the unannealed devices not fit this model very well and another mechanism dominates the dark current.

To investigate the effects of the anneal metal-oxide-semiconductor capacitors (MOSCAPs) were fabricated. The same Ge growth process was used and the wafers underwent the same LTO deposition process and Ti/Al stack processing as the diode wafers. A cross-sectional schematic diagram of the MOSCAPs is shown in Figure 4-11. These capacitors were subjected to PMA in N_2 for 45 minutes at varying temperatures. High-frequency capacitance-voltage (C-V) curves are shown in Figure 4-11. The PMA shifts the flatband voltage to higher voltages, suggesting that the anneal induces a



Figure 4-10: Perimeter/area analysis at -1 V for devices (a) without and (b) with a PMA. The anneal decreases the J_P values by ~1000X, indicating that the anneal primarily affects the perimeter of the devices. The J_P and J_A values of both annealed and unannealed devices can be seen in the figure.

negative fixed charge at the LTO/Ge interface. Other ambients were tested as well, as shown in figure 4-12. The shift is consistent in different ambients, indicating that the change in flatband voltage is due to a temperature effect and not an ambient effect.

Without an anneal, the Ge surface is depleted at 0 V. This large depletion region creates a high electric field at the surface and increases the surface recombination velocity, consistent with the large leakage current observed for diodes without the PMA. After an anneal at 400°C, the C-V characteristics indicate that the Ge surface is accumulated with holes at 0 V. This hole charge better isolates the device and reduces the region of high electric field, thereby reducing the dark current. Schematic illustrations of this are in Figure 4-13. It is important to note that the annealed devices will still have a small depletion region around the perimeter of the device due to the accumulated holes and n+ implant. This depletion region, however, is much



Figure 4-11: Capacitance versus voltage characteristics for 300 x 300 μ m devices annealed at different temperatures in N₂. The flatband voltage shifts to the right with increasing temperature and the hysteresis improves slightly. These devices were fabricated on wafer #6725.



Figure 4-12: Capacitance versus voltage characteristics for 300 x 300 μ m devices annealed in different ambients at 400°C. The flatband voltage shift is consistent across all ambients, indicating that the shift is due to temperature and not to ambient. The variation in C_{ox} can be attributed to oxide thickness variation across the wafer. These devices were fabricated on wafer #6725.



Figure 4-13: Schematic cross-sectional diagrams of the diodes (a) without and (b) with an anneal. Without an anneal, the Ge surface is depleted, causing a high electric field at the surface and a large recombination velocity. The anneal introduces a negative fixed charge at the Ge/LTO interface which attracts holes to the surface of the germanium. These holes better isolate the devices and reduce the depletion region width.

smaller than that in an unannealed device.

The flatband voltage shift from a device without an anneal to a device annealed at 400°C is 3.55 V. Assuming that this shift is entirely due to a change in the fixed charge, ΔQ_f , this corresponds to a $\Delta Q_f = C_{ox} * \Delta V_{FB} = 1.5 \times 10^{12} \text{ cm}^{-2}$.

Positive shifts in the flatband voltage with nitrogen, forming gas, or oxygen annealing have been reported for Ge MOS capacitors passivated with GeON [73], Al₂O₃ [74], HfO₂ [75], and a SiO₂/GeO₂ bilayer [76]. These results are consistent with theoretical calculations that dangling bonds in Ge are always negatively charged [77] at the Ge/dielectric interface and that an anneal increases the number of dangling bonds. Using the Berglund method [78], the interface state density (D_{it}) of the devices in this work was found to be in the 2 - 4 x 10¹² cm⁻² eV⁻¹ range at the flatband voltage. There is a slight increase in D_{it} upon annealing. This increase in D_{it} does not have a deleterious effect on the diodes, though, because the effect is small compared to the effect of the increased fixed charge. To confirm that the anneal draws holes to the surface, diodes were fabricated from germanium which was *in-situ* doped *p*-type with B_2H_6 at the end of the growth. This caused the top 0.2 μ m to be doped to ~5 x 10¹⁷ cm⁻³. The goal of this was to intentionally introduce holes at the interface to compensate for the depletion region. A cross-sectional schematic diagram of this device can be seen in Figure 4-14(b).

Figure 4-14(a) shows the current versus voltage characteristics for an unannealed device without any *p*-type doping at the top of the germanium, and an annealed device with *p*-type doping at the top of the germanium. The devices were all 5 x 5 μ m. The *p*-type doping reduces the dark current at low biases by ~400X. Furthermore, the PMA does not reduce the dark current by a significant amount, indicating that the PMA introduces fewer holes than 5 x 10¹⁷ cm⁻³. This is further supported by the fact that the dark current takes on a steep slope around -1 V. For devices that just received a PMA (like those in Figure 4-9(a)), the steep slope does not start until around -3 V. This steep slope is likely the result of a tunneling effect due to a high electric field. Since the doping introduces more holes, there will be a higher built-in electric field and it will require a lower bias to induce tunneling. This tunneling current will be discussed in greater detail in Chapter 5.

Reproducibility

The consistency of results from run to run was excellent. The diode wafers and MOSCAP wafers above were fabricated months apart, indicating that these results are robust and reproducible. To ensure the reproducibility of the diode results, the same process was done approximately one year after the original results were found. Figure 4-15 shows the current versus voltage characteristics for 10 x 10 μ m and 100 x 100 μ m devices fabricated in these two different runs. "Run A" is the early run and "Run B" is the later run. The dark current curves are nearly identical with the



Figure 4-14: (a) Current versus voltage characteristics for 5 x 5 μ m devices with different PMA and *p*-type doping characteristics. (b) A cross-sectional schematic diagram of a device doped *in-situ p*-type at the surface. The *p*-type doping reduces the dark current by ~400X as compared to an unannealed device without any *p*-type doping. When the doping is present at the top of the germanium, a PMA does not have a large effect. The device grown without a *p*-type layer was from wafer #6881 and the devices grown with a *p*-type layer were from wafer #6610.



Figure 4-15: Current versus voltage characteristics for (a) 10 x 10 μ m devices and (b) 100 x 100 μ m devices fabricated in different runs. These devices were passivated with LTO and received a PMA at 400°. The dark current characteristics are nearly identical for these runs, indicating that the results are reproducible and consistent. The device fabricated in Run A was from wafer #6596 and the device fabricated in Run B was from wafer #6881.

only appreciable difference being the series resistance. It seems that the later run has a better series resistance which is most likely due to better simulation of the ion implant because the screen oxide thickness was more consistent among wafers. These results show that the effect of the fixed charge is consistent for the materials available and has a large effect on the dark current.

4.3 Photoresponse of diodes fabricated from blanket Ge-on-Si

The photoresponse of both the mesa-isolated and non-mesa isolated devices was tested. Details of the photoresponse measurement and calibration procedure can be found in Appendix B. The photoresponse of both types of devices are similar,



Figure 4-16: Photoresponse versus wavelength characteristic for a 200 x 200 μ m device passivated with Al₂O₃ and fabricated with a mesa etch. The lack of a voltage dependence indicates a high internal electric field. At 1300 nm, the responsivity is 0.51 A/W. At 1550 nm, it has fallen to 0.36 A/W. This device did not receive a PMA. This device was from wafer #6508.

indicating that the PMA does not have much effect on the photoresponse.

Figure 4-16 shows the photoresponse versus wavelength characteristic for a 200 x 200 μ m mesa-isolated device. The device was passivated with Al₂O₃. At 1300 nm, the photoresponse is 0.51 A/W. At 1550 nm, the photoresponse falls to 0.36 A/W. The photoresponse does not depend much on voltage, indicating a large internal electric field at 0 V. The photoresponse was similar for mesa-isolated devices passivated with LTO.

Figure 4-17 shows the photoresponse versus wavelength characteristic for a 20 x 20 μ m non-mesa isolated device. This device was passivated with LTO and saw a PMA at 400° in N₂ for 45 minutes. At 1300 nm, the photoresponse is 0.44 A/W and it falls



Figure 4-17: Responsivity versus wavelength curve for a device fabricated without mesa isolation. This is a 20 x 20 μ m device passivated with LTO and subjected to a PMA at 400°C for 45 minutes. The curve shows comparable photoresponse to the mesa-isolated device. Inset: dark current of the device. This device was fabricated from wafer #6596.

to 0.30 A/W at 1550 nm. Again, there is not much voltage dependence, indicating a large internal electric field. The inset to figure 4-17 shows the current versus voltage characteristic for this device. The dark current at -1 V is \sim 30 nA.

A 200 x 200 μ m device passivated with LTO and with a PMA was also measured. The responsivity of this device was comparable to the 200 μ m device shown in Figure 4-16, indicating that the mesa-etch does not effect the photoresponse. At 1300 nm, the responsivity was 0.52 A/W and at 1550 nm, it was 0.40 A/W. A comparison of the photoresponse of the device in Figure 4-16 and a 200 x 200 μ m device fabricated in blanket Ge-on-Si that received a PMA can be seen in Figure 4-18. The curves are nearly identical, indicating that the PMA does not substantially effect the residual



Figure 4-18: Responsivity versus wavelength curves for 200 x 200 μ m devices measured at -1 V. The solid red curve was measured on a device that was mesa isolated and did not receive a PMA. The dashed blue curve was measured on a device that did not receive a mesa etch but that did receive a PMA. The nearly identical curves indicate that the PMA does not change the residual tensile strain in the films after growth. The device fabricated with a mesa etch was from wafer #6508 and the device fabricated without a mesa etch was from wafer #6881.

tensile strain present in the films after growth. The dark currents of these devices at -1 V are 0.70 μ A for the device with the PMA and 18 μ A for the device without the PMA.

4.4 Chapter summary

This chapter discussed the fabrication, electrical characterization, and optical characterization of germanium photodiodes fabricated from a blanket Ge-on-Si growth process. It was found that mesa-isolated devices were problematic due to the uncontrolled nature of the wet etch. This caused processing problems which probably also increased the dark current of the devices. To solve this problem, the process was modified to eliminate the mesa etch. Devices fabricated without the mesa etch demonastrated a reduction in dark current. However, a much larger improvement occurred when these devices were annealed after metallization. MOSCAP measurements show that the anneal changes the fixed charge at the Ge/oxide interface, which better isolates the device and reduces the dark current.

Finally, optical measurements of both types of devices show that the photoresponse is roughly the same, indicating that the anneal does not have much effect on the optical properties of the film.

Chapter 5

Dark current leakage mechanisms

In this chapter the mechanisms that are responsible for the leakage current in the diodes fabricated from blanket Ge-on-Si are explored. Both annealed and unannealed devices are analyzed. For the purposes of this chapter an "annealed" device is one annealed at 400°C for 45 minutes and an "unannealed" device is one that did not receive a post-metallization anneal (PMA). The scaling behavior and temperature dependence of the dark current of these devices and simulations performed in Sentaurus Device are presented.

5.1 Overview of dark current mechanisms

There are four main sources of reverse bias leakage current in diodes. These have been studied by others and details can be found in [79, 80]. A short overview will be given here.

The first source of leakage current is due to the diffusion of minority carriers in the quasi-neutral regions of the device. This component is given by

$$J_{diffusion} = q D_n \frac{n_i^2}{N_A L_n} [exp(\frac{qV}{kT}) - 1]$$
(5.1)

where q is the charge of an electron, D_n is the diffusion constant of electrons, n_i is the intrinsic carrier density, N_A is the concentration of acceptors, and L_n is the diffusion length of electrons given by $\sqrt{D_n\tau}$ where τ is the minority carrier lifetime. In reverse bias, this equation simplifies to

$$J_{diffusion} = q D_n \frac{n_i^2}{N_A L_n} \tag{5.2}$$

That is, the leakage current is constant with constant doping and temperature. As a function of temperature, the dominant temperature dependence of the diffusion current is proportional to $e^{(-E_g/kT)}$, where E_g is the bandgap of the material. This temperature dependence is due to the n_i^2 term. The intrinsic carrier concentration can be writted as

$$n_i = N_s exp(\frac{-E_g}{2kT}) \tag{5.3}$$

where N_s is the density of states and the other variables are defined as before. Diodes that are dominated by this form of leakage current are very high quality with good passivation.

The second form of leakage current is due to generation of carriers in the depletion region defined by the metallurgical junction. This mechanism can be described by

$$J_{bulk} = qW \frac{n_i}{\tau_g} [exp(\frac{qV}{2kT}) - 1]$$
(5.4)

where all constants are defined as before and W is the width of the depletion region and τ_g is the minority carrier generation lifetime. When a reverse bias is applied to the device, the depletion region will grow and the leakage current will increase. The extent to which the current increases is defined by how the depletion region grows, which in turn is dependent on the doping profile. For a diode with constant doping, the depletion region grows as a function of V^{α} where $\alpha = \frac{1}{2}$. For a linearly graded doping profile, the depletion region grows as a function of V^{α} where $\alpha = \frac{1}{3}$. For devices with a doping profile in between, the voltage dependence will be in between $\frac{1}{3}$ and $\frac{1}{2}$.

For diodes with leakage current dominated by the generation of carriers in the bulk, there is also a temperature dependence of the dark current. For these devices, an exponential dependence on $\sim \frac{E_q}{2}$ is expected due to the n_i term.

The third form of leakage current is generation of carriers at the surface of the semiconductor. It should be noted that this is different from the generation of carriers in a depletion region at the surface. This current is due to electrically active states at the semiconductor/insulator interface. The equation for this current can be written as

$$J_{surface} = qS_0 n_i [exp(\frac{qV}{2kT}) - 1]$$
(5.5)

where S_0 is the surface recombination velocity and is proportional to the interface trap concentration and the other variables are the same as before. In reverse bias, this current has no dependence on voltage (*i.e.* $\alpha = 0$). As a function of temperature, however, devices dominated by this mechanism will show a dominant exponential dependence on $\sim \frac{E_g}{2}$.

The fourth source of leakage current in diodes is tunneling and field assisted emission. Current due to these effects can vary widely depending on the exact mechanism at work. The general electric field dependence of these types of leakage currents can be written as

$$J_{tunnel} \propto exp(E^x) \tag{5.6}$$

where E is the electric field and x is a fitting parameter.

Figure 5-1 is a schematic representation of where these types of leakage currents



Figure 5-1: A schematic diagram of where the different sources of leakage current originate. (1) refers to the quasi-neutral regions where diffusion current originates. (2) refers to the depletion region of the device where generation of minority carriers occurs. (3) refers to the semiconductor/insulator interface, also the source of generation of minority carriers.

originate for the devices in this thesis. The regions labeled with a "1" indicate the quasi-neutral regions where diffusion current occurs. The region labeled with "2" indicate the depletion region in the bulk of the device where carrier generation occurs. The region labeled with "3" indicates the surface where surface generation occurs. Tunneling and field assisted emission can occur both in the bulk and at the surface and have been omitted from the figure for clarity.

5.2 Experimental scaling and temperature dependencies

To investigate the mechanisms behind the leakage current in the diodes fabricated from blanket Ge-on-Si films, the scaling, voltage, and temperature dependencies were all investigated for annealed and unannealed devices.

5.2.1 Scaling and voltage dependence of devices

Investigating how the dark current scales with different sizes was the first step to exploring the mechanisms. Only square diodes were available on the mask set utilized. Figure 5-2 shows the dark current for unannealed devices scaled by both the perimeter and the area of the devices. It is clear from the figure that the dark current of these devices does not scale with the area. Furthermore, the larger devices may be scaling with perimeter, although it is not a strong dependence. The smaller devices do not scale with anything. This indicates that unannealed devices are not dominated by diffusion or by generation current in the depletion region since both of these would scale with area.

Figure 5-3 shows the dark current scaled by perimeter and area for the annealed devices. From this figure, it is clear that the small devices scale with the perimeter and the large devices scale with the area of the device. This indicates that the large devices may be dominated by diffusion current or generation in the bulk while the small devices are dominated by a different effect. This is likely due to the sizes of the devices. The larger devices have small perimeter/area ratios and will be dominated by an area effect. The smaller devices have large perimeter/area ratios and will be dominated by a perimeter effect. All sized devices will have both effects, but they will be more or less pronounced based on the size of the device.

Temperature-dependent current versus voltage measurements were also taken. The devices were cooled using a liquid nitrogen source and current versus voltage curves were taken at seven different temperatures for two different device sizes. For some devices at lower temperatures, the current values were below the noise level of the parameter analyzer ($\sim 10^{-13}$ A). In that case, the data has not been shown.

Figure 5-4 and Figure 5-5 show the temperature-dependent current versus voltage



Figure 5-2: Current versus voltage characteristics scaled by (a) perimeter and (b) area for unannealed diodes. The larger devices scale with perimeter while the smaller devices do not scale at all. These measurements were all taken on devices fabricated on wafer #6596.



Figure 5-3: Current versus voltage characteristics scaled by (a) perimeter and (b) area for annealed diodes. The larger devices scale with area while the smaller devices scale with perimeter. This apparent difference is likely due to the difference in perimeter/area ratios of the different devices. These measurements were all taken on devices fabricated on wafer #6596.



Figure 5-4: Current versus voltage characteristics at different temperatures for (a) a 10 x 10 μ m device and (b) a 300 x 300 μ m device on an unannealed wafer. Both sizes demonstrate an activation energy of the dark current at -1 V of 0.12 eV. These measurements were all taken on devices fabricated on wafer #6596.



Figure 5-5: Current versus voltage characteristics at different temperatures for (a) a 10 x 10 μ m device and (b) a 300 x 300 μ m device on an annealed wafer. The 10 x 10 μ m device has an activation energy of 0.54 eV at -1 V while the 300 x 300 μ m device has an activation energy of 0.33 eV. These measurements were all taken on devices fabricated on wafer #6596.



Figure 5-6: Activation energy extraction for (a) unannealed devices and (b) annealed devices. The activation energies for the unannealed devices was 0.12 eV. The activation energy for large annealed devices was 0.33 eV and for small annealed devices was 0.54 eV. These measurements were all taken on devices fabricated on wafer #6596.

curves for both unannealed devices and annealed devices of two different sizes. From Figure 5-4, it is shown that the current is weakly temperature at higher temperatures and does not decrease significantly until below 230 K. Furthermore, extracting the activation energy at -1 V yields a value of 0.12 eV for both devices. This activation energy is consistent with tunneling or field-assisted emission as the source of dark current. The plot used to extract the activation energy can be seen in Figure 5-6(a).

Figure 5-5 shows different behavior. The IV curves are well-spaced as a function of temperature. Extracting the activation energy of these devices yields two different values: 0.54 eV for the 10 x 10 μ m device and 0.33 eV for the 300 x 300 μ m device. The activation energy of the larger device matches the value of half of the bandgap for germanium, possibly indicating the presence of minority carrier generation in the bulk. The plots used to extract the activation energies can be seen in Figure 5-6(b).

The final piece of analysis that was done on these devices was to extract the voltage dependence of the IV curve in reverse bias. By taking the natural log of both



Figure 5-7: The natural log of the current v. the natural log of the voltage for a (a) 10 x 10 μ m device and a (b) 300 x 300 μ m device from an unannealed wafer. At low biases, the value of α is 0.82 for the small device and 0.66 for the large device. This analysis was done on devices fabricated on wafer #6596.

the current and the voltage and fitting the points to a line, the power dependence of the voltage, or α can be extracted. For these devices α should be in between $\frac{1}{3}$ (the value for a linearly graded doping profile) and $\frac{1}{2}$ (the value for a constant doping profile).

Figure 5-7 shows the α extraction for the unannealed devices. The "low V" range is defined as -0.4 to -1 V, the "mid V" range is defined as -1.5 to -2.5 V, and the "high V" range is defined as -4 to -5 V. The figure shows that, at low voltages, the value of α is slightly different for the two different sizes: 0.82 for a 10 x 10 μ m device and 0.66 for a 300 x 300 μ m device. Both values are higher than 0.5, indicating that the dominant leakage mechanism is not bulk generation of carriers in the depletion region.

Figure 5-8 shows the α extraction for the annealed devices. The voltage ranges are defined as before. These devices show very different behavior. The 10 x 10 μ m device has $\alpha = 0.15$ while the 300 x 300 μ m device has $\alpha = 0.39$. These values are much lower than the values for the unannealed devices, indicating an entirely different



Figure 5-8: The natural log of the current versus the natural log of the voltage for a (a) 10 x 10 μ m device and a (b) 300 x 300 μ m device from an annealed wafer. At low biases, the value of α is 0.15 for the small device and 0.39 for the large device. This analysis was done on devices fabricated on wafer #6596.

mechanism for the source of the dark current.

5.2.2 Summary of scaling and voltage dependencies

Table summarizes the scaling dependencies, activation energies, and voltage dependencies of large and small devices on both unannealed and annealed wafers in reverse bias.

a large active mea	ourob ooo n ooo piin		
Device description	Current scaling	Activation	Voltage de-
		energy at -1	pendence (α)
		V (eV)	at low bias
Small, unannealed	None	0.12	0.82
Large, unannealed	Perimeter	0.12	0.66
Small, annealed	Perimeter	0.54	0.15
Large, annealed	Area	0.33	0.39

Table 5.1: Summary of device dependencies. A "small" device measures 10 x 10 μ m and a "large" device measures 300 x 300 μ m.

The large annealed devices show characteristics that match perfectly with the

characteristics of leakage current that is dominated by bulk generation in the depletion region. The small annealed devices will also have generation of carriers in the depletion region, but their α dependence and scaling dependence indicate that there is another mechanism at work. The α of 0.15 indicates that the other mechanism is most likely surface generation current. This is supported by the fact that the dark current scales with the perimeter of the devices.

The unannealed devices have activation energies that do not match either the bandgap nor half of the bandgap. Furthermore, the values of α are larger than what one would expect for current dominated by generation in the bulk. Thus, the dark current of these devices may be dominated by tunneling (*e.g.* band-to-band tunneling or trap assisted tunneling) or some field assisted emission. This is most likely due to the large depletion region at the surface of the Ge, as indicated by the MOSCAP measurements in Chapter 4.

5.3 Modeling of annealed devices

To confirm the dominant dark current mechanisms, the devices were modeled in Sentaurus Device. The most straightforward device to model was the large annealed device. The hypothesis is that the dark current is dominated by generation of minority carriers in the depletion region. This is responsible for the $\alpha = 0.39$ dependence and the activation energy of 0.33 eV. To model the device, it is necessary to get an accurate doping profile of the *p*-type autodoping. Spreading resistance profiling was done on the samples, seen in Figure 5-9, and the profile was imported into Sentaurus Device. Shockley-Read-Hall recombination/generation models were activated and a lifetime of 0.2 μ s was used. A cross-section of the device used in the simulation can be seen in Figure 5-10.



Figure 5-9: Spreading resistance profiling of boron in the germanium film. The autodoping from the seed layer causes the top of the germanium film to be doped $\sim 5 \times 10^{14} \text{ cm}^{-3}$. This analysis was done on devices fabricated on wafer #6596. Analysis courtesy of Solecon Laboratories.



Figure 5-10: Cross-sectional diagram of device modeled in Sentaurus Device. The p-type doping profile was imported from the spreading resistance profiling and the n-type doping was estimated using a gaussian distribution.



Figure 5-11: (a) Current versus voltage curves for a 300 x 300 μ m device and simulated device with a minority carrier lifetime of 0.2 μ s. (b) ln(J) versus ln(V) of experimental and simulated devices. Both the current density and the α dependence match very well at biases below -2 V. The experimental data was taken from a device on wafer #6596.

Figure 5-11 shows the current density versus voltage characteristics of the simulated structure as compared to an actual device that was 300 x 300 μ m large. 5-11(a) shows the full current versus voltage characteristic. It shows that there is excellent agreement between simulation and experiment out to ~-2 V. Figure 5-11(b) shows the natural log of the current density versus the natural log of the voltage for the simulated structure and the experimental device. The α values match very well in the -0.4 to -1 V range. The excellent agreement in both absolute current density and α indicate that the dominant leakage current mechanism for the large and annealed devices is Shockley-Read-Hall generation with a carrier lifetime of 0.2 μ s.

Figure 5-12 shows the sensitivity of the dark current density on the minority carrier lifetime parameter, τ . For a 10X increase in the lifetime, the dark current drops by 10X and the α dependence remains the same. The figure shows current density versus voltage for four different values of the lifetime, including $\tau = 0.2 \ \mu$ s.



Figure 5-12: Simulated current density versus voltage for different values of minority carrier lifetime. As the lifetime increases, the dark current decreases by an equal magnitude. A value of $\tau = 0.2 \ \mu s$ was found to best fit the large and annealed devices in this study.

5.4 Modeling of small devices

To accurately model the current versus voltage characteristics for small devices which were both annealed and unannealed the simulations had to be changed to 3D simulations to more accurately represent the perimeter/area ratio. A slice of the simulation can be seen in figure 5-13. It should be noted that the actual devices are square, but the most efficient way to do three dimensional simulations in Sentaurus Device is to exploit the use of cylindrical coordinates. Thus, the modeled devices will be circles rather than squares. A drawback to this choice is that the perimeter/area ratios will not be perfect for the simulations, making it difficult to match both the absolute current and the voltage dependence of the current simultaneously. For the sake of these simulations, it was more important to match the trend in α than it was to perfectly match the current. Thus, the surface recombination velocity was extracted to match the α values for the three smaller devices.


Figure 5-13: A cross-sectional schematic diagram of the structure used to calculate the surface recombination velocity. This is a slice of the device which is rotated radially to create a cylindrical device. The colors indicate the doping level.

For reference, the α values of experimental devices at low bias voltages are given in Figure 5-14(b). This figure shows that, as the device size grows, so does α , until it reaches a value of 0.39 for a 300 x 300 μ m device (not shown). The steepness of the slope reflects the balance of current due to generation in the depletion region of the bulk of the device and generation of current at the surface. The more surface generation current that is present, the flatter the curve will be (*i.e.* α approaches 0). The more bulk generation is present, the more α will trend towards 0.39.

Before S_0 was extracted, an estimate of the fixed charge both before and after anneal had to be calculated. MOSCAP modeling in Sentaurus Device using the same doping profile in the germanium indicated that the fixed charge of the device without a PMA was ~0.5 x 10¹² cm⁻² and devices with an anneal had a fixed charge of ~1 x 10^{12} cm⁻².

Experimental data was used to extract an estimate of the surface recombination velocity. For the small sizes (*i.e.* 5 μ m, 10 μ m, 20 μ m), the total device current can be written as

$$I_{total} = J_A A + I_{surface} \tag{5.7}$$



Figure 5-14: (a)Current versus voltage characteristics and (b) α extraction for devices of different sizes after an anneal. As the device size grows, the current is dominated by generation of carriers in the depletion region and α approaches 0.39. As the device size shrinks, the current is dominated by surface generation of carriers and α approaches 0. This analysis was done on devices fabricated on wafer #6596.

where J_A is the areal component of the current, A is the area of the n+ region of the device, and $I_{surface}$ is the current due to surface generation. In reverse bias, the surface generation current can be written as

$$I_{surface} = qA_{eff}S_0n_i \tag{5.8}$$

where q is the charge on an electron, A_{eff} is the area over which surface generation takes place (not the area of the $n \neq$ region), S_0 is the surface recombination velocity, and n_i is the intrinsic carrier concentration. By knowing the total measured current and the value of J_A (~1 mA/cm²) from the large area devices, the total surface current can be calculated. The charge of an electron and the intrinsic carrier concentration are constant, which leaves only the surface recombination velocity and the area. Thus, both S_0 and A_{eff} are unknowns.

To get an estimate of the area, simulations were run with a small active device



Figure 5-15: (a) A two dimensional plot of the surface generation current and (b) a one dimensional plot of the surface generation current in the y direction just below the surface of the Ge. The maximum surface generation takes place within a $\sim 2 \ \mu m$ region from the edge of the n+ region.

area and a large Ge/oxide interface area. This helped to generate an estimate of how far the surface generation current extended from the edge of the n+ region. Figure 5-15 shows both a two dimensional plot of the surface generation current and a cut in the y direction of the surface generation. The majority of the surface generation takes place over the first ~2 μ m of the Ge/oxide interface. Using both 2 μ m and 4 μ m as the estimates of the distance over which surface generation occurs, the values of S_0 are summarized in table 5.2.

Device size	Surface current (at	S_0 (cm/s) assuming	S_0 (cm/s) assuming
	-1 V)	$2 \ \mu m$ width	$4 \ \mu m$ width
$5 \ge 5 \mu m$	4.9 nA	3460	1350
$10 \ge 10 \ \mu \mathrm{m}$	7.2 nA	2660	1156
$20 \ge 20 \ \mu m$	11.2 nA	2270	1057

Table 5.2: Summary of surface recombination velocity extraction

The table demonstrates that there is reasonable agreement among S_0 values for various device sizes. Furthermore, the value of S_0 is expected to be in the 10³ cm/s range.

The goal of the simulations was to match the α values for the three smallest device

sizes. Again, since the perimeter/area ratios will be different for square and circular devices, the absolute current is not expected to match. Since the dark current of the three smallest device sizes scales with the perimeter of the devices, it was decided to simulate devices with the same perimeter. Thus, radii of the simulated devices were chosen to be 2.5 μ m, 5.75 μ m, and 12 μ m to match square devices with sides of the n+ regions measuring 4 μ m, 9 μ m, and 19 μ m.

An S_0 value of 5000 cm/s was found to give the best fit for the α values. Figure 5-16 shows the simulated absolute dark current and the α values for the three different device sizes with $S_0 = 5000$ cm/s. The absolute dark current is of the right order of magnitude as compared to the experimental devices shown in Figure 5-14(a). The α values match experiment very well. From run to run, it was found that the α value of a 4 x 4 μ m device varied between 0.9 and 0.12 for an actual device. Similar variations were found for the other sizes, though the trend is always the same. Thus, it was concluded that using a value of $S_0 = 5000$ cm/s is a reasonable estimate of the surface recombination velocity.

After extracting the value of S_0 for the devices, it was necessary to show the reduction in dark current for different surface conditions of the germanium. It was hypothesized previously that the anneal introduces a negative fixed charge which reduces the depletion region around the device. Thus, by switching the fixed charge from $-1 \ge 10^{12}$ cm⁻² to $0.5 \ge 10^{12}$ cm⁻², the simulations should show a drastic reduction in dark current. Figure 5-17 shows the current versus voltage characteristics for identical devices with different fixed charges. The reduction in dark current at -1 V is ~1000X.

Furthermore, the change in fixed charge accurately models the experimentally observed trend in ideality factor. The ideality factor of a diode is a measure of how closely the IV curve follows the ideal diode curve in forward bias. For an ideal diode, the ideality factor (η) is equal to 1. η usually ranges from 1 to 2 based on the quality



Figure 5-16: (a) Current versus voltage characteristics for devices of three different sizes and (b) their extracted α dependence. As the device size grows, α increases, keeping with the same trend as experimental devices. These simulations were done with a surface recombination velocity $S_0 = 5000$ cm/s.

of the diode. For experimental devices, η is 1.1 while unannealed devices have $\eta =$ 1.6. For the modeled devices, the idealty factor is 1.1 for the devices with a negative fixed charge while $\eta = 1.4$ for a positive fixed charge.

The final simulation to check was to investigate the effect of p-type doping in the top of the film. Figure 5-18 shows the current versus voltage characteristics of "unannealed" devices (*i.e.* those with a positive fixed charge at the Ge/oxide interface) with and without p-type doping at the top 200 nm of the Ge film. The addition of a p-type layer has the same effect as the anneal. The p-type layer reduces the dark current by ~1000X, even with a positive fixed charge at the interface.

Furthermore, Figure 5-19 shows simulated current versus voltage characteristics for devices with a *p*-type layer at the top of the Ge with a positive fixed charge and with a negative fixed charge. There is virtually no difference in the dark current for different fixed charge scenarios as expected. Moreover, when comparing the ex-



Figure 5-17: Current versus voltage characteristics for simulated devices with $Q_f = -1 \ge 10^{12} \text{ cm}^{-2}$ and $Q_f = 0.5 \ge 10^{12} \text{ cm}^{-2}$. The radius of the device was 5.75 μ m and $S_0 = 5000 \text{ cm/s}$. The change in fixed charge reduces the dark current by ~1000X.



Figure 5-18: Current versus voltage characteristics for simulated devices with $Q_f = 0.5 \times 10^{12} \text{ cm}^{-2}$ with and without a *p*-type layer. The addition of a *p*-type layer reduces the dark current in a similar fashion to the change in fixed charge.



Figure 5-19: Current versus voltage characteristics for simulated devices with p-type layers with $Q_f = -1 \ge 10^{12} \text{ cm}^{-2}$ and $Q_f = 0.5 \ge 10^{12} \text{ cm}^{-2}$. The change in fixed charge has virtually no effect on the dark current when there is a p-type layer at the top of the Ge.

perimental current versus voltage curves for devices with a p-type layer and devices without a p-type layer but with a PMA, there is a "kink" in the curve where the leakage current begins to increase rapidly. Due to the steep slope and the pinning of the current at high temperatures, this is most likely a tunneling effect, caused by the onset of a high electric field. When comparing these kinks, it is clear that the devices with the p-type doping have tunneling occur at lower bias (\sim -1 V as compared to \sim -2.5 V). This is most likely because there is a high electric field surrounding the entire n+ region for a device with the p-type doping. For devices with only the PMA, the high electric field is only in a small region around the perimeter of the device. The high electric field regions for both types of devices are illustrated schematically in Figure 5-21.



Figure 5-20: Current versus voltage characteristics for 5 x 5 μ m experimental devices. One device received a PMA while the other had a *p*-type layer grown in the top of the germanium. The circles indicate the "kink" in the curve where the dark current increases rapidly. The reduction in dark current is similar for both devices, but the kink for a device with a *p*-type layer happens at a lower reverse bias. The device with the *p*-type layer was from wafer #6610 and the device without the *p*-type layer but with a PMA was from wafer #6596.



Figure 5-21: Schematic diagrams of the location of the high electric field for (a) a device doped p-type in the top of the germanium and (b) a device which received a PMA. When devices were doped p-type, a high electric field forms around the n+ region and results in tunneling beginning at a lower applied bias. When the devices received a PMA, there was only a small region of the device where there was a high electric field. Thus, tunneling began at a higher applied bias.



Figure 5-22: Defect levels in germanium due to material contaminants. Candidates for mid-gap defect levels are Fe and Cu. Shallower defects may be caused by Ag, Au, or Cr. From [88].

5.5 Origin of dark current

Experimental and modeling results indicate that the dark current of large germanium photodiodes is dominated by generation in the depletion region as defined by the n+ implant of the device. Furthermore, the dark current of small germanium photodiodes is dominated by generation of carriers at the surface in the region outside the n+ implant. These results, however, do not indicate what is responsible for generation of carriers. The activation energies of the dark current can give us some hints.

At -1 V, the activation energy of small and large devices was 0.54 eV and 0.33 eV respectively. These activation energies indicate that there is a trap mechanism at work, somewhere in the bandgap. For the large devices, the trap is in the middle. For the small devices, the trap is either 0.54 eV away from the valence band or 0.54 eV away from the conduction band (*i.e.* 0.12 eV from one of the band edges). These trap levels are caused by some sort of defect, either crystal defects or material contaminants. Traps in germanium devices fabricated from a bulk wafer have been studied in detail previously [81, 82, 83, 84, 85, 86, 87]. A summary of the activation energies of a range of defects in germanium can be found in figure 5-22.

Mid-gap states in germanium may be caused by either iron or copper while shallower states (at 0.54 eV) may be caused by silver, gold, or chromium. To further



Figure 5-23: SIMS profiles for (a) Ge, Si, O, C, and Au and (b) Ge, Si, Cu, Ag, Fe, and Cr. The concentrations of Cu, Ag, Fe, and Cr were all below the detection level of SIMS. The concentration of oxygen, however, is in the 10^{17} cm⁻³ range in the top 0.4 μ m of the film. This analysis was done on wafer #6596.

investigate the source of the contaminants, samples were sent for secondary ion mass spectroscopy (SIMS). SIMS profiles of gold, copper, silver, iron, and chromium can be seen in Figure 5-23. The figure shows that these metals are not present in any appreciable amount in our samples. However, we cannot rule out material contaminants as the source of traps since trap densities expected from the deep level transient spectroscopy (DLTS) data (discussed next) are below the SIMS detection limit.

Deep level transient spectroscopy is another way to analyze the activation energy of defects in semiconductors [89]. In this process, the sample is cooled to cryogenic temperatures and a voltage is pulsed across the device. Based on how the current recovers at different temperatures, trap levels can be extracted. Furthermore, the shape of the recovery can indicate mechanisms of leakage current. A previous study on plastically deformed germanium has shown that an anneal in the range of 580°C reduces the amplitude of the DLTS signal by 10X while not affecting the threading defect density [90]. This suggests that the anneal temperatures in this study may change the magnitude of the number of traps without changing the number of defects. Previous DLTS studies on germanium have indicated a wide range of traps due to many different mechanisms. They are summarized in tables 5.3 and 5.4.

Trap level,	Author and citation	Due to
$E (E_V +$		
E)		
0.1 eV	Simoen [91]	dislocations
$0.15 \ \mathrm{eV}$	Auret [92]	
$0.16 \ \mathrm{eV}$	Poulin [93]	divacancy
$0.18 \mathrm{eV}$	Auret [92]	divacancy
$0.25~{\rm eV}$	Mooney [94]	vacancy-oxygen
$0.27~{ m eV}$	Markevich, Auret [95] [96]	vacancy-oxygen
$0.30 \ \mathrm{eV}$	Coutinho, Auret, Poulin	divacancy, impurity + de-
	[97] [96] [93]	fect
$0.31 \ \mathrm{eV}$	Lindberg [98]	vacancy-Sb
$0.52 \ \mathrm{eV}$	Poulin [93]	impurity + defect

Table 5.3: Hole traps in Ge

Table 5.4: Electron traps in Ge

Trap level,	Author and citation	Due to
$E (E_C - E)$		
0.12 eV	Mooney [94]	interstitial complex
$0.13 \ \mathrm{eV}$	Auret [96]	
$0.15~{\rm eV}$	Auret [92]	implant defects
$0.20 \ \mathrm{eV}$	Auret [96]	interstitial $+$ Sb
$0.21 \ \mathrm{eV}$	Markevich, Auret [95] [96]	Sb, vacancy $+$ oxygen
$0.24 \ \mathrm{eV}$	Auret [96]	interstitial $+$ Sb
$0.26 \ \mathrm{eV}$	Mooney [94]	
$0.27~{ m eV}$	Poulin [99]	impurity + defect
$0.29~{\rm eV}$	Kolkovsky [100]	
$0.30 \ \mathrm{eV}$	Simoen, Auret [101] [92]	Ni, implant defects
$0.31~{\rm eV}$	Auret [96]	vacancy
$0.32 \ \mathrm{eV}$	Poulin [99]	divacancy
0.53 eV	Mooney [94]	

Samples from this work were studied using DLTS. Due to capacitance constraints, only the largest diodes were suitable for DLTS studies. Analysis on the 300 x 300 μ m squares was performed by Daniel Johnstone from Semetrol on both annealed and unannealed wafers, looking for defects at 0.33 eV and 0.12 eV. DLTS measurements



Figure 5-24: Arrhenius plot of different trap levels found in a 300 x 300 μ m device that received a PMA. The analysis indicates that there is one minority trap and three majority traps present. Analysis courtesy of Daniel Johnstone, Semetrol. This analysis was done on wafer #6596.

on the annealed sample showed the presence of four trap levels at $E = E_v + 0.055$, $E = E_v + 0.13$, $E = E_v + 0.18$, and $E = E_c - 0.3$, as indicated in Figure 5-24. Unfortunately, the leakage current of the unannealed sample was too high to give an accurate measurement of trap concentration. Capacitance transients as a function of filling pulse duration were taken at 100 K on the annealed sample and the DLTS signal versus filling pulse duration can be seen in Figure 5-25. At 100 K, there were two trap levels very close to one another. The solid line represents a trap higher in energy while the dashed line represents a trap lower in energy. The sum of these is given in the dotted curve. The logarithmic shape of the sum is consistent with this trap level being associated with dislocations [91, 102].

To test this hypothesis, photodiodes were fabricated from blanket Ge-on-Si where the post-growth anneal (a cyclic anneal between 800°C and 450°C) was either elimi-



Figure 5-25: DLTS signal at 100K as a function of filling pulse duration for a 300 x 300 μ m device that received a PMA. There were two trap states close in energy at 100K. The solid line is the DLTS signal for a trap slightly higher in energy and the dashed line is the signal for the trap slightly lower in energy. The dotted line is the sum. The logarithmic shape of the dotted curve is consistent with this trap level being associated with dislocations [91, 102]. Analysis courtesy of Daniel Johnstone, Semetrol. This analysis was done on wafer #6596.



Figure 5-26: Threading dislocation density as measured by plan-view transmission electron microscopy versus thickness for different annealing cycles. At 1 μ m, the threading dislocations in germanium from an unannealed sample to one annealed for four cycles are reduced by ~4X. From [26].

nated or reduced to change the dislocation density. The thickness of the germanium was reduced to 1 μ m in an attempt to obtain a large spread in the threading dislocation density. Previous studies on similarly grown layers have shown that the dislocation density for 1 μ m films without an anneal should be on the order of 2 x 10^8 cm⁻² while the defects for a 4X anneal should be 5 x 10^7 cm⁻² [26]. Threading defect density as a function of film thickness for different annealing cycles can be seen in Figure 5-26.

Comparison of the dark current for 300 x 300 μ m devices with different anneals can be seen in Figure 5-27. The inclusion of a two cycle high temperature post-growth anneal reduces the dark current by 45X. The reduction decreases to 14X for 10 x 10 μ m devices that received two anneal cycles, indicating that the effect of the anneal is mostly an areal effect. There is little benefit observed for annealing beyond two cycles. This suggests that the threading defects are associated with the dark current



Figure 5-27: Current versus voltage for 300 x 300 μ m devices annealed for a different number of cycles. These devices were passivated with LTO and received a postmetallization anneal. The reduction in dark current at -1 V is ~45X and decreases to 14X for 10 x 10 μ m devices. Devices without a post-growth anneal were from wafer #6879 while devices with a 2X anneal were from wafer #6884 and devices with a 4X anneal were from wafer #6882.

when the defect level is high, but when the density is reduced to less than $\sim 10^8$ cm⁻², another mechanism is dominant. Furthermore, the dark current for the devices fabricated without a post-growth anneal scales with the area of the devices as seen in Figure 5-28. This further supports the hypothesis that the threading defects are the dominant mechanism responsible for the dark current when the defect level is in the 10^8 cm⁻² range.

It should be mentioned that it is possible the threading dislocations are not acting alone. It is possible that the dislocations are serving as gettering sites for heavy metal impurities. While SIMS did not detect heavy metals in a large concentration, there is a chance that they are still present in the films.

The defects contributing to the dark current could also be due to residual damage from phosphorus implantation. Previous studies have shown that germanium can



Figure 5-28: Dark current density versus voltage for different device sizes fabricated on a wafer which did not see a post-growth anneal. The dark current scales with the area of the device, indicating that, when the defect density is on the order of 10^8 cm⁻² the defects are the main source of leakage current. These devices were all on wafer #6879.

recrystallize by solid phase epitaxy after phosphorus implantation with an anneal at 400°C for 60 seconds [103]. Since the activation anneal after implantation was done at 550°C for 30 seconds, it was expected that most of the damage caused by the implant would be gone and there would not be an effect on the dark current. However, end-of-range damage would still be present [104] [105] and this might affect the dark current. To test this, devices were fabricated with identical processing steps except for the dose of the implant. One wafer was implanted with a dose of 5 x 10¹⁴ cm⁻² and another was implanted with a dose of 1 x 10¹⁴ cm⁻². Both of these doses were expected to amorphize the germanium; however, the higher dose was expected to leave more end-of-range damage. Thus, the wafer with the lower dose implant had less damage and might be expected to have a lower dark current.

Figure 5-29 shows the current versus voltage characteristics for 300 x 300 μ m devices with and without a PMA for different implant doses. The largest device sizes



Figure 5-29: Current versus voltage for 300 x 300 μ m devices with different implant doses (a) with a PMA and (b) without a PMA. Without a PMA, there is virtually no effect on the dark current. With a PMA, the higher dose implant has a slightly lower leakage current which is the opposite effect that is expected if implant damage were the source of the leakage current. The devices with an implant dose of 5 x 10¹⁴ cm⁻² were fabricated on wafer #6881 and the devices with an implant dose of 1 x 10¹⁴ cm⁻² were fabricated on wafer #6988.

were chosen because the end-of-range implant damage is expected to be an areal effect and these devices should have minimal perimeter effects. For devices without a PMA, the leakage current is virtually the same for the different doses. For devices with a PMA, the higher dose implant has a slightly lower leakage current which is the opposite of what was expected based upon simplified assumptions. The increase in dark current for the lower dose may be explained by the slightly larger depletion region at any given bias.

It should be noted that end-of-range damage from implantation in germanium has been shown to change in size and density with an anneal around 400°C [104] [106]. It is tempting to attribute the reduction in dark current to these changing defects. However, the MOSCAP samples discussed in chapter 4 did not receive an implant at all and the flatband voltage still shifted. Thus, it is unlikely that the end-of-range damage is the dominant factor for the leakage current. It is also possible that impurities such as oxygen, carbon, or hydrogen contribute to the dark current. These may act alone or may form complexes with crystal defects. The 0.12 eV activation energy for the leakage current is associated with all sizes of unannealed devices, which are believed to have large perimeter depletion regions at the surface. This indicates that this defect is associated with the surface. Tables 5.3 and 5.4 indicate that a crystal defect may be responsible for this trap state, perhaps coupled with an impurity. Previous work has indicated that oxygen forms thermal donors in germanium when annealed in the range of 300°C - 500°C [107] [108]. Hall effect measurements have shown activation energies of these thermal donors at 0.017, 0.04, and 0.2 eV [109]. SIMS analysis on our films showed oxygen content in the top 300 nm of the film to be $\sim 10^{17}$ cm⁻³, suggesting that oxygen complexes may be at play during the PMA.

It is further hypothesized that the equilibrium result of this reaction is to form GeO_4 complexes which act as thermal donors - *i.e.* nearly all are ionized. However, the GeO_4 complex must first form a GeO_3 complex which must first form a GeO_2 complex. Fuller has hypothesized that, for an incomplete reaction, the primary effect will be mostly GeO_2 complexes [109]. The GeO_3 complex would be neutral or partially ionized while the GeO_2 complex would result in an overall decrease in donors. Figure 5-30 shows the donor concentration as a function of temperature and time for various oxygen-doped germanium samples. The equilibrium value of donors differs based on the temperature of the reaction. At 350°C, equilibrium is not reached for ~100 hours. This implies that, for the temperatures and times in this study, the reaction is not reaching the final stage, but rather being stopped in an intermediate part where GeO_2 can form. It is possible that these complexes are responsible for the increase in holes at the surface of the germanium.

Another hypothesis is that the 0.12 eV activation energy of the unannealed devices is caused by hydrogen. Since the germanium is grown epitaxially using GeH_4 as its



Figure 5-30: Thermal donor concentration as a function of time and temperature. For the times and temperatures in this work, the reaction as not reached equilibrium and an intermediate complex of GeO_2 is formed. From [110].

source, there is most likely an appreciable level of hydrogen in the germanium film. Dobaczewski, *et al.* have found a donor level with activation energy of 0.11 eV due to hydrogen [111]. Furthermore, hydrogen is known to form both thermal donors and thermal acceptors based on various annealing temperatures and other atomic elements present [112]. More experiments would be needed to fully elucidate the effects of both hydrogen and oxygen on the germanium.

Finally, it should be noted that Van de Walle, *et al.* have hypothesized that dangling bonds at any interface involving germanium will always be negatively charged [113]. This is unlike their behavior at silicon interfaces and implies that even materials such as HfO_2 or $LaYO_3$, which have shown promising results for Ge MOSFET applications [114] [115], will result in a similar problem for germanium diodes as observed in this study. The dangling bonds lead to a fixed charge which will cause a large depletion region at the surface, regardless of the passivation material. This suggests that care must be taken to design the surface of the germanium such that this depletion region is eliminated, especially when fabricating small devices.

5.6 Chapter summary

This chapter examined the underlying physics and operation of the germanium diodes fabricated and characterized in chapter 4. It discussed the scaling, temperature dependence, and voltage dependence of the devices and formulated a theory for their operation. Modeling and sensitivity analysis in Sentaurus Device showed that the minority carrier lifetime was ~0.2 μ s and that a reasonable estimate of the surface recombination velocity was ~5000 cm/s. The modeling further showed the large effect that the fixed charge at the Ge/oxide interface can have on the dark current of the diode.

Finally, the chapter showed that the threading dislocations play a significant role in the dark current for densities in the 10^8 cm⁻² range. Below that, the dark current is most likely limited by another mechanism. It is hypothesized that either oxygen or hydrogen impurities are associated with the dark current, perhaps aided by the presence of dislocations in the film.

Chapter 6

Thesis summary and future work

This chapter will summarize the thesis and ennumerate the specific contributions to the field of Ge-on-Si photodiodes. It will end with suggestions for future work.

6.1 Thesis summary

The goal of this thesis was to understand the leakage current mechanisms present in Ge-on-Si photodiodes. Diodes were fabricated both from selectively-grown Ge-on-Si and from blanket Ge-on-Si deposited by low-pressure chemical vapor deposition. Both the dark current and the responsivity were studied in this thesis.

It was found that a responsivity of ~0.25 A/W for a 1 μ m-thick film and ~0.50 A/W for a 2 μ m-thick film were typical values at 1550 nm and -1 V. Furthermore, it was also confirmed that the presence of tensile strain in the film shifts the responsivity curve out to longer wavelengths, as observed in earlier studies. This tensile strain is due to differences in the thermal expansion coefficients between the Ge and the underlying Si. This strain shrinks the bandgap slightly and makes absorption at longer wavelengths possible.

Furthermore, it was found that the surface conditions of the germanium were very important to the leakage current of the device. As deposited, a low-pressure CVD oxide has a positive fixed charge at the germanium/oxide interface. This causes a large depletion region to form around the device, which dwarfs the actual size of the device. This depletion region causes a high electric field to be near the surface, increasing the leakage current. After a post-metallization anneal (PMA), the fixed charge becomes negative. The negative fixed charge draws holes to the surface which better isolates the device and reduces the leakage current. Growing a germanium film with the top part doped p-type confirmed the interpretation of these results.

Beyond the surface conditions other leakage current mechanisms were investigated. It was found that, when the threading defect density is increased above $\sim 10^8$ cm⁻³, the threading defects are the dominant source of leakage current. Deep level transient spectroscopy confirmed the importance of the defects in a dominant trap level. However, attempting to reduce the defect density much below that by annealing does not seem to have an effect on the dark current. Based on previous studies in the literature, it seems likely that either oxygen or hydrogen are causing the trap levels.

6.2 Contributions

- Photodiodes fabricated from selectively-grown germanium films were characterized and it was found that, below a certain defect density, the dark current density is not dependent on defects. The dark current is likely dependent on something else, such as generation in the depletion region due to another contaminant.
- The impact of strain on the photoresponse for selectively-grown diodes was characterized and confirmed with Raman analysis.
- The impact of germanium passivation was studied and it was found that similar devices patterned with Al_2O_3 have $\sim 6X$ less leakage current than devices patterned with a low temperature oxide.

- The impact of a post-metallization anneal was studied on photodiodes fabricated from blanket Ge-on-Si. Through MOSCAP measurements, it was found that the anneal was changing the fixed charge of the Ge/oxide interface, thereby drawing holes to the surface and better isolating the devices.
- The devices made from blanket Ge-on-Si were modeled using Sentaurus Device. A minority carrier lifetime of 0.2 μ s and a surface recombination velocity of 5000 cm/s were extracted. These values were used to demonstrate that the fixed charge at the surface is responsible for the ~1000X reduction in dark current. Furthermore, the simulations show that including a *p*-type layer at the top of the Ge will reduce the dark current by ~1000X regardless of the fixed charge at the interface.

6.3 Future work

For infrared imaging applications, germanium photodiodes should absorb light out to longer wavelengths (1.6 - 1.7 μ m). To do this, more strain is needed in the film. Introducing strain across the entire wafer proved unfruitful in this thesis, but introducing strain in a localized way may be promising. A high-stress material either above or around the device (much like the high-stress material currently used in nMOSFETs) is one possibility to introduce stress.

The reduction in dark current after the PMA indicates that the surface conditions of the germanium are vitally important to fabricating a photodiode with low leakage current. While the data is clear that a PMA changes the fixed charge at the interface, the physics behind this change are not well understood. To further explore the origin of fixed charge, different annealing conditions should be studied, both with MOSCAPs and photodiodes. Detailed DLTS analysis would be helpful for different annealing conditions. For example, the oxygen complexes described in Chapter 5 have been shown to disappear upon annealing at higher temperatures [110]. When films were annealed above 470°C, the thermal donors disappear and the germanium returns to its original state. If oxygen thermal donors are the source of the fixed charge, then a higher temperature anneal should be able to confirm this. The anneal should be done prior to the deposition of metal. It was found that annealing germanium photodiodes at temperatures higher than 425°C results in the spiking of the contacts. For this reason, the films should be annealed at higher temperatures and then the metal should be deposited.

To check for hydrogen, SIMS or surface SIMS should be done to quantify the concentration of hydrogen. This should be done both before and after the PMA. It is possible that the hydrogen is diffusing out of the germanium during the PMA, thus removing the defect energy level.

Furthermore, to fully optimize the doping structure of the germanium, gated diodes should be made. The gate can be used to modulate the surface conditions of the germanium, thus allowing the optimal concentration of holes. It was seen in chapter 4 that the addition of too many holes at the surface of the germanium can allow tunneling current to begin at relatively low biases. Gated diodes would allow the optimal concentration of holes to be found so that future diodes could be made from films doped at the optimum values.

Finally, with the knowledge of the optimal doping profiles, integration of selectively grown Ge-on-Si should be studied again with the goal of lowering the dark current. Aberg, *et al.* have shown promising results with their work on selectively grown Ge diodes [45], but with the knowledge of fixed charge and doping profiles in this thesis, those devices could be better understood and perhaps improved upon.

Appendix A

Process flows used in this thesis

A.1 Mesa isolation flow

This process outlines the steps and tools used in MIT MTL to fabricate mesa isolated Ge-on-Si photodiodes. Note that standard photoresist in ICL would not adhere to germanium, so a sacrificial oxide layer was used. The following process flow does not include all of the film thickness checks on the UV1280 or any scanning electron microscopy. These tools were used often, whenever they would be useful. In particular, the UV1280 was used to calibrate every etch on a dummy wafer and to verify that films were gone when they should be.

\mathbf{Step}	Tool	Notes
RCA clean + HF dip	ICL RCA station	Pre-epi clean
Ge growth	Epi reactor	$2 \ \mu \mathrm{m}$
Ge RCA clean	ICL RCA station	pre-LTO deposition clean
LTO deposition	ICL tube 6C	$\sim 10 \text{ nm}$ for hard mask
Coat frontside of wafer	ICL coater	standard recipe

Table A.1: Process flow for mesa isolation

Etch backside Ge	ICL Lam490B	Poly-Si timed etch, 8 - 10 min.
Ash photoresist	ICL asher	standard recipe
Coat wafers with photoresist	ICL coater	standard recipe
Expose wafers	ICL i-stepper	Expose mesa mask
Develop pattern	ICL coater	standard recipe
Etch LTO	ICL BOE station	10 - 30 s.
Ash photoresist	ICL asher	standard recipe
Etch mesas	TRL acidhood2	$H_2O:H_2O_2:HCl 4:3:1$ for 30 - 60 s.
Remove LTO	TRL acidhood2 or	BOE for 10 - 30 s.
	ICL BOE station	
Ge RCA clean	TRL acidhood2 or	pre-passivation clean
	ICL RCA station	
Passivation deposition	ICL ALD or ICL	$\sim 10 \text{ nm of Al}_2\text{O}_3 \text{ or LTO}$
	tube 6C	
Coat frontside of wafer	ICL coater	standard recipe
Expose wafers	ICL i-stepper	expose $n + \text{mask}$
Develop wafers	ICL coater	standard recipe
Ion implant	Sent to Innovion	implant P, energy and dose vary
Ash photoresist	ICL asher	standard recipe
Ge RCA clean	TRL acidhood2	use acidhood2 instead of RCA
		station because of alumina
Deposit interlayer dielectric	ICL DCVD	$\sim 150 \text{ nm oxide}$
Ge RCA clean	TRL acidhood2	use acidhood2 instead of RCA
		station because of alumina
Activate dopants	ICL RTP	550° - 600°, 10 - 60 s.
Coat wafers with photoresist	ICL coater	standard recipe

Expose wafers	ICL i-stepper	Expose contact via mask
Develop pattern	ICL coater	standard recipe
Dry etch of contact vias	ICL AME5000	recipe: NICOLEOXIDE
Wet etch of contact vias	TRL acidhood2	BOE, 15 - 30 s.
Pre-metal clean	TRL acidhood2	50:1 HF, 30 - 60 s.
Metal deposition	ICL Endura	recipe: CAITTIAL, 100 nm Ti
		and 1 μ m Al
Coat wafers with photoresist	ICL coater	standard recipe
Expose wafers	ICL i-stepper	Expose metal via mask
Develop pattern	ICL coater	standard recipe
Dry etch of metal contact	ICL Rainbow	recipe: Nicole_TiAl

A.2 Process flow without mesa isolation

This process flow was used to fabricate diodes without the mesa isolation etch. The mesa etch rate was highly variable and hard to control. Furthermore, the effect of the mesa sidewalls and corners was unknown. This flow was developed to use the same mask set but skip the mesa etch. To fully delineate which part of the flow was the part to preserve the alignment marks, that part is given in the next section.

Table A.2: Process flow without mesa i	isolatio	n
--	----------	---

\mathbf{Step}	Tool	Notes
RCA clean + HF dip	ICL RCA station	Pre-epi clean
Ge growth	Epi reactor	$2~\mu{ m m}$
Coat frontside of wafer	ICL coater	standard recipe

Etch backside Ge
Ash photoresist
Alignment mark protection
flow
Ge RCA clean
Screen oxide deposition
Coat frontside of wafer
Expose wafers
Develop wafers
Ion implant
Ash photoresist
Ge RCA clean
Activate dopants
Remove screen oxide
Ge RCA clean
Passivation deposition
Interlayer dielectric deposition
Coat frontside of wafer
Expose wafers
Develop wafers
Dry etch of contact vias
Wet etch of contact vias
Pre-metal clean
Metal deposition

ICL Lam490B ICL asher Various

ICL RCA station ICL tube 6C ICL coater ICL i-stepper ICL coater Sent to Innovion ICL asher ICL RCA station ICL RTP ICL BOE station ICL RCA station ICL tube 6C or ICL ALD ICL DCVD ICL coater ICL i-stepper ICL coater ICL AME5000 TRL acidhood2 TRL acidhood2 ICL Endura

Poly-Si timed etch, 8 - 10 min. standard recipe Various

pre-LTO deposition clean ~10 nm of LTO standard recipe expose $n \neq$ mask standard recipe implant P, energy and dose vary standard recipe pre-activation clean 550° - 600°, 10 - 60 s. BOE for 10 - 30 s. pre-passivation clean ~10 nm of LTO or Al₂O₃

~150 nm of oxide standard recipe expose contact via mask standard recipe recipe: NICOLEOXIDE BOE, 15 - 30 s. 50:1 HF, 30 - 60 s. recipe: CAITTIAL, 100 nm Ti and 1 μ m Al

ICL coater	standard recipe
ICL i-stepper	Expose metal via mask
ICL coater	standard recipe
ICL Rainbow	recipe: Nicole_TiAl
TRL tube A3 or	anneal in N_2 , forming gas, or vac-
XX	cum between 300°C - 425 °C
	ICL coater ICL i-stepper ICL coater ICL Rainbow TRL tube A3 or XX

A.3 Alignment mark protection flow

This process flow was developed so that the mask set used for the mesa-isolation process could also be used in a process without mesa isolation. This need stems from the fact that the mesa etch made the first level of alignment marks. The subsequent implant does not leave marks that are usable for alignment in later steps. In order to align to the implants, they had to be aligned to something. This process uses a PE-CVD oxide to create the alignment marks and then shutters the exposure in later steps to protect the marks but leave everything else open. The rest of the oxide structures are then removed with BOE.

Table A.3: Process flow for mesa isolatic	n
---	---

Step	Tool	Notes
Ge RCA clean	ICL RCA station	Pre-deposition clean
PE-CVD deposition	ICL DCVD	100 - 150 nm
Coat wafers with photoresist	ICL coater	standard recipe
Expose n+ mask	ICL i-stepper	standard recipe
Develop pattern	ICL coater	standard recipe

Etch pattern	ICL BOE station	30 - 90 s.
Ash photoresist	ICL asher	standard recipe
Coat wafers with photoresist	ICL coater	standard recipe
Expose wafers	ICL i-stepper	No mask is used. Do not expose
		entire reticle
Develop pattern	ICL coater	Resist should only be around
		perimeter of die. Alignment
		marks should be covered.
Remove oxide over most of die	ICL BOE station	30 - 90 s.
Ash photoresist	ICL asher	standard recipe

Appendix B

Procedure for taking photoresponse measurements

This appendix details the procedure and experimental setup for measuring the photoresponse of Ge-on-Si photodiodes fabricated in this thesis. The procedure changed over the years, but this is the final procedure that was used and very similar to previous versions.

In short, a Santac broadband laser (1260 nm - 1630 nm) coupled to a fiber was used to shine light on the devices and the output current was measured. The input power of the laser was varied and the photoresponse was calculated either from averaging the different current/power values or by measuring the slope of the line formed by the values. The measurements were done in Prof. Rajeev Ram's Physical Optics and Electronics Group's facilities. A schematic diagram of the setup can be seen in Figure B-1 and a more complete description of the measurement is included there.

Measurement setup

As mentioned before, a broadband Santac laser was used which is actually the combination of three different laser sources. A fiber was connected to the laser and



Figure B-1: A schematic diagram of the photoresponse measurement setup. The red boxes are the measurement instruments and the blue boxes are optical instruments. Furthermore, the green lines represent optical signals while the black ones represent electrical signals.

eventually split so that 90% of the light went to the device and 10% went to a lightwave mainframe that measured the output power of the laser. This split served as the power measurement of the laser. The rest of the light was then coupled to the device under test (DUT) either using a lensed fiber with a $\sim 3 \mu m$ spot size or a cleaved fiber with a $\sim 10 \mu m$ spot size. The fibers were chosen based on the size of the device. Large devices almost always used a cleaved fiber while small devices and waveguide-integrated devices always used a lensed fiber. The fibers went through three polarizers before reaching the DUT so that the polarization could be changed by the user.

The fiber was brought to the measurement setup stage and mounted on a custommachined fiber mount. The fiber could then be positioned with the use of 3 micrometers. The DUT was place on a custom-machined stage and electrical contact was made using a standard probe. The current was measured using a Keithley source meter (though previous iterations used an Agilent parameter analyzer). A diagram of the system can be seen in Figure B-1.

Jason Orcutt wrote python and Matlab scripts which controlled the power from the laser and the voltage applied to the device using GPIB. These scripts also sensed the output current of the device and saved them as Matlab data files. Python was used for responsivity measurements while Matlab was often used to measure the current versus voltage curve.

Each time a new set of measurements was going to be made, the setup had to be calibrated using a known device. To that end, a germanium photodiode from Thor Labs was purchased. Thor Labs measured the responsivity of the device from 800 nm to 1700 nm in increments of 10 nm. By measuring the Thor Labs device, a comparison could be made with the measured responsivity and the known responsivity and calibration coefficients could be determined at different wavelengths.

To measure new devices, a device was placed on the stage and contacted with a probe. A standard current versus voltage curve was taken in the dark and the data was saved. Next, the fiber had to be positioned carefully. The positioning of the fiber was a little different for vertically-coupled devices and edge-coupled devices and will be described separately.

Positioning of fiber for vertically-coupled devices

This fiber positioning system was used for all stand-alone vertical pin devices. The fiber was positioned over the DUT. Using a camera mounted on the table, the vertical distance from the laser to the DUT could be observed (but not measured with good accuracy). Using the micrometers, the fiber was lowered to an estimate of the focal length of the beam to ensure that as much light as possible was coupled into the device itself and not scattering to the field regions. To make sure the fiber was accurately placed, the laser was turned on and a bias (usually -1 V) was applied to the DUT. The fiber was moved in the x, y, and z directions using the micrometers while the output current was monitored on the Keithley. When the position which generated the maximum current was reached, the polarizers were moved to make sure the output current was at its maximum. For the vertically-coupled devices, the polarizers did

not change the signal by much. A few iterations of using the micrometers and the polarizers were done to ensure that the maximum signal was attained.

Positioning of the fiber for edge-coupled devices

This fiber positioning system was used for all waveguide-integrated devices. For these devices, the samples were cleaved carefully using a die saw and the sides were polished to give a clean waveguide surface on the edge of the test chip. The fiber was mounted horizontally rather than vertically and moved close to the edge of the chip. Using the optical microscope, the fiber was positioned close to the waveguide of interest. Making sure the fiber and the edge of the chip were both in focus at the same time, gives a reasonable starting position for alignment. The laser was then turned on. Using an infrared camera mounted to the microscope, it is possible to see the light being coupled into the waveguide. The z position was varied to get the light to travel as far down the waveguide as possible. After the z position was determined, a voltage was applied to the device and the current was monitored on the Keithley. The fiber was moved in the x and y directions to maximize the current. (At this point, the fiber may be moved in the z direction a bit, but a lot of adjustment should be unnecessary.) Finally, the polarizers were adjusted to maximize the output current of the DUT. It should be noted that the waveguides fabricated in this study were often highly polarized and, by using the polarizers, the responsivity could increase by as much as 10X.

Measurement and responsivity calculations

Once the fiber was positioned, contact was made to the device again using the probes and the python routine "responsivity.py" was run. This routine controls the wavelength and power of the laser and measures the ouptut current of the device. Typically, vertical devices were measured from 1300 nm to 1600 nm while waveguide-integrated devices were measured from 1540 to 1600 nm. This was because the waveguides were highly polarized and outside of that wavelength range, the responsivity would be very small and recalibrating the polarization for different wavelength ranges is very time consuming. Two or three different power levels were used from the laser and the current was measured at each.

After the raw data was measured, the numbers had to be processed a bit to get an accurate measurement of the responsivity. First, the current/power ratios were either averaged (for two powers) This is where the calibration coefficients were used. By comparing the measured photoresponse on the Thor Labs diode to the calibrated response given by the company, scaling coefficients could be determined. These scaling coefficients were then used to calculate the accurate value of the responsivity of the test devices. MATLAB code that both calculated the calibration coefficients and the final responsivity values is included below.

It should be noted that the Python responsivity routine often used three output powers, but the lowest one usually did not work, so those data points were ignored. The responsivity was then calculated by averaging the other two measured values (and multiplying by the calibration coefficient for the corresponding wavelength). Furthermore, the Thor Labs diode had measured responsivities spaced at every 10 nm and sometimes the measurement was taken at every 5 nm. In this casse, the calibration coefficients were interpolated to generate data for the Thor Labs diode at the intermediate steps.

MATLAB code to calculate responsivities

%{

First read in correction factors. 'correctionfactor.txt' is a text file with the correction coefficients calculated from multiplying the measured responsivity on the Thor Labs diode with the actual responsivity as given on the data sheet.

%}

corr_factor = textread('correctionfactor.txt'); corr_factor(31)=1.284;

%Need to interpolate correction factors in between

```
for i=1:30

corr_factor_interp(2*i-1) = corr_factor(i);

corr_factor_interp(2*i) = (corr_factor(i)+corr_factor(i+1))/2;

end
```

```
\operatorname{corr\_factor\_interp}(61) = \operatorname{corr\_factor}(31);
```

%Dark current values at different voltages

 $Idark_0V = 8.04e-10;$ $Idark_1V = 3.1e-7;$ $Idark_3V = 1.04e-6;$

 $\% {\rm Read}$ in wavelengths

waves = $responsivity_data(:,1);$

% At 0 V

 $power_meas_0V = [responsivity_data(:,3), responsivity_data(:,5), responsivity_data(:,7)];$
power_corr_0V = power_meas_0V*9; %to nominally correct for 90/10 split

$$\label{eq:linear} \begin{split} I_photo_meas_0V = [responsivity_data(:,4), responsivity_data(:,6), responsivity_data(:,8)]; \\ I_photo_corr_0V = I_photo_meas_0V - Idark_0V; \end{split}$$

 $R_meas_0V = I_photo_corr_0V./power_corr_0V;$ $R_0V = (R_meas_0V(:,2) + R_meas_0V(:,3))/2;$

 $R_{orr_0V} = R_0V.*corr_factor_interp.';$

% At -1 V

power_meas_ $1V = [responsivity_data(:,10), responsivity_data(:,12), responsivity_data(:,14)];$ power_corr_ $1V = power_meas_1V*9;$ %to nominally correct for 90/10 split

$$\label{eq:linear} \begin{split} I_photo_meas_1V = [responsivity_data(:,11), responsivity_data(:,13), responsivity_data(:,15)]; \\ I_photo_corr_1V = I_photo_meas_1V - Idark_1V; \end{split}$$

 $R_meas_1V = I_photo_corr_1V./power_corr_1V;$ $R_1V = (R_meas_1V(:,2) + R_meas_1V(:,3))/2;$

 $R_{corr_1V} = R_1V.*corr_factor_interp.';$

plot(waves, R_corr_0V, waves, R_corr_1V);

% At -3 V power_meas_ $3V = [responsivity_data(:,17), responsivity_data(:,19), responsivity_data(:,21)];$ power_corr_ $3V = power_meas_<math>3V^*9$; %to nominally correct for 90/10 split

I_photo_meas_3V = [responsivity_data(:,18), responsivity_data(:,20), responsivity_data(:,22)]; I_photo_corr_3V = I_photo_meas_3V - Idark_3V;

 $R_meas_3V = I_photo_corr_3V./power_corr_3V;$ $R_3V = (R_meas_3V(:,2) + R_meas_3V(:,3))/2;$

 $R_{corr_3V} = R_3V.*corr_factor_interp.';$

plot(waves, R_corr_0V, waves, R_corr_1V, waves, R_corr_3V);

%Generates matrix with wavelengths and responsivities at different biases

 $R_{tot} =$ [waves, R_corr_0V, R_corr_1V, R_corr_3V];

Appendix C

Inducing strain in blanket Ge-on-Si

This appendix describes attempts to induce strain in photodiodes frabricated from blanket Ge-on-Si. The diodes in this appendix were all fabricated using the mesa isolation flow described in chapter 4. The goal was to use backside metallization to induce bowing in the wafer and increase the tensile strain in the germanium film. Efforts were ultimately unsuccessful.

C.1 Wafer bending theory

So-and-so *et al.* reported an increase in strain due to backside silicidation [36]. It was found that 2 μ m of silicide deposited on the back of the wafer increased the strain in the frontside Ge by 0.05%. The strain in a thin film deposited on a wafer can be given by

$$\sigma_f = \frac{E_s d_s^3}{6(1 - \nu_s) R d_f^2 (1 + d_s/d_f)} \tag{C.1}$$

where σ_f is the strain in the film, R is the radius of curvature of the substrate, E_s and ν_s are Young's modulus and Poisson's ratio of the substrate, and d_s and d_f are the thicknesses of the substrate and the film respectively [116]. Furthemore, if the radius of curvature is much larger than the radius of the wafer, L, then the stress in the film can be related to the bow in the wafer and the equation becomes

$$\sigma_f = \frac{E_s d_s^3 B}{3(1 - \nu_s) L^2 d_f^2 (1 + d_s/d_f)}$$
(C.2)

where all variables are defined as above, B is the wafer bow, and L is the wafer radius. It should be noted that, in the above equations, the "film" is the film that is deposited on the substrate. For the samples in this work, the germanium on the top of the wafer and the metal on the back of the wafer should each follow this equation. Since the germanium has some residual stress from the high temperature growth, there is already a bow in the wafer. The metal on the backside should act to increase the overall bow in the wafer. To increase the overall bow the most, a high stress metal should be used on the back. It was further hypothesized that the substrate should be thinned, although the implications of this will be described in later sections.

C.2 Fabrication

After devices were fabricated using the mesa flow, different metals were sputtered onto the backsides of the wafer using the AJA Sputterer in the exploratory materials lab (EML). Both aluminum and tungsten were used. Since tungsten is a very high stress material, it was found that ~0.25 μ m could be deposited before the film started to flake. Aluminum is a much lower stress material, so up to 1 μ m was used. The KLA-Tencor FLX tool in TRL was used to measure the bow in the wafer.

Wafer thinning was also studied. Samples were thinned to $\sim 300 \ \mu m$ (normally the thickness was $\sim 675 \ \mu m$) by MIT Lincoln Laboratory and tungsten was deposited on the back of these thinned samples.

It should be noted that the samples in this study were smaller than a full wafer. The full wafers were usually cleaved in quarter wafer samples to make studying different metallization methods easier.

C.3 Results

Figure C-1 shows the responsivity versus wavelength curves for 200 x 200 μ m devices passivated with Al₂O₃ fabricated with a mesa etch. There is no appreciable difference in all the curves, indicating that the backside metallization did not substantially change the strain in the film. Aluminum is a low-stress metal, so the lack of increasing strain is somewhat expected. Furthermore, it is possible that when the sample was thinned, some of the original tensile strain in the germanium film was lost and the additional W could not replace it. It is unclear why the W sample did not show a marked difference in responsivity, but perhaps the limited thickness of the W film is at fault.

It is hypothesized that in order to increase the strain in the devices, the strain must be more localized and closer to the device itself. This would involve putting a high stress film either over the top of the device (similar to the high stress nitride layer in state-of-the-art nMOSFETs) or surround the device with a high stress film. Unfortunately, photolithography limitations in MTL prohibit that from happening.



Figure C-1: Responsivity versus wavelength curves for 200 x 200 μ m devices with various backside metallization schemes. These devices were fabricated using a mesa etch and passivated with Al₂O₃. Different backside metallization schemes had virtually no effect on the responsivity.

Appendix D

Frequency response of diodes fabricated in selectively-grown Ge-on-Si

Diodes fabricated in selectively-grown Ge-on-Si were made as part of a high speed analog-to-digital converter effort. The frequency response of these devices was measured by Jason Orcutt to measure the bandwidth of the devices. Due to the length of time required to run a measurement, only a few devices were measured. Results from both vertical *pin* devices and waveguide-integrated lateral *pin* devices are presented here.

D.1 Frequency response of vertical devices

Figure D-1 shows the frequency response of a 100 x 100 μ m vertical *pin* structure. The germanium is 2 μ m thick. The bandwidth is ~3 GHz and changes slightly with applied bias. This frequency response of this device is likely limited by the *RC* time constant of the device due mainly to a large series resistance. The current versus voltage characteristic can be seen in Figure D-2.



Figure D-1: Frequency response of a vertical *pin* device. The device is 100 x 100 μ m square fabricated on wafer #5365. The bandwidth is ~3 GHz and changes only slightly with applied bias. Measurement courtsey of Jason Orcutt.

D.2 Frequency response of waveguide-integrated devices

The waveguide-integrated devices with measured frequency response are the ones described in section 3.5.1 grown on 50 nm of silicon. The frequency response for a 100 μ m long device is shown in figure D-3 for different biases. The bandwidth of these devices is ~1 - 2 GHz. Based on some estimates of the series resistance and the capacitance of the device, the frequency response is likely not *RC* limited but rather limited by either germanium passivation problems or the material quality. Since the germanium is grown in a trench of silicon, the germanium will grow from three directions (rather than just one for stand alone devices). These additional Ge/Si interfaces likely increase the threading defects in the film. Furthermore, the films are not cyclically annealed after growth (to minimize dopant diffusion from the Si wings),



Figure D-2: Current versus voltage characteristic of a vertical *pin* device. The device is 100 x 100 μ m square fabricated on wafer #5365. The series resistance on this device is ~1000 Ω , limiting the bandwidth to ~3 GHz. Measurement courtsey of Jason Orcutt.



Figure D-3: Frequency response of a waveguide-integrated device. The device is 100 μ m long and has a 3 dB bandwidth of 1 - 2 GHz. Measurement courtsey of Jason Orcutt. From wafer #5800.

leaving a high defect density.

Bibliography

- A. Liu, R. Jones, L. Liao, D. Samara-Rubio, D. Rubin, O. Cohen, R. Nicolaescu, and M. Paniccia, "A high-speed silicon optical modulator based on a metaloxide-semiconductor capacitor," *Nature*, vol. 427, pp. 615–618, 2004.
- [2] R. Soref, "The Past, Present, and Future of Silicon Photonics," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 12, no. 6, pp. 1678–1687, 2006.
- [3] J. Liu, X. Sun, R. Camacho-Aguilera, L. C. Kimerling, and J. Michel, "Ge-on-Si laser operating at room temperature," *Optics Letters*, vol. 35, no. 5, pp. 679– 681, 2010.
- [4] J. Liu, M. Beals, A. Pomerene, S. Bernardis, R. Sun, J. Cheng, L. C. Kimerling, and J. Michel, "Waveguide-integrated, ulralow-energy GeSi electro-absorption modulators," *Nature Photonics*, vol. 2, pp. 433–437, 2008.
- [5] J. Liu, D. Pan, S. Jongthammanurak, K. Wada, L. C. Kimerling, and J. Michel, "Design of monolithically integrated GeSi electro-absorption modulators and photodetectors on an SOI platform," *Optics Express*, vol. 15, no. 2, pp. 623– 628, 2007.
- [6] T. Ghani, M. Armstrong, C. Auth, M. Bost, P. Charvat, G. Glass, T. Hoffmann, K. Johnson, C. Kenyon, J. Klaus, B. McIntyre, K. Mistry, A. Murthy, J. Sandford, M. Silberstein, S. Sivakumar, P. Smith, K. Zawadzki, S. Thompson, and M. Bohr, "A 90 nm High Volume Manufacturing Logic Technology Featuring Novel 45 nm Gate Length Strained Silicon CMOS Transistors," *IEEE International Electron Devices Meeting Technical Digest*, pp. 11.6.1–11.6.3, 2003.
- [7] A. Kilo, S. J. Spector, M. E. Grein, A. H. Nejadmalayeri, C. W. Holzwarth, M. Y. Sander, M. S. Dahlem, M. Y. Peng, M. W. Geiss, N. A. DiLello, J. U. Yoon, A. Motamedi, J. S. Orcutt, J. P. Wang, C. M. Sorace-Agaskar, M. A. Popovic, J. Sun, G.-R. Zhou, H. Byun, J. Chen, J. L. Hoyt, H. I. Smith, R. J. Ram, M. Perrott, T. M. Lyszczarz, E. P. Ippen, and F. X. Kaertner, "Photonic ADC: overcoming the bottleneck of electronic jitter," *Optics Express*, In Publication.
- [8] A. K. Sood, R. A. Richwine, Y. R. Puri, N. DiLello, J. L. Hoyt, N. Dhar, R. S. Balcerak, and T. G. Bramhall, "Development of SiGe Arrays for Visible-Near IR Imaging Applications," *Proceedings of SPIE*, vol. 7780, 2010.

- [9] B. M. Onat, W. Huang, N. Masaun, M. Lange, M. H. Ettenberg, and C. Dries, "Ultra Low Dark Current InGaAs Technology for Focal Plane Arrays for Low-Light Level Visible-Shortwave Infrared Imaging," *Proceedings of SPIE*, vol. 6542, 2007.
- [10] M. Jutzi, M. Berroth, G. Wohl, M. Oehme, and E. Kasper, "Ge-on-Si Vertical Incidence Photodiodes With 39-GHz Bandwidth," *IEEE Photonics Technology Letters*, vol. 17, no. 7, pp. 1510–1512, 2005.
- [11] S. L. Chuang, *Physics of Optoelectronic Devices*. John Wiley & Sons, Inc., 1995.
- [12] Semicondcutor Germanium Detectors: on silicon apperformance. semiconductors in Nov. proaches III-V 11, 2011<http://www.laserfocusworld.com/articles/print/volume-43/issue-5/features/semiconductor-detectors-germanium-on-silicon-approaches-iii-vsemiconductors-in-performance.html>.
- [13] Germanium band structure and carrier concentration. Nov. 11, 2011 http://www.ioffe.ru/SVA/NSM/Semicond/Ge/bandstr.html.
- [14] A. Nayfeh, C. O. Chui, T. Yonehara, and K. C. Saraswat, "Fabrication of High-Quality p-MOSFET in Ge Grown Heteroepitaxially on Si," *IEEE Electron Device Letters*, vol. 26, no. 5, pp. 311–313, 2005.
- [15] A. Ritenour, S. Yu, M. L. Lee, N. Lu, W. Bai, A. Pitera, E. A. Fitzgerald, D. L. Kwong, and D. A. Antoniadis, "Epitaxial Strained Germanium p-MOSFET with HfO₂ Gate Dielectric and TaN Gate Electrode," *IEEE International Electron Devices Meeting Technical Digest*, pp. 18.2.1–18.2.4, 2003.
- [16] D. A. Antoniadis, I. Aberg, C. Ni Cleirigh, O. M. Nayfeh, A. Khakifirooz, and J. L. Hoyt, "Continuous MOSFET performance increase with device scaling: The role of strain and channel material variations," *IBM Journal of Research* and Development, vol. 50, no. 4.5, pp. 363–376, 2006.
- [17] S. S. Iyer, G. L. Patton, J. M. C. Stork, B. S. Meyerson, and D. L. Harame, "Heterojunction Bipolar Transistors Using Si-Ge alloys," *IEEE Transactions* on *Electron Devices*, vol. 36, no. 10, pp. 2043–2064, 1989.
- [18] J. D. Cressler, "SiGe HBT Technology: A New Contender for Si-Based RF and Microwave Circuit Applications," *IEEE Transactions on Microwave Theory and Techniques*, vol. 46, no. 5, pp. 572–589, 1998.
- [19] C. N. Chleirigh, N. D. Theodore, H. Fukuyama, S. Mure, H.-U. Ehrke, A. Domenicucci, and J. L. Hoyt, "Thickness Dependence of Hole Mobility in Ultrathin SiGe-Channel p-MOSFETs," *IEEE Transactions on Electron Devices*, vol. 55, no. 10, pp. 2687–2694, 2008.

- [20] R. People and J. C. Bean, "Calculation of critical layer thickness versus lattice mismatch for Ge_xSi_{1-x}/Si strained layer heterostructures," *Applied Physics Letters*, vol. 47, no. 3, pp. 322–324, 1985.
- [21] M. Kim, P. Hashemi, and J. L. Hoyt, "Increased critical thickness for high Gecontent strained SiGe-on-Si using selective epitaxial growth," *Applied Physics Letters*, vol. 97, no. 26, pp. 26106–26106–3, 2010.
- [22] S. B. Samavedam, M. T. Currie, T. A. Langdo, and E. A. Fitzgerald, "Highquality germanium photodiodes integrated on silicon substrates using optimized relaxed graded buffers," *Applied Physics Letters*, vol. 73, no. 15, pp. 2125–2127, 1998.
- [23] W. Y. Loh, J. Wang, J. D. Ye, R. Yang, H. S. Nguyen, K. T. Chua, J. F. Song, T. H. Loh, Y. Z. Xiong, S. J. Lee, M. B. Yu, G. Q. Lo, and D. L. Kwong, "Impact of Local Strain from Selective Epitaxial Germanium With Thin Si/SiGe Buffer on High-Performance p-i-n Photodetectors With a Low Thermal Budget," *IEEE Electron Device Letters*, vol. 28, no. 11, pp. 984–986, 2007.
- [24] O. O. Olubuyide, D. T. Danielson, L. C. Kimerling, and J. L. Hoyt, "Impact of seed layer on material quality of epitaxial germanium on silicon deposited by low pressure chemical vapor deposition," *Thin Solid Films*, vol. 508, no. 1-2, pp. 14–19, 2006.
- [25] H.-C. Luan, D. R. Lim, K. K. Lee, K. M. Chen, J. G. Sandland, K. Wada, and L. C. Kimerling, "High-quality Ge epilayers on Si with low threadingdislocations densities," *Applied Physics Letters*, vol. 75, no. 19, pp. 2909–2911, 1999.
- [26] M. Kim, Limited-Area Growth of Ge and SiGe on Si. PhD thesis, MIT, 2011.
- [27] J.-S. Park, J. Bai, M. Curtin, B. Adekore, M. Carroll, and A. Lochtefield, "Defect reduction of selective Ge epitaxy in trenches on Si(100) substrates using aspect ratio trapping," *Applied Physics Letters*, vol. 90, no. 5, pp. 052113– 052113–3, 2007.
- [28] T. A. Langdo, C. W. Leits, M. T. Currie, E. A. Fitzgerald, A. Lochtefeld, and D. A. Antoniadis, "High quality Ge on Si by epitaxial necking," *Applied Physics Letters*, vol. 76, no. 25, pp. 3700–3702, 2000.
- [29] M. Oehme, J. Werner, M. Jutzi, G. Wohl, E. Kasper, and M. Berroth, "Highspeed germanium photodiodes monolithically integrated on silicon with MBE," *Thin Solid Films*, vol. 508, pp. 393–395, 2006.
- [30] S. Fama, L. Colace, G. Masini, G. Assanto, and H.-C. Luan, "High performance germanium-on-silicon detectors for optical communications," *Applied Physics Letters*, vol. 81, no. 4, pp. 586–588, 2002.

- [31] L. Colace, M. Balbi, G. Masini, G. Assanto, H.-C. Luan, and L. C. Kimerling, "Ge on Si p-i-n photodiodes operating at 10 Gbit/s," *Applied Physics Letters*, vol. 88, no. 10, pp. 101111–101111–3, 2006.
- [32] J. M. Hartmann, A. Abbadie, A. M. Papon, P. Hollinger, G. Rolland, T. Billon, and J. M. Fedeli, "Reduced pressure-chemical vapor deposition of Ge thick layers on Si(001) for 1.3 - 1.55 μm photodetection," *Journal of Applied Physics*, vol. 95, no. 10, pp. 5905–5913, 2004.
- [33] D. Suh, S. Kim, J. Joo, and G. Kim, "36-GHz High-Responsivity Ge Photodetectors Grown by RPCVD," *IEEE Photonics Technology Letters*, vol. 21, no. 10, pp. 672–674, 2009.
- [34] J. Osmond, G. Isella, D. Chrastina, R. Kaufmann, M. Acciarri, and H. von Kanel, "Ultralow dark current Ge/Si(100) photodiodes with low thermal budget," *Applied Physics Letters*, vol. 94, no. 20, pp. 201106–201106–3, 2009.
- [35] M. Oehme, J. Werner, E. Kasper, M. Jutzi, and M. Berroth, "High bandwidth Ge p-i-n photodetector integrated on Si," *Applied Physics Letters*, vol. 89, no. 7, pp. 071117–071117–3, 2006.
- [36] J. Liu, D. D. Cannon, K. Wada, Y. Ishikawa, S. Jongthammanurak, D. T. Danielson, J. Michel, and L. C. Kimerling, "Tensile strained Ge p-i-n photode-tectors on Si platform for C and L band telecommunications," *Applied Physics Letters*, vol. 87, no. 1, pp. 011110–011110–3, 2005.
- [37] J. Osmond, L. Vivien, J.-M. Fedeli, D. Marris-Morini, P. Crozat, J.-F. Damlencourt, E. Cassan, and Y. Lecunff, "40 Gb/s surface-illuminated Ge-on-Si photodetectors," *Applied Physics Letters*, vol. 95, no. 15, pp. 151116–151116–3, 2009.
- [38] K. Morii, T. Iwasaki, R. Nakane, M. Takenaka, and S. Takagi, "High Performance GeO₂/Ge nMOSFETs with Source/Drain Junctions Formed by Gas Phase Doping," *IEEE Electron Devices Meeting*, pp. 29.3.1–29.3.4, 2009.
- [39] Y.-L. Chao and C. S. Woo, "Germanium n+/p Diodes: A Dilemma Between Shallow Junction Formation and Reverse Leakage Current Control," *IEEE Transactions on Electron Devices*, vol. 57, no. 3, pp. 665–670, 2010.
- [40] Y. Ishikawa, K. Wada, D. D. Cannon, J. Liu, H.-C. Luan, and L. C. Kimerling, "Strain-induced band gap shrinkage in Ge grown on Si substrate," *Applied Physics Letters*, vol. 82, no. 13, pp. 2044–2047, 2003.
- [41] H.-Y. Yu, D. Kim, S. Ren, M. Kobayashi, D. A. B. Miller, Y. Nishi, and K. C. Saraswat, "Effect of uniaxial-strain on Ge p-i-n photodiodes integrated on Si," *Applied Physics Letters*, vol. 95, no. 16, pp. 161106–161106–3, 2009.

- [42] J. Liu, D. D. Cannon, K. Wada, Y. Ishikawa, D. T. Danielson, S. Jongthammanurak, J. Michel, and L. C. Kimerling, "Deformation potential constants of biaxially tensile stressed Ge epitaxial films on Si(100)," *Physical Review B*, vol. 70, no. 15, pp. 155309–155309–5, 2004.
- [43] L. M. Giovane, H.-C. Luan, A. M. Agarwal, and L. C. Kimerling, "Correlation between leakage current density and threading dislocation density in SiGe *p-i-n* diodes grown on relaxed graded buffer layers," *Applied Physics Letters*, vol. 78, no. 4, 2000.
- [44] H.-Y. Yu, S. Ren, W. S. Jung, A. K. Okyay, D. A. B. Miller, and K. C. Saraswat, "High-Efficiency p-i-n Photodetectors on Selective-Area-Grown Ge for Monolithic Integration," *IEEE Electron Device Letters*, vol. 30, no. 11, pp. 1161–1163, 2009.
- [45] I. Aberg, B. Ackland, J. V. Beach, C. Godek, R. Johnson, C. A. King, A. Lattes, J. O'Neill, S. Pappas, T. S. Sriram, and C. S. Rafferty, "A low dark current and high quantum efficiency monolithic germanium-on-silicon CMOS imager technology for day and night imaging applications," *IEEE Electron Devices Meeting*, pp. 2.3.1–2.3.4, 2010.
- [46] T. H. Loh, R. Nguyen, R. Murthy, M. B. Yu, W. Y. Loh, G. Q. Lo, N. Balasubramanian, D. L. Kwong, J. Wang, and S. J. Lee, "Selective epitaxial germanium on silicon-on-insulator high speed photodetectors using low-temperature ultrathin Si_{0.8}Ge_{0.2} buffer," *Applied Physics Letters*, vol. 91, no. 7, pp. 073503– 073503–3, 2007.
- [47] D. Feng, S. Liao, P. Dong, N.-N. Feng, H. Liang, D. Zheng, C.-C. Kung, J. Fong, R. Shafiiha, J. Cunningham, A. V. Krishnamoorthy, and M. Asghari, "High-speed Ge photodetector monolithically integrated with large crosssection silicon-on-insulator waveguide," *Applied Physics Letters*, vol. 95, no. 26, pp. 261105–261105–3, 2009.
- [48] L. Vivien, J. Osmond, J.-M. Fedeli, D. Marris-Morini, P. Crozat, J.-F. Damlencourt, E. Cassan, Y. Lecunff, and S. Laval, "42 GHz p.i.n Germanium photodetector integrated in a silicon-on-insulator waveguide," *Optics Express*, vol. 17, no. 8, pp. 6252–6257, 2009.
- [49] T. Yin, R. Cohen, M. M. Morse, G. Sarid, Y. Chetrit, D. Rubin, and M. J. Paniccia, "40Gb/s Ge-on-SOI waveguide photodetectors by selective Ge growth," *Conference on Optical Fiber Communication/National Fiber Optic Engineers Conference*, pp. 1–3, 2008.
- [50] N.-N. Feng, P. Dong, D. Zheng, S. Liao, H. Liang, R. Shafiiha, D. Feng, G. Li, J. E. Cunningham, A. V. Krishamoorthy, and M. Asghari, "Vertical p-i-n germanium photodetector with high external responsivity integratd with large core Si waveguides," *Optics Express*, vol. 18, p. FIX THIS, 2010.

- [51] J. Wang, W. Y. Loh, K. T. Chua, H. Zang, Y. Z. Xiong, S. M. F. Tan, M. B. Yu, S. J. Lee, G. Q. Lo, and D. L. Kwong, "Low-Voltage High-Speed (18 GHz/1 V) Evanescent-Coupled Thin-Film-Ge Lateral PIN Photodetectors Integrated on Si Waveguide," *IEEE Photonics Technology Letters*, vol. 20, p. 17, 2008.
- [52] J. Wang, W. Y. Loh, K. T. Chua, H. Zang, Y. Z. Xiong, T. H. Loh, M. B. Yu, S. J. Lee, G.-Q. Lo, and D.-L. Kwong, "Evanescent-Coupled Ge p-i-n Photodetectors on Si-Waveguide with SEG-Ge and Comparative Study of Lateral and Vertical p-i-n Configurations," *IEEE Electron Device Letters*, vol. 29, no. 5, pp. 445–448, 2008.
- [53] C. Kopp, A. Ferron, J.-M. Hartmann, M. Fournier, E. Augendre, P. Grosse, J.-M. Fedeli, and H. Derks, "Wafer Level High-Speed Germanium Photodiode Array Integration," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 17, no. 3, pp. 526–530, 2011.
- [54] L. Vivien, M. Rouviere, J.-M. Fedeli, D. Marris-Morini, J.-F. Damlencourt, J. Mangeney, P. Crozat, L. El Melhaoui, E. Cassan, X. Le Roux, D. Pascal, and S. Laval, "High speed and high responsivity germanium photodetector integrated in a Silicon-On-Insulator microwaveguide," *Optics Express*, vol. 15, no. 15, pp. 9843–9848, 2007.
- [55] T. Temkin, T. P. Pearsall, J. C. Bean, R. A. Logan, and S. Luryi, "Ge_xSi_{1-x} strained-layer superlattice waveguide photodetectors operating near 1.3 μ m," *Applied Physics Letters*, vol. 48, no. 15, pp. 963–965, 1986.
- [56] Y. Kang, H.-D. Liu, M. Morse, M. J. Paniccia, M. Zadka, S. Litski, G. Sarid, A. Pauchard, Y.-H. Kuo, H.-W. Chen, W. S. Zaoui, J. E. Bowers, A. Beling, D. C. McIntosh, X. Zheng, and J. C. Campbell, "Monolithic germanium/silicon avalanche photodiodes with 340 GHz gain-bandwidth product," *Nature Photonics*, vol. 3, pp. 59–63, 2009.
- [57] S. Assefa, F. Xia, S. W. Bedell, Y. Zhang, T. Topuria, P. M. Rice, and A. Vlasov, Yurii, "CMOS-integrated high-speed MSM germanium waveguide photodetector," *Optics Express*, vol. 18, no. 5, pp. 4986–4999, 2010.
- [58] L. Chen and M. Lipson, "Ultra-low capacitance and high speed germanium photodetectors on silicon," *Optics Express*, vol. 17, no. 10, pp. 7901–7906, 2009.
- [59] G. Masini, V. Cencelli, L. Colace, F. De Notaristefani, and G. Assanto, "Monolithic integration of near-infrared Ge photodetectors with Si complementary metal-oxide-semiconductor readout electronics," *Applied Physics Letters*, vol. 80, no. 18, pp. 3268–3270, 2002.
- [60] RCA clean. Dec. 13, 2011 < http://en.wikipedia.org/wiki/RCA_clean>.
- [61] H.-H. Park, K.-H. Kwon, J.-L. Lee, K.-S. Suh, O.-J. Kwon, K.-I. Cho, and S.-C. Park, "Characterization and removal of silicon surface residue resulting

from chf3/c2f6 reactive ion etching," *Journal of Applied Physics*, vol. 76, no. 8, pp. 4596–4602, 1994.

- [62] K. Sakuma, K. Machida, K. Kamoshida, Y. Sato, K. Imai, and E. Arai, "Silicon surface cleaning by oxidation with electron cyclotron resonance oxygen plasma after contact hole dry etching," *Journal of Vacuum Science and Technology B*, vol. 13, no. 3, pp. 902–907, 1995.
- [63] M. Kim, O. Olubuyide, J. Yoon, and J. L. Hoyt, "Selective epitaxial growth of Ge-on-Si for photodiode applications," *ECS Transactions*, vol. 16, no. 10, pp. 837–847, 2008.
- [64] M. L. Lee, E. A. Fitzgerald, M. T. Bulsara, M. T. Currie, and A. Lochtefeld, "Strained Si, SiGe, and Ge channels for high-mobility metal-oxidesemiconductor field-effect transistors," *Journal of Applied Physics*, vol. 97, no. 1, pp. 011101–011101–27, 2005.
- [65] N. Cerniglia and P. Wang, "Dissolution of germanium in aqueous hydrogen peroxide solution," *Journal of the Electrochemical Society*, vol. 109, no. 6, pp. 508– 512, 1962.
- [66] K. Sawano, K. Kawaguchi, S. Koh, Y. Hirose, Y. attori, K. Nakagawa, and Y. Shiraki, "Surface planarization of strain-relaxed sige buffer layers by cmp and post cleaning," *Journal of the Electrochemical Society*, vol. 150, no. 7, pp. G376–G379, 2003.
- [67] T. I. Kamins, K. Nauka, R. D. Jackowitz, J. L. Hoyt, D. B. Noble, and J. F. Gibbons, "Electrical characteristics of diodes fabricated in selective Si/Si_{1-x}ge_x,"
- [68] J. Liu, GeSi Photodetectors and Electro-absorption Modulators for Si Electronicphotonic Integrated Circuits. PhD thesis, MIT, 2007.
- [69] M. W. Geis, S. J. Spector, M. E. Grein, R. J. Schulein, J. U. Yoon, D. M. Lennon, C. M. Wynn, S. T. Palmacci, F. Gan, F. X. Kaertner, and T. M. Lyszczarz, "All silicon infrared photodiodes: photo response and effects of processing temperature," *Optics Express*, vol. 15, no. 25, pp. 16886–16895, 2007.
- [70] O. Olubuyide, Low Pressure Epitaxial Growth, Fabrication and Characterization of Ge-on-Si Photodiodes. PhD thesis, MIT, 2007.
- [71] E. A. Polyzoeva, "Hole mobility in strained Ge/relaxed SiGe with a Highk/metal gate stack," master's thesis, Massachusetts Institute of Technology, Electrical Engineering and Computer Science, 2011.
- [72] J. Hennessy, High Mobility Germanium MOSFETs: Study of Ozone Surface Passivation and n-type Dopant Channel Implants Combined with ALD Dielectrics. PhD thesis, MIT, 2010.

- [73] C. O. Chui, F. Ito, and K. C. Saraswat, "Scalability and electrical properties of germanium oxynitride MOS dielectrics," *IEEE Electron Device Letters*, vol. 25, no. 9, pp. 613–615, 2004.
- [74] J. J.-H. Chen, J. Bojarezuk, N. A., H. Shang, M. Copel, J. B. Hannon, J. Karasinski, E. Preisler, S. K. Banerjee, and S. Guha, "Ultrathin Al₂O₃ and HfO₂ gate dielectrics on surface-nitrided Ge," *IEEE Transactions on Electron Devices*, vol. 51, no. 9, pp. 1441–1447, 2004.
- [75] W. P. Bai, N. Lu, and D.-L. Kwong, "Si interlayer passivation on germanium MOS capacitors with high-k dielectric and metal gate," *IEEE Electron Device Letters*, vol. 26, no. 6, pp. 378–380, 2005.
- [76] H. Nakashima, Y. Iwamura, K. Sakamoto, D. Wang, K. Hirayama, K. Yamamoto, and H. Yang, "Postmetallization annealing effect of TiN-gate Ge metal-oxide-semiconductor capacitor with ultrathin SiO₂/GeO₂ bilayer passivation," *Applied Physics Letters*, vol. 98, no. 25, pp. 252102 – 252102–3, 2011.
- [77] J. R. Weber, A. Janotti, P. Rinke, and C. G. Van de Walle, "Dangling-bond defects and hydrogen passivation in germanium," *Applied Physics Letters*, vol. 91, no. 14, pp. 142101 – 142101–3, 2007.
- [78] D. K. Schroder, Semiconductor Material and Device Characterization, 3rd ed. John Wiley & Sons, Inc., 2006.
- [79] A. S. Grove and D. J. Fitzgerald, "Surface effects on p-n junctions: characteristics of surface space-charge regions under non-equilibrium conditions," *Solid-State Electronics*, vol. 9, pp. 783–806, 1966.
- [80] Y. Murakami and T. Shingyouji, "Separation and analysis of diffusion and generation components of pn leakage current in various silicon wafers," *Journal of Applied Physics*, vol. 75, no. 1, pp. 3548–3552, 1994.
- [81] J. Lauwaert, J. Van Gheluwe, J. Vanhellemont, E. Simoen, and P. Clauws, "Electronic properties of titanium and chromium impurity centers in germanium," *Journal of Applied Physics*, vol. 105, no. 7, pp. 073707–073707–6, 2009.
- [82] P. Clauws, J. Van Gheluwe, J. Lauwaert, E. Simoen, J. Vanhallemont, M. Meuris, and A. Theuwis, "Deep level transient spectroscopy of transition metal impurities in germanium," *Physica B*, vol. 401-402, pp. 188–191, 2007.
- [83] P. Clauws and E. Simoen, "Metals in germanium," Materials Science in Semiconductor Processing, vol. 9, pp. 546–553, 2006.
- [84] E. Simoen, P. Clauws, G. Huylebroeck, and J. Vennik, "DLTS of gold impurities in germanium," *Semiconductor Science and Technology*, vol. 2, pp. 507–512, 1987.

- [85] P. Clauws, G. Huylebroeck, E. Simoen, P. Vermaercke, F. De Smet, and J. Vennik, "DLTS of the third acceptor level of substitutional copper in germanium," *Semiconductor Science and Technology*, vol. 4, pp. 910–914, 1989.
- [86] G. Huylebroeck, P. Clauws, E. Simoen, and J. Vennik, "DLTS of nickel impurities in germanium," *Solid State Communications*, vol. 82, no. 5, pp. 367–369, 1992.
- [87] W. W. Tyler and H. H. Woodbury, "Properties of Germanium Doped with Iron. I. Electrical Conductivity," *Physical Review*, vol. 96, no. 4, pp. 874–882, 1954.
- [88] S. M. Sze and J. C. Irvin, "Resistivity, mobility and impurity levels in GaAs, Ge, and Si at 300k," *Solid-State Electronics*, vol. 11, no. 6, pp. 599–602, 1968.
- [89] D. V. Lang, "Deep-level transient spectroscopy: A new method to characterize traps in semiconductors," *Journal of Applied Physics*, vol. 45, no. 7, pp. 3023– 3032, 1974.
- [90] F. H. Baumann and W. Schroter, "Deformation-induced point defects in germanium," *Philosophical Magzine B*, vol. 48, no. 1, pp. 55–61, 1983.
- [91] E. Simoen, P. Clauws, and J. Vennik, "DLTS of grown-in dislocations in p- and n-type high-purity germanium," *Solid State Communications*, vol. 54, no. 12, pp. 1025–1029, 1985.
- [92] F. D. Auret, P. J. Janse van Rensburg, M. Hayes, J. M. Nel, W. E. Meyer, S. Decoster, V. Matias, and A. Vantomme, "Electrical characterization of defects introduced in *n*-type Ge during indium implantation," *Applied Physics Letters*, vol. 89, no. 15, pp. 152123–152123–3, 2006.
- [93] F. Poulin and J. C. Bourgoin, "Minority carrier traps in electron irradiated n-type germanium," *Recent Developments in Condensed Matter Physics*, vol. 3, pp. 83–88, 1981.
- [94] P. M. Mooney, F. Poulin, and J. C. Bourgoin, "Annealing of electron-induced defects in *n*-type germanium," *Physical Review B*, vol. 28, no. 6, pp. 3372–3377, 1983.
- [95] V. P. Markevich, I. D. Hawkins, A. R. Peaker, V. V. Litvinov, L. I. Murin, L. Dobaczewski, and J. L. Lindstrom, "Electronic properties of vacancy-oxygen complex in Ge crystals," *Applied Physics Letters*, vol. 81, no. 10, pp. 1821–1823, 2002.
- [96] F. D. Auret, W. E. Meyer, S. Coelho, and M. Hayes, "Electrical characterization of defects introduced during electron beam deposition of Pd Schottky contacts on *n*-type Ge," *Applied Physics Letters*, vol. 88, no. 24, pp. 242110–242110–3, 2006.

- [97] J. Coutinho, V. J. B. Torres, R. Jones, A. Carvalho, S. Oberg, and P. R. Briddon, "Calculation of deep carrier traps in a divacancy in germanium crystals," *Applied Physics Letters*, vol. 88, no. 9, pp. 091919–091919–3, 2006.
- [98] C. E. Lindberg, J. Lundsgaard Hansen, P. Bomholt, A. Mesli, K. Bonde Nielsen, A. Nylandsted Larsen, and L. Dobaczewski, "The antimony-vacancy defect in *p*type germanium," *Applied Physics Letters*, vol. 87, no. 17, pp. 172103–172103–3, 2005.
- [99] F. Poulin and J. C. Bourgoin, "Threshold energy for atomic displacement in electron irradiated germanium," *Revue de Physique Appliquee*, vol. 15, pp. 15– 19, 1980.
- [100] V. I. Kolkovsky, M. C. Petersen, and A. N. Larsen, "Alpha-particle irradiationinduced defects in n-type germanium," *Applied Physics Letters*, vol. 90, no. 11, pp. 112110–112110–3, 2007.
- [101] E. Simoen, K. Opsomer, C. Claeys, K. Maex, C. Detavernier, R. L. Van Meirhaeghe, S. Forment, and P. Clauws, "Deep level transient spectroscopy study of nickel-germanide Schottky barriers on *n*-type germanium," *Applied Physics Letters*, vol. 88, no. 18, pp. 183506–183506–3, 2006.
- [102] F. Poulin and J. C. Bourgoin, "Characteristics of the electron traps produced by electron irradiation in n-type germanium," *Physical Review B*, vol. 26, no. 12, pp. 6788–6794, 1982.
- [103] A. Satta, T. Janssens, T. Clarysse, E. Simoen, M. Meuris, A. Benedetti, I. Hofilijk, B. De Jaeger, C. Demeurisse, and W. Vandervorst, "P implantation doping of Ge: Diffusion, activation, and recrystallization," *Journal of Vacuum Science* and Technology B, vol. 24, no. 1, pp. 494–498, 2006.
- [104] S. Koffel, N. Cherkashin, F. Houdellier, M. J. Hytch, G. Benassayag, P. Scheiblin, and A. Claverie, "End of range defects in Ge," *Journal of Applied Physics*, vol. 105, no. 12, pp. 126110–126110–3, 2009.
- [105] A. Claverie, S. Koffel, N. Cherkashin, G. Benassayag, and P. Scheiblin, "Amorphization, recrystallization and end of range defects in germanium," *Thin Solid Films*, vol. 518, pp. 2307–2313, 2010.
- [106] F. Panciera, P. F. Fazzini, M. Collet, J. Boucher, E. Bedel, and F. Cristiano, "End-of-range defects in germanium and their role in boron deactivation," Applied Physics Letters, vol. 97, no. 1, pp. 012105–012105–3, 2010.
- [107] P. Clauws, "Oxygen related defects in germanium," Materials Science and Engineering, vol. B36, pp. 213–220, 1996.
- [108] J. Bloem, C. Haas, and P. Penning, "Properties of oxygen in germanium," Journal of Physics and Chemistry of Solids, vol. 12, pp. 22–27, 1959.

- [109] C. S. Fuller and F. H. Doleiden, "The ionization behavior of donors formed from oxygen in germanium," *Journal of Physics and Chemistry of Solids*, vol. 19, no. 3-4, pp. 251–260, 1961.
- [110] C. S. Fuller, W. Kaiser, and C. D. Thurmond, "Donor equilibria in the germanium-oxygen system," *Journal of Physics and Chemistry of Solids*, vol. 17, no. 3-4, pp. 301–307, 1961.
- [111] L. Dobaczewski, K. Bonde Nielsen, N. Zangenberg, B. Bech Nielsen, A. R. Peaker, and V. P. Markevich, "Donor level of bond-center hydrogen in germanium," *Physical Review B*, vol. 69, no. 24, pp. 245207–245207–6, 2004.
- [112] E. E. Haller, B. Joos, and L. M. Falicov, "Acceptor complexes in germanium: Systems with tunneling hydrogen," *Physical Review B*, vol. 21, no. 10, pp. 4729– 4739, 1980.
- [113] J. R. Weber, A. Janotti, P. Rinke, and C. G. Van de Walle, "Dangling-bond defects and hydrogen passivation in germanium," *Applied Physics Letters*, vol. 91, no. 14, pp. 142101–142101–3, 2007.
- [114] A. Toriumi, T. Tabata, C. H. Lee, T. Nishimura, K. Kita, and K. Nagashio, "Opportunities and challenges for Ge CMOS - Control of interfacing field on Ge is a key," *Microelectronic Engineering*, vol. 86, pp. 1571–1576, 2009.
- [115] K. Kita, T. Takahashi, H. Nomura, S. Suzuki, T. Nishimura, and A. Toriumi, "Control of high-k/germanium interface properties through selection of highk materials and suppression of GeO volatilization," *Applied Surface Science*, vol. 254, pp. 6100–6105, 2008.
- [116] K.-S. Chen and K.-S. Ou, "Modification of curvature-based thin-film residual stress measurement for MEMS applications," *Journal of Micromechanics and Microengineering*, vol. 12, pp. 917–924, 2002.